UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

FORM 10-Q

□ QUARTERLY REPORT PURSUANT TO SECTION	ON 13 OR 15(d) OF THE SECURI	TIES EXCHANGE ACT OF 1934	
	For the quarterly period en	ded February 28, 2011	
	OR		
☐ TRANSITION REPORT PURSUANT TO SECTIO	N 13 OR 15(d) OF THE SECURIT	TIES EXCHANGE ACT OF 1934	
1	For the transition period fromCOMMISSION FILE N		
TORTOISE	CAPITAL RESC		RATION
MARYLAND			20-3431375
(State or other jurisdiction of incorporation or	organization)	(I.R.S. Er	mployer Identification No.)
	11550 ASH STREE LEAWOOD, KA (Address of principal execu	NSAS 66211	
	(913) 981- (Registrant's telephone numl		
(Former na	Not Appli ame, former address and former		report)
Indicate by check mark whether the registrant (1) has fil months (or such shorter period that the registrant was re			
Indicate by check mark whether the registrant has submposted pursuant to Rule 405 of Regulation S-T ($\S 232.4$ and post such files). Yes \square No \square			
Indicate by check mark whether the registrant is a larg "large accelerated filer", "accelerated filer", and "smalle			
Large accelerated filer □	Accelerated filer (Do not check if a smaller	Non-accelerated filer ☑ reporting company)	Smaller reporting company □
Indicate by check mark whether the registrant is a shell	company (as defined in Rule 12b-2	of the Exchange Act). Yes \square	No ☑
The number of shares of the issuer's Common Stock, \$0	0.001 par value, outstanding as of M	March 31, 2011 was 9,156,931.	

Tortoise Capital Resources Corporation

FORM 10-Q

FOR THE QUARTERLY PERIOD ENDED FEBRUARY 28, 2011

TABLE OF CONTENTS

PART I.	FINANCIAL INFORMATION	
Item 1.	Financial Statements	
	Statements of Assets and Liabilities as of February 28, 2011 (unaudited) and November 30, 2010	1
	Schedules of Investments as of February 28, 2011 (unaudited) and November 30, 2010	2
	Statements of Operations for the three months ended February 28, 2011 (unaudited) and the	
	three months ended February 28, 2010 (unaudited)	4
	Statements of Changes in Net Assets for the three months ended February 28, 2011 (unaudited),	
	the three months ended February 28, 2010 (unaudited) and the year ended November 30, 2010	6
	Statements of Cash Flows for the three months ended February 28, 2011 (unaudited) and the	
	three months ended February 28, 2010 (unaudited)	7
	Financial Highlights for the three months ended February 28, 2011 (unaudited), the	
	three months ended February 28, 2010 (unaudited) and the year ended November 30, 2010	8
	Notes to Financial Statements (unaudited)	9
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	20
Item 3.	Quantitative and Qualitative Disclosures About Market Risk	24
Item 4.	Controls and Procedures	25
PART II.	OTHER INFORMATION	
Item 1.	<u>Legal Proceedings</u>	26
Item 1A.	Risk Factors	26
Item 2.	Unregistered Sales of Equity Securities and Use of Proceeds	26
Item 3.	<u>Defaults Upon Senior Securities</u>	26
Item 4.	(Removed and Reserved)	26
Item 5.	Other Information	26
Item 6.	<u>Exhibits</u>	26
SIGNATUR	<u>RES</u>	



STATEMENTS OF ASSETS & LIABILITIES

	February 28, 2011	November 30, 2010
	(Unaudited)	
Assets		
Investments at fair value, control (cost \$4,593,000 and \$18,122,054, respectively)	\$ 8,736,367	\$ 23,260,566
Investments at fair value, affiliated (cost \$44,774,854 and \$31,329,809, respectively)	63,310,963	49,066,009
Investments at fair value, non-affiliated (cost \$22,311,998 and \$21,628,965, respectively)	24,058,040	22,875,848
Total investments (cost \$71,679,852 and \$71,080,828, respectively)	96,105,370	95,202,423
Receivable for Adviser expense reimbursement	117,340	109,145
Receivable for investments sold	_	5,198
Interest receivable from control investments	_	42,778
Dividends receivable	50	83
Deferred tax asset	919,005	656,743
Prepaid expenses and other assets	30,994	25,023
Total assets	97,172,759	96,041,393
Liabilities		
Base management fees payable to Adviser	352,020	327,436
Distribution payable to common stockholders	914,654	_
Accrued expenses and other liabilities	228,293	234,784
Total liabilities	1,494,967	562,220
Net assets applicable to common stockholders	\$ 95,677,792	\$ 95,479,173
Net Assets Applicable to Common Stockholders Consist of:		
Warrants, no par value; 945,594 issued and outstanding at February 28, 2011 and		
November 30, 2010 (5,000,000 authorized)	\$ 1,370,700	\$ 1,370,700
Capital stock, \$0.001 par value; 9,146,506 shares issued and outstanding at		
February 28, 2011 and November 30, 2010 (100,000,000 shares authorized)	9,147	9,147
Additional paid-in capital	97,530,301	98,444,952
Accumulated net investment loss, net of income taxes	(3,195,579)	(3,308,522)
Accumulated realized loss, net of income taxes	(17,730,436)	(18,532,648)
Net unrealized appreciation of investments, net of income taxes	17,693,659	17,495,544
Net assets applicable to common stockholders	\$ 95,677,792	\$ 95,479,173
Net Asset Value per common share outstanding (net assets applicable		
to common stock, divided by common shares outstanding)	\$ 10.46	\$ 10.44
<u> </u>		



SCHEDULE OF INVESTMENTS

February 28, 2011 (Unaudited)

Energy Infrastructure

	inii asti uctui c			
Company	Segment	Type of Investment	Cost	Fair Value
Control Investments(1)				
Mowood, LLC	Midstream/	Equity Interest (100%) ₍₂₎	\$ 793,000 \$	4,936,367
	Downstream	Subordinated Debt (14.0% Due 12/31/11)(2)	3,800,000	3,800,000
Total Control Investments — 9.1%(3)			4,593,000	8,736,367
Affiliated Investments(4)				
High Sierra Energy, LP	Midstream	Common Units (1,042,685)(2)(5)	19,823,161	20,520,033
International Resource Partners LP	Coal	Class A Units (500,000)(2)	9,237,333	30,455,000
LONESTAR Midstream Partners, LP	Midstream	Class A Units (1,327,900)(2)(5)(6)	2,149,269	216,000
LSMP GP, LP	Midstream	GP LP Units (180)(2)(5)(6)	120,046	39,000
VantaCore Partners LP	Aggregates	Common Units (933,430)(2)	12,941,737	11,639,872
, unition 2 unition 22	11881084103	Preferred Units (23,185)(2)(5)	359,367	405,737
		Incentive Distribution Rights (988)(2)(5)	143,941	35,321
Total Affiliated Investments — 66.2%(3)		3 (, , , , , , , , , , , , , ,	44,774,854	63,310,963
,			.,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	32,233,33
Non-affiliated Investments				
Buckeye Partners, L.P.	Midstream	Common Units (38,400) ₍₇₎	2,464,393	2,489,088
Chesapeake Midstream Partners, L.P.	Midstream	Common Units (55,300)(7)	1,479,755	1,440,012
El Paso Pipeline Partners, L.P.	Midstream	Common Units (56,700)(7)	2,001,136	2,137,590
Energy Transfer Partners, L.P.	Midstream	Common Units (55,600)(7)	2,644,492	3,048,548
Enterprise Products Partners L.P.	Midstream	Common Units (37,600)(7)	1,205,729	1,639,360
EV Energy Partners, L.P.	Upstream	Common Units (78,900)(7)	2,267,420	3,599,418
High Sierra Energy GP, LLC	Midstream	Equity Interest (2.37%)(2)(5)	1,999,275	53,486
Inergy, L.P.	Midstream	Common Units (7,100)(7)	283,859	294,508
Kinder Morgan Management, LLC	Midstream	Common Units (21,038)(7)(8)	1,139,279	1,380,310
ONEOK Partners, L.P.	Midstream	Common Units (17,100)(7)	972,313	1,421,865
Regency Energy Partners LP	Midstream	Common Units (86,800)(7)	2,203,867	2,410,436
Williams Partners L.P.	Midstream	Common Units (37,500)(7)	1,451,811	1,944,750
Fidelity Institutional Government	Short-term	Class I Shares (2,198,669)	2,198,669	2,198,669
Portfolio	investment			
Total Non-affiliated Investments — 25.1%(3)			22,311,998	24,058,040
Total Investments — 100.4%(3)			\$ 71,679,852	96,105,370

- (1) Control investments are generally defined under the Investment Company Act of 1940 as companies in which at least 25% of the voting securities are owned; see Note 8 to the financial statements for further disclosure.
- (2) Restricted securities have been fair valued in accordance with procedures approved by the Board of Directors and have a total fair value of \$72,100,816, which represents 75.4% of net assets applicable to common stockholders; see Note 7 to the financial statements for further disclosure.
- (3) Calculated as a percentage of net assets applicable to common stockholders.
- (4) Affiliated investments are generally defined under the Investment Company Act of 1940 as companies in which at least 5% of the voting securities are owned. Affiliated investments in which at least 25% of the voting securities are owned are generally defined as control investments as described in footnote 1; see Note 8 to the financial statements for further disclosure.
- (5) Currently non-income producing.
- (6) In July 2008, LONESTAR Midstream Partners, LP sold its assets to Penn Virginia Resource Partners, L.P. (PVR). LONESTAR has no continuing operations, but currently holds certain rights to receive future payments from PVR relative to the sale. LSMP GP, LP indirectly owns the general partner of LONESTAR Midstream Partners, LP. See Note 9 to the financial statements for additional information.
- (7) Publicly-traded company.
- (8) Security distributions are paid-in-kind.

SCHEDULE OF INVESTMENTS

November 30, 2010

	Elier 67			
	Infrastructure			
Company	Segment	Type of Investment	Cost	Fair Value
Control Investments(1)				
Mowood, LLC	Midstream/	Equity Interest (100%)(2)	\$ 793,000	\$ 5,492,247
	Downstream	Subordinated Debt (14.0% Due 12/31/11)(2)	3,800,000	3,800,000
VantaCore Partners LP	Aggregates	Common Units (933,430)(2)	13,385,113	13,814,764
		Incentive Distribution Rights (988)(2)(5)	143,941	153,555
Total Control Investments — 24.3%(3)			18,122,054	23,260,566
Affiliated Investments(4)				
High Sierra Energy, LP	Midstream	Common Units (1,042,685)(2)(5)	19,823,161	20,666,009
International Resource Partners LP	Coal	Class A Units (500,000)(2)	9,237,333	28,155,000
LONESTAR Midstream Partners, LP	Midstream	Class A Units (1,327,900)(2)(5)(6)	2,149,269	208,000
LSMP GP, LP	Midstream	GP LP Units (180)(2)(5)(6)	120,046	37,000
Total Affiliated Investments — 51.4%(3)			31,329,809	49,066,009
Non-affiliated Investments				

Energy

Non-affiliated Investments				
Abraxas Petroleum Corporation	Upstream	Common Units (1,646,376)(5)(7)	2,448,984	7,013,562
Energy Transfer Partners, L.P.	Midstream	Common Units (50,900)(7)	2,431,551	2,579,103
Enterprise Products Partners L.P.	Midstream	Common Units (37,600)(7)	1,227,913	1,582,208
EV Energy Partners, L.P.	Upstream	Common Units (78,900)(7)	2,291,374	3,011,613
High Sierra Energy GP, LLC	Midstream	Equity Interest (2.37%)(2)(5)	1,999,275	602,834
Inergy, L.P.	Midstream	Common Units (7,100)(7)	288,864	277,042
Kinder Morgan Management, LLC	Midstream	Common Units (20,678)(7)(8)	1,139,279	1,323,212
ONEOK Partners, L.P.	Midstream	Common Units (17,100)(7)	991,807	1,354,491
PostRock Energy Corporation	Upstream	Common Units (260,500)(5)(7)	4,949,500	950,825
Regency Energy Partners LP	Midstream	Common Units (46,500)(7)	1,165,596	1,195,050
Williams Partners L.P.	Midstream	Common Units (32,300)(7)	1,228,629	1,519,715
Fidelity Institutional Government	Short-term	Class I Shares (1,466,193)	1,466,193	1,466,193
Portfolio	investment			
Total Non-affiliated Investments — 24.0%(3)			21,628,965	22,875,848
Total Investments — 99.7%(3)			\$71,080,828	\$95,202,423

- Control investments are generally defined under the Investment Company Act of 1940 as companies in which at least 25% of the voting securities are owned; see Note 8 to the financial statements for further disclosure. Restricted securities have been fair valued in accordance with procedures approved by the Board of Directors and have a total fair value of \$72,929,409, which represents 76.4% of net assets applicable to common stockholders; see Note 7 to the financial statements for further disclosure.
- Calculated as a percentage of net assets applicable to common stockholders.
- Affiliated investments are generally defined under the Investment Company Act of 1940 as companies in which at least 5% of the voting securities are owned. Affiliated investments in which at least 25% of the voting securities are owned are generally defined as control investments as described in footnote 1; see Note 8 to the financial statements for further disclosure.

Currently non-income producing.

In July 2008, LONESTAR Midstream Partners, LP sold its assets to Penn Virginia Resource Partners, L.P. (PVR). LONESTAR has no continuing operations, but currently holds certain rights to receive future payments from PVR relative to the sale. LSMP GP, LP indirectly owns the general partner of LONESTAR Midstream Partners, LP. See Note 9 to the financial statements for additional information. Publicly-traded company.

Security distributions are paid-in-kind.

STATEMENTS OF OPERATIONS (Unaudited)

Base management fees 352,020 309,922 Professional fees 80,876 85,162 Directors' fees 14,573 26,161 Stockholder communication expenses 12,912 15,703 Administrator fees 9,387 14,460 Fund accounting fees 7,328 6,972 Registration fees 6,160 6,355 Stock transfer agent fees 3,353 3,130 Franchise tax expense 4,998 2,572 Custodian fees and expenses 1,382 1,575 Other expenses 12,874 12,478 Total Operating Expenses 505,863 484,490 Interest expense — 45,619 Total Expenses 505,863 530,109 Less expense reimbursement by Adviser (117,340) (51,654) Net Expenses 388,523 478,455	Investment Income	For the three months ended February 28, 2011	For the three months ended February 28, 2010
Affiliated investments 384,009 856,892 Non-affiliated investments 277,815 75,985 Total distributions from investments 331,991 1,488,756 Less return of capital on distributions 305,725 (998,640) Net distributions from investments 426,266 490,116 Interest income from control investments 135,330 191,431 Dividends from money market mutual funds 190 217 Fee income — 10,392 Total Investment Income 352,020 309,222 Base management fees 352,020 309,922 Professional fees 80,876 85,162 Directors' fees 40,873 26,161 Stockholder communication expenses 12,912 15,733 Fund accounting fees 9,387 14,460 Foliation from samples for gent fees 1,382 1,575 Stock transfer agen	Distributions from investments		
Non-affiliated investments 277,815 75,985 Total distributions from investments 305,725 (998,640) Net distributions from investments 426,266 490,116 Interest income from control investments 135,330 191,431 Dividends from money market mutual funds 9 217 Fee income - 1,392 Total Investment Income \$561,786 692,156 Operating Expenses Base management fees 352,020 309,922 Professional fees 8,876 85,162 Directors' fees 14,573 26,161 Stockholder communication expenses 12,912 15,703 Administrator fees 9,387 14,460 Fund accounting fees 9,387 14,460 Fundaceounting fees 3,353 3,130 Franchise tax expense 4,998 2,572 Other expense 1,382 1,575 Other expenses 12,874 12,478 Total Dyerating Expenses 505,863 380,109 Custodian	Control investments	\$ 70,167	\$ 555,879
Total distributions from investments 731,991 1,488,750 Less return of capital on distributions 305,725) (998,640) Net distributions from investments 426,666 490,116 Interest income from control investments 195,333 191,431 Dividends from money market mutual funds 190 217 Fee income — 10,392 Total Investment Income \$61,786 692,156 Operating Expenses Seas management fee 352,020 309,922 Professional fees 352,020 309,922 20,616 850,162 850,162 850,162 14,573 20,161 20,161 850,162 14,573 20,161 20,161 850,162 14,573 20,161 850,162 15,003 14,480 14,573 20,161 20,161 850,162 15,003 14,480 15,003 14,480 10,162 10,162 10,162 10,162 10,162 10,162 10,162 10,162 10,162 10,162 10,162 10,162 10,162 10,162 10,162 10,162 </td <td>Affiliated investments</td> <td>384,009</td> <td>856,892</td>	Affiliated investments	384,009	856,892
Less return of capital on distributions (305,725) (998,640) Net distributions from investments 426,266 490,116 Interest income from control investments 135,330 191,431 Dividends from money market mutual funds 190 217 Fee income — 10,392 Total Investment Income — 10,392 Operating Expenses Base management fees 352,020 309,922 Professional fees 80,876 85,162 Directors' fees 14,573 26,161 Stockholder communication expenses 12,912 15,703 Administrator fees 9,387 14,660 Fund accounting fees 9,387 14,660 Fund accounting fees 6,160 6,355 Stock transfer agent fees 3,353 3,130 Franchise tax expense 4,998 2,572 Custodian fees and expenses 1,382 1,575 Other expenses 1,287 1,278 Total Operating Expenses 50,863 384,490	Non-affiliated investments	277,815	75,985
Net distributions from investments 426,266 490,116 Interest income from control investments 135,330 191,431 Dividends from money market mutual funds 190 217 Fee income — 10,392 Total Investment Income 561,786 692,156 Operating Expenses Base management fees 352,020 309,922 Professional fees 80,876 85,162 Directors' fees 14,573 26,161 Stockholder communication expenses 12,912 15,703 Administrator fees 9,387 14,460 Fund accounting fees 7,328 6,972 Registration fees 6,160 6,355 Stock transfer agent fees 3,353 3,130 Franchise tax expense 4,998 2,572 Custodian fees and expenses 1,382 1,575 Other expenses 1,2874 12,478 Total Operating Expenses - 45,619 Interest expense - 45,619 Total Expenses 505	Total distributions from investments	731,991	1,488,756
Interest income from control investments 135,330 191,431 Dividends from money market mutual funds 190 217 Fee income — 10,392 Total Investment Income 561,786 692,156 Operating Expenses Base management fees 352,020 309,922 Professional fees 80,876 85,162 Directors' fees 14,573 26,161 Stockholder communication expenses 12,912 15,703 Administrator fees 9,387 14,460 Fund accounting fees 7,328 6,972 Registration fees 6,160 6,355 Stock transfer agent fees 3,353 3,130 Franchise tax expense 4,998 2,572 Custodian fees and expenses 1,382 1,575 Other expenses 12,874 12,478 Total Deprating Expenses 505,863 484,490 Interest expense — 45,619 Total Expenses 505,863 530,109 Less expense reimbursement by Adviser	Less return of capital on distributions	(305,725)	(998,640)
Dividends from money market mutual funds 190 217 Fee income — 10,392 Total Investment Income 561,786 692,156 Operating Expenses Sease management fees 352,020 309,922 Professional fees 80,876 85,162 20,162 15,703 26,161 20,161	Net distributions from investments	426,266	490,116
Fee income — 10,392 Total Investment Income 561,786 692,156 Operating Expenses Season management fees 352,020 309,922 Professional fees 80,876 85,162 Directors' fees 14,573 26,161 Stockholder communication expenses 12,912 15,703 Administrator fees 9,387 14,460 Fund accounting fees 9,387 14,460 Registration fees 6,160 6,355 Stock transfer agent fees 3,333 3,130 Franchise tax expense 4,998 2,572 Custodian fees and expenses 1,382 1,575 Other expenses 12,874 12,478 Total Operating Expenses 50,863 484,490 Interest expense 50,863 50,109 Less expenses reimbursement by Adviser 50,563 484,490 Net Expenses 38,8523 478,455 Net Investment Income, before Income Taxes 173,263 213,701 Deferred tax expense (60,320) (32	Interest income from control investments	135,330	191,431
Operating Expenses 561,786 692,156 Base management fees 352,020 309,922 Professional fees 80,876 85,162 Directors' fees 14,573 26,161 Stockholder communication expenses 12,912 15,703 Administrator fees 9,387 14,460 Fund accounting fees 7,328 6,972 Registration fees 6,160 6,355 Stock transfer agent fees 3,353 3,130 Franchise tax expense 4,998 2,572 Custodian fees and expenses 1,382 1,575 Other expenses 12,874 12,478 Total Operating Expenses 505,863 484,490 Interest expense - 45,619 Total Expenses 505,863 530,109 Les expense erinbursement by Adviser (117,340) (51,654) Net Expenses 388,523 478,455 Net Investment Income, before Income Taxes (60,320) (32,694)	Dividends from money market mutual funds	190	217
Operating Expenses Base management fees 352,020 309,922 Professional fees 80,876 85,162 Directors' fees 14,573 26,161 Stockholder communication expenses 12,912 15,703 Administrator fees 9,387 14,460 Fund accounting fees 7,328 6,972 Registration fees 6,160 6,355 Stock transfer agent fees 3,353 3,130 Franchise tax expense 4,998 2,572 Custodian fees and expenses 1,382 1,575 Other expenses 12,874 12,478 Total Operating Expenses 505,863 484,490 Interest expense — 45,619 Total Expenses 505,863 530,109 Net Expenses 388,523 478,855 Net Investment Income, before Income Taxes 173,263 213,701 Deferred tax expense (60,320) (32,694)	Fee income	_	10,392
Base management fees 352,020 309,922 Professional fees 80,876 85,162 Directors' fees 14,573 26,161 Stockholder communication expenses 12,912 15,703 Administrator fees 9,387 14,460 Fund accounting fees 7,328 6,972 Registration fees 6,160 6,355 Stock transfer agent fees 3,353 3,130 Franchise tax expense 4,998 2,572 Custodian fees and expenses 12,874 12,478 Total Operating Expenses 12,874 12,478 Total Operating Expenses 505,863 530,109 Interest expense - 45,619 Total Expenses 505,863 530,109 Less expense reimbursement by Adviser (117,340) (51,654) Net Expenses 388,523 478,455 Net Investment Income, before Income Taxes 173,263 213,701 Deferred tax expense (60,320) (32,694)	Total Investment Income	561,786	692,156
Professional fees 80,876 85,162 Directors' fees 14,573 26,161 Stockholder communication expenses 12,912 15,703 Administrator fees 9,387 14,460 Fund accounting fees 7,328 6,972 Registration fees 6,160 6,355 Stock transfer agent fees 3,353 3,130 Franchise tax expense 4,998 2,572 Custodian fees and expenses 1,382 1,575 Other expenses 12,874 12,478 Total Operating Expenses 505,863 484,490 Interest expense 505,863 530,109 Less expense reimbursement by Adviser (117,340) (51,654) Net Expenses 388,523 478,455 Net Investment Income, before Income Taxes 173,263 213,701 Deferred tax expense (60,320) (32,694)	Operating Expenses		
Directors' fees 14,573 26,161 Stockholder communication expenses 12,912 15,703 Administrator fees 9,387 14,460 Fund accounting fees 7,328 6,972 Registration fees 6,160 6,355 Stock transfer agent fees 3,353 3,130 Franchise tax expense 4,998 2,572 Custodian fees and expenses 1,382 1,575 Other expenses 12,874 12,478 Total Operating Expenses 505,863 484,490 Interest expense — 45,619 Total Expenses 505,863 530,109 Less expense reimbursement by Adviser (117,340) (51,654) Net Expenses 388,523 478,455 Net Investment Income, before Income Taxes 173,263 213,701 Deferred tax expense (60,320) (32,694)	Base management fees	352,020	309,922
Stockholder communication expenses 12,912 15,703 Administrator fees 9,387 14,460 Fund accounting fees 7,328 6,972 Registration fees 6,160 6,355 Stock transfer agent fees 3,353 3,130 Franchise tax expense 4,998 2,572 Custodian fees and expenses 1,382 1,575 Other expenses 12,874 12,478 Total Operating Expenses 505,863 484,490 Interest expense - 45,619 Total Expenses 505,863 530,109 Less expense reimbursement by Adviser (117,340) (51,654) Net Expenses 388,523 478,455 Net Investment Income, before Income Taxes 173,263 213,701 Deferred tax expense (60,320) (32,694)	Professional fees	80,876	85,162
Administrator fees 9,387 14,460 Fund accounting fees 7,328 6,972 Registration fees 6,160 6,355 Stock transfer agent fees 3,353 3,130 Franchise tax expense 4,998 2,572 Custodian fees and expenses 1,382 1,575 Other expenses 12,874 12,478 Total Operating Expenses 505,863 484,490 Interest expense - 45,619 Total Expenses 505,863 530,109 Less expense reimbursement by Adviser (117,340) (51,654) Net Expenses 388,523 478,455 Net Investment Income, before Income Taxes 173,263 213,701 Deferred tax expense (60,320) (32,694)	Directors' fees	14,573	26,161
Fund accounting fees 7,328 6,972 Registration fees 6,160 6,355 Stock transfer agent fees 3,353 3,130 Franchise tax expense 4,998 2,572 Custodian fees and expenses 1,382 1,575 Other expenses 12,874 12,478 Total Operating Expenses 505,863 484,490 Interest expense 505,863 530,109 Less expense reimbursement by Adviser (117,340) (51,654) Net Expenses 388,523 478,455 Net Investment Income, before Income Taxes 173,263 213,701 Deferred tax expense (60,320) (32,694)	Stockholder communication expenses	12,912	15,703
Registration fees 6,160 6,355 Stock transfer agent fees 3,353 3,130 Franchise tax expense 4,998 2,572 Custodian fees and expenses 1,382 1,575 Other expenses 12,874 12,478 Total Operating Expenses 505,863 484,490 Interest expense - 45,619 Total Expenses 505,863 530,109 Less expense reimbursement by Adviser (117,340) (51,654) Net Expenses 388,523 478,455 Net Investment Income, before Income Taxes 173,263 213,701 Deferred tax expense (60,320) (32,694)	Administrator fees	9,387	14,460
Stock transfer agent fees 3,353 3,130 Franchise tax expense 4,998 2,572 Custodian fees and expenses 1,382 1,575 Other expenses 12,874 12,478 Total Operating Expenses 505,863 484,490 Interest expense 505,863 530,109 Less expense reimbursement by Adviser (117,340) (51,654) Net Expenses 388,523 478,455 Net Investment Income, before Income Taxes 173,263 213,701 Deferred tax expense (60,320) (32,694)	Fund accounting fees	7,328	6,972
Franchise tax expense 4,998 2,572 Custodian fees and expenses 1,382 1,575 Other expenses 12,874 12,478 Total Operating Expenses 505,863 484,490 Interest expense — 45,619 Total Expenses 505,863 530,109 Less expense reimbursement by Adviser (117,340) (51,654) Net Expenses 388,523 478,455 Net Investment Income, before Income Taxes 173,263 213,701 Deferred tax expense (60,320) (32,694)	Registration fees	6,160	6,355
Custodian fees and expenses 1,382 1,575 Other expenses 12,874 12,478 Total Operating Expenses 505,863 484,490 Interest expense — 45,619 Total Expenses 505,863 530,109 Less expense reimbursement by Adviser (117,340) (51,654) Net Expenses 388,523 478,455 Net Investment Income, before Income Taxes 173,263 213,701 Deferred tax expense (60,320) (32,694)	Stock transfer agent fees	3,353	3,130
Other expenses 12,874 12,478 Total Operating Expenses 505,863 484,490 Interest expense - 45,619 Total Expenses 505,863 530,109 Less expense reimbursement by Adviser (117,340) (51,654) Net Expenses 388,523 478,455 Net Investment Income, before Income Taxes 173,263 213,701 Deferred tax expense (60,320) (32,694)	Franchise tax expense	4,998	2,572
Total Operating Expenses 505,863 484,490 Interest expense — 45,619 Total Expenses 505,863 530,109 Less expense reimbursement by Adviser (117,340) (51,654) Net Expenses 388,523 478,455 Net Investment Income, before Income Taxes 173,263 213,701 Deferred tax expense (60,320) (32,694)	Custodian fees and expenses	1,382	1,575
Interest expense — 45,619 Total Expenses 505,863 530,109 Less expense reimbursement by Adviser (117,340) (51,654) Net Expenses 388,523 478,455 Net Investment Income, before Income Taxes 173,263 213,701 Deferred tax expense (60,320) (32,694)	Other expenses	12,874	12,478
Total Expenses 505,863 530,109 Less expense reimbursement by Adviser (117,340) (51,654) Net Expenses 388,523 478,455 Net Investment Income, before Income Taxes 173,263 213,701 Deferred tax expense (60,320) (32,694)	Total Operating Expenses	505,863	484,490
Less expense reimbursement by Adviser (117,340) (51,654) Net Expenses 388,523 478,455 Net Investment Income, before Income Taxes 173,263 213,701 Deferred tax expense (60,320) (32,694)	Interest expense		45,619
Net Expenses 388,523 478,455 Net Investment Income, before Income Taxes 173,263 213,701 Deferred tax expense (60,320) (32,694)	Total Expenses	505,863	530,109
Net Investment Income, before Income Taxes 173,263 213,701 Deferred tax expense (60,320) (32,694)	Less expense reimbursement by Adviser	(117,340)	(51,654)
Deferred tax expense (60,320) (32,694)	Net Expenses	388,523	478,455
	Net Investment Income, before Income Taxes	173,263	213,701
Net Investment Income 112,943 181,007	Deferred tax expense	(60,320)	(32,694)
	Net Investment Income	112,943	181,007

${\bf STATEMENTS~OF~OPERATIONS~(} \textit{Unaudited)} \\ \textit{(Continued)}$

	n	For the three months ended February 28, 2011		For the three nonths ended bruary 28, 2010
Realized and Unrealized Gain (Loss) on Investments				
Net realized gain on control investments	\$	_	\$	1,578,001
Net realized loss on affiliated investments		_		(17,445)
Net realized gain on non-affiliated investments		373,822		27,612
Net realized gain, before income taxes		373,822		1,588,168
Deferred tax benefit (expense)		428,390		(242,971)
Net realized gain on investments		802,212		1,345,197
Net unrealized appreciation (depreciation) of control investments		(995,145)		1,535,457
Net unrealized appreciation of affiliated investments		799,909		1,208,074
Net unrealized appreciation of non-affiliated investments		499,159		197,774
Net unrealized appreciation, before income taxes		303,923		2,941,305
Deferred tax expense		(105,808)		(449,986)
Net unrealized appreciation of investments		198,115		2,491,319
Net Realized and Unrealized Gain on Investments		1,000,327		3,836,516
Net Increase in Net Assets Applicable to Common Stockholders				
Resulting from Operations	\$	1,113,270	\$	4,017,523
Net Increase in Net Assets Applicable to Common Stockholders				
Resulting from Operations Per Common Share:				
Basic and Diluted	\$	0.12	\$	0.44
Weighted Average Shares of Common Stock Outstanding:				
Basic and Diluted		9,146,506		9,078,090

STATEMENTS OF CHANGES IN NET ASSETS

	For the three months ended February 28, 2011		ded months ended , 2011 February 28, 2010		No	Year ended vember 30, 2010
		(Unaudited)		(Unaudited)		
Operations Not investigate to the control of the c	e.	112.042	Φ.	101.007	e	(4.106)
Net investment income (loss)	\$	112,943	\$	181,007	\$	(4,106)
Net realized gain (loss) on investments		802,212		1,345,197		(4,491,034)
Net unrealized appreciation of investments		198,115	_	2,491,319		19,162,014
Net increase in net assets applicable to common stockholders						
resulting from operations		1,113,270		4,017,523		14,666,874
Distributions to Common Stockholders						
Return of capital		(914,651)		(1,180,152)		(3,915,124)
Total distributions to common stockholders		(914,651)		(1,180,152)		(3,915,124)
					_	
Capital Stock Transactions						
Issuance of 68,416 common shares from reinvestment of distributions						
to stockholders		_		_		430,838
Net increase in net assets, applicable to common stockholders,						
from capital stock transactions		_		_		430,838
Total increase in net assets applicable to common stockholders		198,619		2,837,371	_	11,182,588
••						
Net Assets						
Beginning of period		95,479,173		84,296,585		84,296,585
End of period	\$	95,677,792	\$	87,133,956	\$	95,479,173
Accumulated net investment loss, net of income taxes, at the end of period	\$	(3,195,579)	\$	(3,123,409)	\$	(3,308,522)

STATEMENTS OF CASH FLOWS (Unaudited)

	For the three months ended February 28, 2011	For the three months ended February 28, 2010	
Cash Flows From Operating Activities			
Distributions received from investments	\$ 731,990	\$ 1,488,756	
Interest and dividend income received	178,331	136,600	
Fee income received		10,392	
Purchases of long-term investments	(7,970,756)	(750,000)	
Proceeds from sales of long-term investments	8,177,504	9,839,190	
Purchases of short-term investments, net	(732,476)	(5,605,655)	
Interest expense paid		(66,703)	
Operating expenses paid	(384,593)	(452,580)	
Net cash provided by operating activities		4,600,000	
Cash Flows From Financing Activities			
Repayments on revolving line of credit	_	(4,600,000)	
Net cash used in financing activities		(4,600,000)	
Net change in cash		_	
Cash — beginning of period	_	_	
Cash — end of period	\$ <u> </u>	\$ —	
Reconciliation of net increase in net assets applicable to common stockholders resulting from operations to net cash provided by operating activities			
Net increase in net assets applicable to common stockholders resulting from operations	\$ 1,113,270	\$ 4,017,523	
Adjustments to reconcile net increase in net assets applicable to common stockholders			
resulting from operations to net cash provided by operating activities:			
Purchases of long-term investments	(7,970,756)	(750,000)	
Return of capital on distributions received	305,725	998,640	
Proceeds from sales of long-term investments	8,172,306	9,839,190	
Purchases of short-term investments, net	(732,476)	(5,605,655)	
Deferred income taxes, net	(262,262)	725,651	
Realized gain on investments	(373,822)	(1,588,168)	
Net unrealized appreciation of investments	(303,923)	(2,941,305)	
Changes in operating assets and liabilities:			
(Increase) decrease in interest, dividend and distribution receivable	42,811	(55,047)	
Decrease in receivable for investments sold	5,198	_	
Increase in prepaid expenses and other assets	(5,971)	(1,668)	
Increase in base management fees payable to Adviser,			
net of expense reimbursement	16,389	9,051	
Decrease in accrued expenses and other liabilities	(6,489)	(48,212)	
Total adjustments	(1,113,270)	582,477	
Net cash provided by operating activities	<u> </u>	\$ 4,600,000	

FINANCIAL HIGHLIGHTS

	r	For the three months ended February 28, 2011		onths ended months ended		Year ended November 30, 201	
		(Unaudited)	(Unaudited)				
Common Share Data(1)							
Net Asset Value, beginning of period	\$	10.44	\$	9.29	\$	9.29	
Income from Investment Operations:							
Net investment income(2)(3)		0.01		0.02		0.00	
Net realized and unrealized gain on investments(2)		0.11		0.42		1.58	
Total increase from investment operations		0.12		0.44		1.58	
Less Distributions to Common Stockholders:							
Return of capital		(0.10)		(0.13)		(0.43)	
Total distributions to common stockholders		(0.10)		(0.13)		(0.43)	
Net Asset Value, end of period	\$	10.46	\$	9.60	\$	10.44	
Per common share market value, end of period	\$	8.50	\$	6.85	\$	7.28	
Total Investment Return, based on net asset value(4)		1.38%		5.28%		20.26%	
Total Investment Return, based on market value(s)		18.14%		12.02%		25.04%	
oplemental Data and Ratios							
Net assets applicable to common stockholders, end of period (000's)	\$	95,678	\$	87,134	\$	95,479	
Average net assets (000's)	\$	95,832	\$	85,813	\$	85,950	
Ratio of Expenses to Average Net Assets(6)							
Advisory fees		1.49%		1.46%		1.44%	
Other expenses		0.65		0.83		1.08	
Expense reimbursement		(0.50)		(0.24)		(0.36)	
Subtotal		1.64		2.05		2.16	
Interest expense		_		0.21		0.05	
Income tax expense (benefit)(7)		(1.11)		3.43		5.55	
Total expenses		0.53%		5.69%		7.76%	
Ratio of net investment income (loss) to average net assets,							
before expense reimbursement(6)		(0.02)%		0.61%		(0.36)%	
Ratio of net investment income (loss) to average net assets,							
after expense reimbursement(3)(6)		0.48%		0.85%		(0.00)9	
Portfolio turnover ratero		34.45%		3.82%		12.92%	

Information presented relates to a share of common stock outstanding for the entire period.

The per common share data for the three months ended February 28, 2010 does not reflect the change in estimate of investment income and return of capital, as described in Note 2D.

Less than \$0.01 per share or 0.01% for the year ended November 30, 2010.

Not annualized. Total investment return is calculated assuming a purchase of common stock at the net asset value per share as of the beginning of the period, reinvestment of distributions at actual prices pursuant to the

Company's dividend reinvestment plan and a sale at net asset value at the end of the period.

Not annualized. Total investment return is calculated assuming a purchase of common stock at the market value at the beginning of the period, reinvestment of distributions at actual prices pursuant to the Company's dividend reinvestment plan and a sale at the current market price on the last day of the period (excluding brokerage commissions). (5)

Annualized for periods less than one full year.

For the three months ended February 28, 2011, the Company accrued \$262,262 in net deferred income tax benefit. For the three months ended February 28, 2010, the Company accrued \$725,651 in net deferred income tax expense. For the year ended November 30, 2010, the Company accrued \$4,772,648 in net deferred income tax expense.



NOTES TO FINANCIAL STATEMENTS (Unaudited) February 28, 2011

1. Organization

Tortoise Capital Resources Corporation (the "Company") was organized as a Maryland corporation on September 8, 2005, and is a non-diversified closed-end management investment company focused on the U.S. energy infrastructure sector. The Company invests primarily in privately held and micro-cap public companies operating in the energy infrastructure sector. The Company is regulated as a business development company ("BDC") under the Investment Company Act of 1940, as amended (the "1940 Act"). The Company does not report results of operations internally on an operating segment basis. The Company is externally managed by Tortoise Capital Advisors, L.L.C. (the "Adviser"), an investment adviser specializing in listed energy infrastructure investments, such as pipeline and power companies. The Company's shares are listed on the New York Stock Exchange under the symbol "TTO."

2. Significant Accounting Policies

A. Use of Estimates — The preparation of financial statements in conformity with U.S. generally accepted accounting principles requires management to make estimates and assumptions that affect the reported amount of assets and liabilities, recognition of distribution income and disclosure of contingent assets and liabilities at the date of the financial statements. Actual results could differ from those estimates.

B. Investment Valuation — The Company invests primarily in illiquid securities including debt and equity securities of privately-held companies. These investments generally are subject to restrictions on resale, have no established trading market and are fair valued on a quarterly basis. Because of the inherent uncertainty of valuation, the fair values of such investments, which are determined in accordance with procedures approved by the Company's Board of Directors, may differ materially from the values that would have been used had a ready market existed for the investments. The Company's Board of Directors may consider other methods of valuing investments as appropriate and in conformity with U.S. generally accepted accounting principles.

The Company determines fair value to be the price that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date. The Company has determined the principal market, or the market in which the Company exits its private portfolio investments with the greatest volume and level of activity, to be the private secondary market. Typically, private companies are bought and sold based on multiples of EBITDA, cash flows, net income, revenues, or in limited cases, book value.

For private company investments, value is often realized through a liquidity event of the entire company. Therefore, the value of the company as a whole (enterprise value) at the reporting date often provides the best evidence of the value of the investment and is the initial step for valuing the Company's privately issued securities. For any one company, enterprise value may best be expressed as a range of fair values, from which a single estimate of fair value will be derived. In determining the enterprise value of a portfolio company, an analysis is prepared consisting of traditional valuation methodologies including market and income approaches. The Company considers some or all of the traditional valuation methods based on the individual circumstances of the portfolio company in order to derive its estimate of enterprise value.

The fair value of investments in private portfolio companies is determined based on various factors, including enterprise value, observable market transactions, such as recent offers to purchase a company, recent transactions involving the purchase or sale of the equity securities of the company, or other liquidation events. The determined equity values may be discounted when the Company has a minority position, is subject to restrictions on resale, has specific concerns about the receptivity of the capital markets to a specific company at a certain time, or other comparable factors exist.

For equity and equity-related securities that are freely tradable and listed on a securities exchange or over-the-counter market, the Company fair values those securities at their last sale price on that exchange or over-the-counter market on the valuation date. If the security is listed on more than one exchange, the Company will use the price from the exchange that it considers to be the principal exchange on which the security is traded. Securities listed on the NASDAQ will be valued at the NASDAQ Official Closing Price, which may not necessarily represent the last sale price. If there has been no sale on such exchange or over-the-counter market on such day, the security will be valued at the mean between the last bid price and last ask price on such day.

An equity security of a publicly traded company acquired in a private placement transaction without registration is subject to restrictions on resale that can affect the security's liquidity and fair value. Such securities that are convertible into or otherwise

will become freely tradable will be valued based on the market value of the freely tradable security less an applicable discount. Generally, the discount will initially be equal to the discount at which the Company purchased the securities. To the extent that such securities are convertible or otherwise become freely tradable within a time frame that may be reasonably determined, an amortization schedule may be used to determine the discount.

The Board of Directors undertakes a multi-step valuation process each quarter in connection with determining the fair value of private investments. An independent valuation firm has been engaged by the Board of Directors to provide independent, third-party valuation consulting services based on procedures that the Board of Directors has identified and may ask them to perform from time to time on all or a selection of private investments as determined by the Board of Directors. The multi-step valuation process is specific to the level of assurance that the Board of Directors requests from the independent valuation firm. For positive assurance, the process is as follows:

- The independent valuation firm prepares the preliminary valuations and the supporting analysis. At February 28, 2011, the independent valuation firm performed positive assurance valuation procedures on five portfolio companies comprising approximately 99.6 percent of the total fair value of restricted investments;
- The investment professionals of the Adviser review the preliminary valuations and supporting analyses, and consider and assess, as appropriate, any changes that may be required to the preliminary valuations;
- The Investment Committee of the Adviser reviews the preliminary valuations and supporting analyses, and considers and assesses, as appropriate, any changes that may be required to the preliminary valuations;
- . The Board of Directors assesses the valuations and ultimately determines the fair value of each investment in the Company's portfolio in good faith.

C. Interest and Fee Income — Interest income is recorded on the accrual basis to the extent that such amounts are expected to be collected. When investing in instruments with an original issue discount or payment-in-kind interest (in which case the Company chooses payment-in-kind in lieu of cash), the Company will accrue interest income during the life of the investment, even though the Company will not necessarily be receiving cash as the interest is accrued. Fee income will include fees, if any, for due diligence, structuring, commitment and facility fees, transaction services consulting services and management services rendered to portfolio companies and other third parties. Commitment and facility fees generally are recognized as income over the life of the underlying loan, whereas due diligence, structuring, transaction service, consulting and management service fees generally are recognized as income when services are rendered. For the three months ended February 28, 2011 and February 28, 2010, the Company received \$0 and \$10,392 in fee income, respectively.

D. Security Transactions and Investment Income — Security transactions are accounted for on the date the securities are purchased or sold (trade date). Realized gains and losses are reported on an identified cost basis. Distributions received from the Company's investments in limited partnerships and limited liability companies generally are comprised of ordinary income, capital gains and return of capital. The Company records investment income, capital gains and return of capital based on estimates made at the time such distributions are received. Such estimates are based on information available from each company and/or other industry sources. These estimates may subsequently be revised based on information received from the entities after their tax reporting periods are concluded, as the actual character of these distributions is not known until after the fiscal year end of the Company.

For the period from December 1, 2010 through February 28, 2011, the Company estimated the allocation of investment income and return of capital for the distributions received from its portfolio companies within the Statement of Operations. For this period, the Company has estimated approximately 58 percent as investment income and approximately 42 percent as return of capital.

E. Distributions to Stockholders — The amount of any quarterly distributions will be determined by the Board of Directors. Distributions to stockholders are recorded on the ex-dividend date. If the Company has outstanding leverage, it may not declare or pay distributions to its common stockholders if it does not meet asset coverage ratios required under the 1940 Act. The character of distributions made during the year may differ from their ultimate characterization for federal income tax purposes. For the year ended November 30, 2010 and the period ended February 28, 2011, the Company's distributions for book purposes were comprised of 100 percent return of capital. For the year ended November 30, 2010, the Company's distributions for tax purposes were comprised of 100 percent return of capital. The tax character of distributions paid to common stockholders in the current year will be determined subsequent to November 30, 2011.

F. Federal and State Income Taxation — The Company, as a corporation, is obligated to pay federal and state income tax on its taxable income. Currently, the highest regular marginal federal income tax rate for a corporation is 35 percent; however, the Company anticipates a marginal effective tax rate of 34 percent due to expectations of the level of taxable income relative to the federal graduated tax rates, including the tax rate anticipated when temporary differences reverse. The Company may be subject to a 20 percent federal alternative minimum tax on its federal alternative minimum tax able income to the extent that its alternative minimum tax exceeds its regular federal income tax.

The Company invests its assets primarily in limited partnerships or limited liability companies which are treated as partnerships for federal and state income tax purposes. As a limited partner, the Company reports its allocable share of taxable income in computing its own taxable income. The Company's tax expense or benefit is included in the Statement of Operations based on the component of income or gains (losses) to which such expense or benefit relates. Deferred income taxes reflect the net tax effects of temporary differences between the carrying amounts of assets and liabilities for financial reporting purposes and the amounts used for income tax purposes. A valuation allowance is recognized, if based on the weight of available evidence, it is more likely than not that some portion or all of the deferred income tax asset will not be realized.

G. Offering Costs — Offering costs related to the issuance of common stock are charged to additional paid-in capital when the stock is issued.

H. Indemnifications — Under the Company's organizational documents, its officers and directors are indemnified against certain liabilities arising out of the performance of their duties to the Company. In addition, in the normal course of business, the Company may enter into contracts that provide general indemnification to other parties. The Company's maximum exposure under these arrangements is unknown as this would involve future claims that may be made against the Company that have not yet occurred, and may not occur. However, the Company has not had prior claims or losses pursuant to these contracts and expects the risk of loss to be remote.

3. Concentration of Risk

The Company invests primarily in privately-held and micro-cap public companies in the U.S. energy infrastructure sector. The Company may, for defensive purposes, temporarily invest all or a significant portion of its assets in investment grade securities, short-term debt securities and cash or cash equivalents. To the extent the Company uses this strategy it may not achieve its investment objective.

4. Agreements

The Company has entered into an Investment Advisory Agreement with Tortoise Capital Advisors, L.L.C. Under the terms of the Investment Advisory Agreement, the Adviser is paid a fee consisting of a base management fee and an incentive fee. The base management fee is 0.375 percent (1.5 percent annualized) of the Company's average monthly Managed Assets, calculated and paid quarterly in arrears within thirty days of the end of each fiscal quarter. The term "Managed Assets" as used in the calculation of the management fee means total assets (including any assets purchased with or attributable to borrowed funds but excluding any net deferred tax asset) minus accrued liabilities other than (1) net deferred tax liabilities, (2) debt entered into for the purpose of leverage and (3) the aggregate liquidation preference of any outstanding preferred shares. The base management fee for any partial quarter is appropriately prorated.

On November 11, 2008, the Company entered into an Expense Reimbursement Agreement with the Adviser, for which the Adviser reimbursed the Company for certain expenses incurred beginning January 1, 2009 and ending December 31, 2009 in an amount equal to an annual rate of 0.25 percent of the Company's average monthly Managed Assets. On February 17, 2010, the Company entered into an Expense Reimbursement Agreement with the Adviser under which the Adviser reimbursed the Company for certain expenses incurred beginning January 1, 2010 and ending December 31, 2010 in an amount equal to an annual rate of 0.25 percent of the Company's average monthly Managed Assets. On August 9, 2010, the Company entered into an Amended Expense Reimbursement Agreement with the Adviser under which the Adviser reimbursed the Company for certain expenses incurred beginning June 1, 2010 and ending December 31, 2010 in an amount equal to an annual rate of 0.50 percent of the Company's average monthly Managed Assets. On November 8, 2010, the Company entered into an Expense Reimbursement Agreement with the Adviser under which the Adviser will reimburse the Company for certain expenses incurred beginning January 1, 2011 and ending December 31, 2011 in an amount equal to an annual rate of 0.50 percent of the Company's average monthly Managed Assets. During the three months ended February 28, 2011 and February 28, 2010, the Adviser reimbursed the Company \$117,340 and \$51,654, respectively.

The incentive fee consists of two parts. The first part, the investment income fee, is equal to 15 percent of the excess, if any, of the Company's Net Investment Income for the fiscal quarter over a quarterly hurdle rate equal to 2 percent (8 percent annualized), and multiplied, in either case, by the Company's average monthly Net Assets for the quarter. "Net Assets" means the Managed Assets less deferred taxes, debt entered into for the purposes of leverage and the aggregate liquidation preference of any outstanding preferred shares. "Net Investment Income" means interest income (including accrued interest that we have not yet received in cash), dividend and distribution income from equity investments (but excluding that portion of cash distributions that are treated as a return of capital), and any other income (including any fees such as commitment, origination, syndication, structuring, diligence, monitoring, and consulting fees or other fees that the Company is entitled to receive from portfolio companies) accrued during the fiscal quarter, minus the Company's operating expenses for such quarter (including the base management fee, expense reimbursements payable pursuant to the Investment Advisory Agreement, any interest expense, any accrued income taxes related to net investment income, and distributions paid on issued and outstanding preferred stock, if any, but excluding the incentive fee payable). Net Investment Income also includes, in the case of investments with a deferred interest or income feature (such as

original issue discount, debt or equity instruments with a payment-in-kind feature, and zero coupon securities), accrued income that the Company has not yet received in cash. Net Investment Income does not include any realized capital gains, realized capital losses, or unrealized capital appreciation or depreciation. The investment income fee is calculated and payable quarterly in arrears within thirty (30) days of the end of each fiscal quarter. The investment income fee calculation is adjusted appropriately on the basis of the number of calendar days in the first fiscal quarter the fee accrues or the fiscal quarter during which the Agreement is in effect in the event of termination of the Agreement during any fiscal quarter. During the three months ended February 28, 2011 and February 28, 2010, the Company accrued no investment income fees.

The second part of the incentive fee payable to the Adviser, the capital gain incentive fee, is equal to: (A) 15 percent of (i) the Company's net realized capital gains (realized capital gains less realized capital losses) on a cumulative basis from inception to the end of each fiscal year, less (ii) any unrealized capital depreciation at the end of such fiscal year, less (B) the aggregate amount of all capital gain fees paid to the Adviser in prior fiscal years. The capital gain incentive fee is calculated and payable annually within thirty (30) days of the end of each fiscal year. In the event the Investment Advisory Agreement is terminated, the capital gain incentive fee calculation shall be undertaken as of, and any resulting capital gain incentive fee shall be paid within thirty (30) days of the date of termination. The Adviser may, from time to time, waive or defer all or any part of the compensation described in the Investment Advisory Agreement.

The calculation of the capital gain incentive fee does not include any capital gains that result from that portion of any scheduled periodic distributions made possible by the normally recurring cash flow from the operations of portfolio companies ("Expected Distributions") that are characterized by the Company as return of capital for U.S. generally accepted accounting principles purposes. In that regard, any such return of capital will not be treated as a decrease in the cost basis of an investment for purposes of calculating the capital gain incentive fee. This does not apply to any portion of any distribution from a portfolio company that is not an Expected Distribution. Realized capital gains on a security will be calculated as the excess of the net amount realized from the sale or other disposition of such security over the adjusted cost basis for the security. Realized capital losses on a security will be calculated as the amount by which the net amount realized from the sale or other disposition of such security is less than the adjusted cost basis of such security. Unrealized capital depreciation on a security will be calculated as the amount by which the Company's adjusted cost basis of such security at the end of a fiscal year.

The payable for capital gain incentive fees is a result of the increase or decrease in the fair value of investments and realized gains or losses from investments. For the three months ended February 28, 2011 and February 28, 2010, the Company accrued no capital gain incentive fees. Pursuant to the Investment Advisory Agreement, the capital gain incentive fee is paid annually only if there are realization events and only if the calculation defined in the agreement results in an amount due. No capital gain incentive fees have been paid since the commencement of operations.

U.S. Bancorp Fund Services, LLC serves as the Company's fund accounting services provider. The Company pays the provider a monthly fee computed at an annual rate of \$24,000 on the first \$50,000,000 of the Company's Net Assets, 0.0125 percent on the next \$200,000,000 of Net Assets, 0.0075 percent on the next \$250,000,000 of Net Assets and 0.0025 percent on the balance of the Company's Net Assets.

The Adviser serves as the Company's administrator. The Company paid the administrator a fee equal to an annual rate of 0.07 percent of aggregate average daily Managed Assets up to and including \$150,000,000, 0.06 percent of aggregate average daily Managed Assets on the next \$100,000,000, 0.05 percent of aggregate average daily Managed Assets on the next \$250,000,000, and 0.02 percent on the balance thru November 30, 2010. On December 1, 2010, the Company entered into an Amended Administration Agreement with the administrator that decreased the fee to an amount equal to an annual rate of 0.04 percent of aggregate average daily Managed Assets, with a minimum annual fee of \$30,000. This fee is calculated and accrued daily and paid quarterly in arrears.

Computershare Trust Company, N.A. serves as the Company's transfer agent and registrar and Computershare Inc. serves as the Company's dividend paying agent and agent for the automatic dividend reinvestment plan.

U.S. Bank, N.A. serves as the Company's custodian. The Company pays the custodian a monthly fee computed at an annual rate of 0.004 percent of the Company's portfolio assets, plus portfolio transaction fees.

5. Income Taxes

Deferred income taxes reflect the net tax effect of temporary differences between the carrying amount of assets and liabilities for financial reporting and tax purposes. Components of the Company's deferred tax assets and liabilities as of February 28, 2011 and November 30, 2010 are as follows:

	February 28, 2011	November 30, 2010
Deferred tax assets:		
Organization costs	\$ (20,700)	\$ (21,231)
Capital loss carryforwards	(4,134,627)	(4,268,529)
Net operating loss carryforwards	(6,632,064)	(6,343,988)
AMT and State of Kansas credit	(5,039)	(5,039)
Valuation allowance	_	558,533
	(10,792,430)	(10,080,254)
Deferred tax liabilities:		
Basis reduction of investment in partnerships	1,124,205	783,156
Net unrealized gain on investment securities	8,749,220	8,640,355
	9,873,425	9,423,511
Total net deferred tax asset	\$ (919,005)	\$ (656,743)

At February 28, 2011, a valuation allowance on deferred tax assets was not deemed necessary because the Company believes it is more likely than not that there is an ability to realize its deferred tax assets through future taxable income of the appropriate character. Any adjustments to such estimates will be made in the period such determination is made. The Company's policy is to record interest and penalties on uncertain tax positions as part of tax expense. As of February 28, 2011, the Company had no uncertain tax positions and no interest or penalties were accrued. Tax years subsequent to the year ending November 30, 2006 remain open to examination by federal and state tax authorities.

Total income tax expense (benefit) differs from the amount computed by applying the federal statutory income tax rate of 34 percent to net investment income and realized and unrealized gains on investments before taxes as follows:

	F	or the three	I	For the three
	n	onths ended	n	nonths ended
	Feb	ruary 28, 2011	Feb	oruary 28, 2010
Application of statutory income tax rate	\$	289,343	\$	1,612,679
State income taxes, net of federal taxes		15,488		166,485
Other		(8,560)		_
Change in deferred tax valuation allowance		(558,533)		(1,053,513)
Total income tax expense (benefit)	\$	(262,262)	\$	725,651

The provision for income taxes is computed by applying the federal statutory rate plus a blended state income tax rate. The components of income tax include the following for the periods presented:

	For the three	F	For the three	
	months ended	m	nonths ended	
	February 28, 2011	Feb	February 28, 2010	
Deferred tax expense (benefit)				
Federal	\$ (248,937)	\$	657,748	
State	(13,325)		67,903	
Total deferred expense (benefit)	\$ (262,262)	\$	725,651	

The deferred income tax expense (benefit) for the three months ended February 28, 2011 and February 28, 2010 includes the impact of the change in valuation allowance for such respective periods.

As of November 30, 2010, the Company had a net operating loss for federal income tax purposes of approximately \$17,798,000. The net operating loss may be carried forward for 20 years. If not utilized, this net operating loss will expire as follows: \$3,911,000, \$3,381,000, \$7,119,000 and \$3,387,000 in the years 2027, 2028, 2029, and 2030, respectively. As of November 30, 2010, the Company had a capital loss carryforward of approximately \$12,000,000 which may be carried forward for 5 years. If not utilized, this capital loss will expire in the year ending November 30, 2014. The amount of the deferred tax asset for these items at February 28, 2011 also includes amounts for the period from December 1, 2010 through February 28, 2011. For corporations, capital losses can only be used to offset capital gains and cannot be used to offset ordinary income. As of November 30, 2010, an alternative minimum tax credit of \$3,109 was available, which may be credited in the future against regular income tax. This credit may be carried forward indefinitely.

The aggregate cost of securities for federal income tax purposes and securities with unrealized appreciation and depreciation, were as follows:

	F	ebruary 28, 2011	No	vember 30, 2010
Aggregate cost for federal income tax purposes	\$	68,541,367	\$	68,894,462
Gross unrealized appreciation		29,901,880		32,072,976
Gross unrealized depreciation		(2,337,877)		(5,765,015)
Net unrealized appreciation	\$	27,564,003	\$	26,307,961

6. Fair Value of Financial Instruments

Various inputs are used in determining the fair value of the Company's investments. These inputs are summarized in the three broad levels listed below:

- Level 1 quoted prices in active markets for identical investments
- Level 2 other significant observable inputs (including quoted prices for similar investments, market corroborated inputs, etc.)
- Level 3 significant unobservable inputs (including the Company's own assumptions in determining the fair value of investments)

Valuation Techniques

In general, and where applicable, the Company uses readily available market quotations based upon the last updated sales price from the principal market to determine fair value. This pricing methodology applies to the Company's Level 1 investments.

An equity security of a publicly traded company acquired in a private placement transaction without registration under the Securities Act of 1933, as amended (the "1933 Act"), is subject to restrictions on resale that can affect the security's fair value. If such a security is convertible into publicly-traded common shares, the security generally will be valued at the common share market price adjusted by a percentage discount due to the restrictions. This pricing methodology applies to the Company's Level 2 investments.

For private company investments, value is often realized through a liquidity event of the entire company. Therefore, the value of the company as a whole (enterprise value) at the reporting date often provides the best evidence of the value of the investment and is the initial step for valuing the Company's privately issued securities. For any one company, enterprise value may best be expressed as a range of fair values, from which a single estimate of fair value will be derived. In determining the enterprise value of a portfolio company, the Company prepares an analysis consisting of traditional valuation methodologies including market and income approaches. The Company considers some or all of the traditional valuation methods based on the individual circumstances of the portfolio company in order to derive its estimate of enterprise value. This pricing methodology applies to the Company's Level 3 investments.

The inputs or methodology used for valuing securities are not necessarily an indication of the risk associated with investing in those securities. The following tables provide the fair value measurements of applicable Company assets and liabilities by level within the fair value hierarchy as of February 28, 2011 and November 30, 2010. These assets are measured on a recurring basis.

February 28, 2011

• /	Fair Value at				
Description	February 28, 2011	Level 1	Le	evel 2	Level 3
Equity Investments	\$ 90,106,701	\$ 21,805,885	\$	_	\$ 68,300,816
Debt Investments	3,800,000	_		_	3,800,000
Short-Term Investments	2,198,669	2,198,669		_	_
Total Investments	\$ 96,105,370	\$ 24,004,554	\$		\$ 72,100,816

November 30, 2010

	Fair Value at			
Description	November 30, 2010	Level 1	Level 2	Level 3
Equity Investments	\$ 89,936,230	\$ 20,806,821	\$ — \$	69,129,409
Debt Investments	3,800,000	_	_	3,800,000
Short-Term Investments	1,466,193	1,466,193	_	_
Total Investments	\$ 95,202,423	\$ 22,273,014	s — s	72,929,409

The changes for all Level 3 assets measured at fair value on a recurring basis using significant unobservable inputs for the three months ended February 28, 2011 and February 28, 2010, are as follows:

		ree months ended		ree months ended
	F	ebruary 28, 2011	Fe	bruary 28, 2010
Fair value beginning balance	\$	72,929,409	\$	77,146,520
Total realized and unrealized gains (losses) included in net increase (decrease)				
in net assets applicable to common stockholders		(744,584)		3,998,840
Purchases		400,000		750,000
Sales		(400,000)		(9,839,190)
Return of capital adjustments impacting cost basis of securities		(84,009)		(974,812)
Fair value ending balance	\$	72,100,816	\$	71,081,358
The amount of total gains (losses) for the period included in net increase (decrease)				
in net assets applicable to common stockholders attributable to the change in				
unrealized gains (losses) relating to assets still held at the reporting date	\$	(744,584)	\$	3,981,482

There were no transfers between levels for the three months ended February 28, 2011 and February 28, 2010, respectively.

7. Restricted Securities

Certain of the Company's investments are restricted and are valued as determined in accordance with procedures established by the Board of Directors and more fully described in Note 2. The following tables show the equity interest, number of units or principal amount, the acquisition date(s), acquisition cost (excluding return of capital adjustments), fair value, fair value per unit of such securities and fair value as percent of net assets applicable to common stockholders as of February 28, 2011 and November 30, 2010.

February 28, 2011

		Equity Inte	erest,				Fair	Fair Value as
		Units o	r A	equisition	Acquisition	Fair	Value	Percent of
Investment Se	curity	Principal Ar	mount	Date(s)	Cost	Value	Per Unit	Net Assets
High Sierra Energy, LP	Common Units	1,042,6	585	11/2/06-	\$ 24,828,836	\$ 20,520,033	\$ 19.68	21.5%
			1	11/15/08				
High Sierra Energy GP, LLC	Equity Interest	2.	.37%	11/2/06-	2,015,969	53,486	N/A	0.1
				5/1/07				
International Resource	Class A Units	500,0	000	6/12/07	10,000,000	30,455,000	60.91	31.8
Partners LP								
LONESTAR Midstream	Class A Units	1,327,9	900	7/27/07-	2,149,269	216,000	0.16	0.2
Partners, LP(1)				4/2/08				
LSMP GP, LP(1)	GP LP Units	1	80	7/27/07-	120,046	39,000	216.67	0.0
				4/2/08				
Mowood, LLC	Equity Interest	1	00%	6/5/06-	1,000,000	4,936,367	N/A	5.2
				8/4/08				
	Subordinated Debt	\$ 3,800,0	000	7/28/10	3,800,000	3,800,000	N/A	4.0
VantaCore Partners LP	Common Units	933,4	130	5/21/07-	17,911,082	11,639,872	12.47	12.2
				8/4/08				
	Preferred Units	23,1	185	2/25/11	359,367	405,737	17.50	0.4
	Incentive Distribution	9	988	5/21/07-	143,936	35,321	35.75	0.0
	Rights			8/4/08				
					\$ 62,328,505	\$ 72,100,816		75.4%

⁽¹⁾ See Note 9 — Investment Transactions for additional information.

November 30, 2010

1101cmber 50, 2010							
		Equity Interest,				Fair	Fair Value as
		Units or	Acquisition	Acquisition	Fair	Value	Percent of
Investment Se	curity	Principal Amount	Date(s)	Cost	Value	Per Unit	Net Assets
High Sierra Energy, LP	Common Units	1,042,685	11/2/06- 11/15/08	\$ 24,828,836	\$ 20,666,009	\$ 19.82	21.6%
High Sierra Energy GP, LLC	Equity Interest	2.37%	11/2/06- 5/1/07	2,015,969	602,834	N/A	0.6
International Resource Partners LP	Class A Units	500,000	6/12/07	10,000,000	28,155,000	56.31	29.5
LONESTAR Midstream Partners, LP(1)	Class A Units	1,327,900	7/27/07- 4/2/08	2,149,269	208,000	0.16	0.2
LSMP GP, LP(1)	GP LP Units	180	7/27/07- 4/2/08	120,046	37,000	205.56	0.1
Mowood, LLC	Equity Interest	100%	6/5/06- 8/4/08	1,000,000	5,492,247	N/A	5.8
	Subordinated Debt	\$ 3,800,000	7/28/10	3,800,000	3,800,000	N/A	4.0
VantaCore Partners LP	Common Units	933,430	5/21/07- 8/4/08	18,270,449	13,814,764	14.80	14.5
	Incentive Distribution Rights	988	5/21/07- 8/4/08	143,936	153,555	155.42	0.1
				\$ 62,328,505	\$ 72,929,409		76.4%

 $^{(1) \}hspace{0.5cm} \textit{See Note 9} - \textit{Investment Transactions for additional information}.$

8. Investments in Affiliates and Control Entities

Investments representing 5 percent or more of the outstanding voting securities of a portfolio company result in that company being considered an affiliated company, as defined in the 1940 Act. Investments representing 25 percent or more of the outstanding voting securities of a portfolio company result in that company being considered a control company, as defined in the 1940 Act. The aggregate fair value of all securities of affiliates and controlled entities held by the Company as of February 28, 2011 amounted to \$72,047,330, representing 75.3 percent of net assets applicable to common stockholders. The aggregate fair value of all securities of affiliates and controlled entities held by the Company as of November 30, 2010 amounted to \$72,326,575, representing 75.7 percent of net assets applicable to common stockholders. A summary of affiliated transactions for each company which is or was an affiliate or controlled entity at February 28, 2011 or during the three months then ended and at November 30, 2010 or during the year then ended is as follows:

February 28, 2011

	Units/					Units/	
	Equity Interest/				Gross	Equity Interest/	
	Principal				Distributions	Principal	
	Balance	Gross	Gross	Realized Gain	or Interest	Balance	Fair Value
	11/30/10	Additions	Reductions	(Loss)	Received	2/28/11	2/28/11
High Sierra Energy, LP(1)	1,042,685	\$ —	\$ —	\$ —	\$ —	1,042,685	\$ 20,520,033
International Resource Partners LP	500,000	_	_	_	300,000	500,000	30,455,000
LONESTAR Midstream Partners, LP(1)(2)	1,327,900	_	_	_	_	1,327,900	216,000
LSMP GP, LP(1)(2)	180	_	_	_	_	180	39,000
Mowood, LLC Subordinated Debt	\$3,800,000	400,000	(400,000)	_	135,330	\$ 3,800,000	3,800,000
Mowood, LLC Equity Interest	100%	_	_	_	70,166	100%	4,936,367
VantaCore Partners LP Common Units	933,430	_	(359,367)	_	84,009	933,430	11,639,872
VantaCore Partners LP Preferred Units(1)	_	359,367	_	_	_	23,185	405,737
VantaCore Partners LP Incentive	988	_	_	_	_	988	35,321
Distribution Rights(1)							
		\$ 759,367	\$ (759,367)	\$ —	\$ 589,505		\$ 72,047,330

⁽¹⁾ Currently non-income producing.

November 30, 2010

	Units/					Units/	
	Equity Interest/				Gross	Equity Interest/	
	Principal				Distributions	Principal	
	Balance	Gross	Gross	Realized Gain	or Interest	Balance	Fair Value
	11/30/09	Additions	Reductions	(Loss)	Received	11/30/10	11/30/10
High Sierra Energy, LP(1)	1,042,685	\$ —	s —	s —	\$ 656,891	1,042,685	\$ 20,666,009
International Resource Partners LP	500,000	_	_	_	950,000	500,000	28,155,000
LONESTAR Midstream Partners, LP(1)(2)	1,327,900	_	(890,942)	87,585	_	1,327,900	208,000
LSMP GP, LP(1)(2)	180	_	(17,254)	(1,221)	_	180	37,000
Mowood, LLC Subordinated Debt	\$ 8,800,000	750,000	(5,750,000)	_	720,323	\$ 3,800,000	3,800,000
Mowood, LLC Equity Interest	99.5%	_	(5,528,403)	2,356,404	248,426	100%	5,492,247
Quest Midstream Partners, L.P.	1,216,881	_	(9,915,452)	(9,607,112)	_	_	_
VantaCore Partners LP Common Units	933,430	_	_	_	1,773,517	933,430	13,814,764
VantaCore Partners LP Incentive	988	_	_	_	_	988	153,555
Distribution Rights(1)							
		\$750,000	\$ (22,102,051)	\$ (7,164,344)	\$ 4,349,157		\$ 72,326,575

⁽¹⁾ Currently non-income producing.

9. Investment Transactions

For the three months ended February 28, 2011, the Company purchased (at cost) securities in the amount of \$7,970,756 and sold securities (at proceeds) in the amount of \$8,172,306 (excluding short-term debt securities). For the three months ended February 28, 2010, the Company purchased (at cost) securities in the amount of \$750,000 and sold securities (at proceeds) in the amount of \$9,839,190 (excluding short-term debt securities).

On July 17, 2008, LONESTAR Midstream Partners LP ("LONESTAR") closed a transaction with Penn Virginia Resource Partners, L.P. (NYSE: PVR) for the sale of its gas gathering and transportation assets. LONESTAR distributed substantially all of the initial sales proceeds to its limited partners but did not redeem partnership interests. The Company received a distribution of \$10,476,511 in cash, 468,001 newly issued unregistered common units of PVR, and 59,503 unregistered common units of Penn Virginia GP Holdings, L.P. (NYSE: PVG). On February 3, 2009, the Company received a distribution of 37,305 freely tradable common units of PVR and 4,743 freely tradable common units of PVG. On July 17, 2009, the Company received an additional

⁽²⁾ See Note 9 — Investment Transactions for additional information.

⁽²⁾ See Note 9 — Investment Transactions for additional information.

distribution of 37,304 freely tradable common units of PVR and 4,744 freely tradable common units of PVG. On December 31, 2009, the Company received a cash distribution from LONESTAR of \$804,387. For purposes of the capital gain incentive fee, the realized gain totals \$1,756,189. Pursuant to the Investment Advisory Agreement, the capital gain incentive fee is paid annually only if there are realization events and only if the calculation defined in the agreement results in an amount due. No capital gain incentive fees have been paid since the commencement of operations. There are also two potential future contingent payments due from LONESTAR which are based on the achievement of specific revenue targets by or before June 30, 2013. No payments are due if these revenue targets are not achieved. If received, the Company's expected portion would total approximately \$9,638,829, payable in cash or common units of PVR (at PVR's election). The fair value of the LONESTAR and LSMP GP, LP units, which totals \$255,000 as of February 28, 2011, is based on unobservable inputs related to the potential receipt of these future payments relative to the sales transaction.

10. Common Stock

The Company has 100,000,000 shares authorized and 9,146,506 shares outstanding at February 28, 2011 and November 30, 2010.

11. Warrants

At February 28, 2011 and November 30, 2010, the Company had 945,594 warrants issued and outstanding. The warrants became exercisable on February 7, 2007 (the closing date of the Company's initial public offering of common shares), subject to a lock-up period with respect to the underlying common shares. Each warrant entitles the holder to purchase one common share at the exercise price of \$15.00 per common share. Warrants were issued as separate instruments from the common shares and are permitted to be transferred independently from the common shares. The warrants have no voting rights and the common shares underlying the unexercised warrants will have no voting rights until such common shares are received upon exercise of the warrants. All warrants will currently expire on February 6, 2013.

12. Earnings Per Share

The following table sets forth the computation of basic and diluted earnings per share:

	_	or the three onths ended	_	or the three norths ended
	February 28, 2011		Feb	ruary 28, 2010
Net increase in net assets applicable to common stockholders				
resulting from operations	\$	1,113,270	\$	4,017,523
Basic and diluted weighted average shares(1)		9,146,506		9,078,090
Basic and diluted net increase in net assets applicable				
to common stockholders resulting from operations				
per common share	\$	0.12	\$	0.44

⁽¹⁾ Warrants to purchase shares of common stock at \$15.00 per share were outstanding during the periods reflected in the table above, but were not included in the computation of diluted earnings per share because the warrants' exercise price was greater than the average market value of the common shares and, therefore, the effect would be anti-dilutive.

13. Subsequent Events

On March 1, 2011, the Company paid a distribution in the amount of \$0.10 per common share, for a total of \$914,654. Of this total, the dividend reinvestment amounted to \$88,196.

The Company performed an evaluation of subsequent events through the date the financial statements were issued and determined that no additional items require recognition or disclosure.

ADDITIONAL INFORMATION (Unaudited)

Director and Officer Compensation

The Company does not compensate any of its directors who are "interested persons" (as defined in Section 2 (a) (19) of the 1940 Act) or any of its officers. For the three months ended February 28, 2011, the aggregate compensation paid by the Company to the independent directors was \$21,000. The Company did not pay any special compensation to any of its directors or officers.

Forward-Looking Statements

This report contains "forward-looking statements." By their nature, all forward-looking statements involve risk and uncertainties, and actual results could differ materially from those contemplated by the forward-looking statements.

Certifications

The Company's Chief Executive Officer submitted to the New York Stock Exchange the annual CEO certification as required by Section 303A.12(a) of the NYSE Listed Company Manual.

The Company has filed with the SEC the certification of its Chief Executive Officer and Chief Financial Officer required by Section 302 of the Sarbanes-Oxley Act.

Proxy Voting Policies

A description of the policies and procedures that the Company uses to determine how to vote proxies relating to portfolio securities owned by the Company is available to stockholders (i) without charge, upon request by calling the Company at (913) 981-1020 or toll-free at (866) 362-9331 and on the Company's Web site at www.tortoiseadvisors.com/tto.cfm; and (ii) on the SEC's Web site at www.sec.gov.

Privacy Policy

The Company is committed to maintaining the privacy of its stockholders and safeguarding their non-public personal information. The following information is provided to help you understand what personal information the Company collects, how the Company protects that information and why, in certain cases, the Company may share information with select other parties.

Generally, the Company does not receive any non-public personal information relating to its stockholders, although certain non-public personal information of its stockholders may become available to the Company. The Company does not disclose any non-public personal information about its stockholders or a former stockholder to anyone, except as required by law or as is necessary in order to service stockholder accounts (for example, to a transfer agent).

The Company restricts access to non-public personal information about its stockholders to employees of its Adviser with a legitimate business need for the information. The Company maintains physical, electronic and procedural safeguards designed to protect the non-public personal information of its stockholders.

ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

Statements contained herein, other than historical facts, may constitute "forward-looking statements." These statements may relate to, among other things, future events or our future performance or financial condition. In some cases, you can identify forward-looking statements by terminology such as "may," "might," "believe," "will," "provided," "anticipate," "future," "could," "growth," "plan," "intend," "expect," "should," "would," "if," "seek," "possible," "potential," "likely" or the negative of such terms or comparable terminology. These forward-looking statements involve known and unknown risks, uncertainties and other factors that may cause our actual results, levels of activity, performance or achievements to be materially different from any anticipated results, levels of activity, performance or achievements expressed or implied by such forward-looking statements. For a discussion of factors that could cause our actual results to differ from forward-looking statements contained herein, please see the discussion under the heading "Risk Factors" in Part I, Item 1A. of our most recent Annual Report filed on Form 10-K.

We may experience fluctuations in our operating results due to a number of factors, including the return on our equity investments, the interest rates payable on our debt investments, the default rates on such investments, the level of our expenses, variations in and the timing of the recognition of realized and unrealized gains or losses, the degree to which we encounter competition in our markets and general economic conditions. As a result of these factors, results for any period should not be relied upon as being indicative of performance in future periods.

Overview

We have elected to be regulated as a BDC and we are classified as a non-diversified closed-end management investment company under the 1940 Act. As a BDC, we are subject to numerous regulations and restrictions. Unlike most investment companies, we are taxed as a general business corporation under the Code.

We have invested primarily in privately-held and micro-cap public companies operating in the U.S. energy infrastructure sector. We have invested in companies in the energy infrastructure sector that we expect to produce stable cash flows as a result of their fee-based revenues and proactive hedging programs, which help to limit direct commodity price risk.

Performance Review and Investment Outlook

Our net asset value was \$10.46 as of February 28, 2011, compared to \$10.44 at November 30, 2010. Total investment return, based on net asset value and assuming reinvestment of distributions, was approximately 1.4 percent for the three months ended February 28, 2011. Our stock price also increased this quarter, closing at \$8.50 on February 28, 2011 compared to \$7.28 on November 30, 2010. Total investment return based on market value and assuming reinvestment of distributions, was approximately 18.1 percent for the three months ended February 28, 2011. The fair value of our investment portfolio, excluding short-term investments at February 28, 2011, was approximately \$93.9 million, with approximately 77 percent of the portfolio in private investments totaling \$72.1 million and approximately 23 percent in publicly-traded investments totaling \$21.8 million. Our portfolio is diversified among approximately 51 percent midstream and downstream investments, 4 percent upstream, and 45 percent in aggregates and coal.

This quarter's distribution of \$0.10 per share was in excess of our DCF for the quarter. Assuming the sale of International Resource Partners LP ("IRP") closes in the expected timeframe, we anticipate paying a distribution of not less than \$0.10 per share, per quarter, for the remainder of fiscal 2011. In order to sustain this distribution level for 2011, we may elect to include a small portion of the proceeds from the IRP sale, depending on the operating performance of selected portfolio companies and their ability to return to near historic distribution levels.

On January 31, 2011, we filed a proxy statement with the SEC with respect to our annual meeting. Included within the proxy is a proposal unanimously approved by our Board of Directors to withdraw our election to be treated as a BDC as defined in the 1940 Act.

International Resource Partners LP ("IRP")

IRP, headquartered in Charleston, West Virginia, is a fully integrated coal company, controlling assets across the value chain, including production, logistics and marketing assets. We hold board of director's observation rights for IRP.

The fair value of IRP increased approximately \$2.3 million this quarter. IRP increased its quarterly distribution this quarter, from \$0.55 per unit last quarter to \$0.60 per unit this quarter. On March 6, 2011, IRP entered into a definitive agreement to sell its partnership interests to James River Coal Company (NASDAQ: JRCC). This transaction is expected to close in the first half of 2011 subject to completion of various closing conditions. Our portion of the initial sales proceeds is expected to be approximately \$31.1 million in cash, with an additional \$2.1 million held in escrow for a period of up to 14 months. The release of escrow funds is contingent upon a number of factors, including coal reserves and resolution of outstanding regulatory items. We anticipate

investing the proceeds of the sale of IRP in publicly traded MLPs and cash equivalents, pending being deployed in new qualifying investments and subject to limitations on publicly traded investments in the BDC structure.

Mowood, LLC ("Mowood")

Mowood is the holding company of Omega Pipeline, LLC ("Omega"). Omega is a natural gas local distribution company located on the Fort Leonard Wood military installation in south-central Missouri. Omega serves the natural gas needs of Fort Leonard Wood and other customers in the surrounding area. We hold 100 percent of the equity interests in Mowood and hold a seat on its board of directors.

The fair value of Mowood decreased approximately \$0.6 million this quarter, generally due to a decrease in the probability of receiving future contingent payments related to the Timberline Energy, LLC sale. We provide a revolving line of credit to Mowood with a maximum principal balance of \$5.3 million. The line of credit provides Mowood flexibility related to seasonal fluctuations in working capital. At February 28, 2011, the principal balance outstanding was \$3.8 million. Mowood's subsidiary, Omega, continues to perform at budget. On December 31, 2010, a tornado touched down on Fort Leonard Wood, destroying over 30 houses. Omega's pipeline system was relatively undamaged by the tornado and the financial effects of the tornado are not expected to be significant. Fort Leonard Wood continues to grow, and Omega expects that construction revenues will bolster its performance in 2011.

VantaCore Partners LP ("VantaCore")

VantaCore was formed to acquire companies in the aggregate industry and currently owns a quarry and asphalt plant in Clarksville, Tennessee and sand and gravel operations located near Baton Rouge, Louisiana. We hold a seat on VantaCore's board of directors.

The fair value of VantaCore decreased approximately \$1.9 million this quarter. VantaCore was unable to meet its minimum quarterly distribution ("MQD") of \$0.475 per unit for its quarter ended December 31, 2010. Common and preferred unitholders elected to receive their distributions as a combination of \$0.09 in cash and the remainder in newly issued preferred units. We received 23,185 preferred units in addition to the \$0.09 in cash per common unit. VantaCore reported year-to-date EBITDA through December 31, 2010 below budget. The Southern Aggregates property in Louisiana continues to fall short of budget but VantaCore is optimistic that signs of improvement in building industry in the area, along with cost-cutting measures imposed by management, will help improve their performance in 2011. Results from operations in Clarksville, Tennessee were strong, but 2011 margins are expected to come under pressure as new competition has entered the market and several larger construction projects are nearing completion.

High Sierra Energy, LP ("High Sierra")

High Sierra is a holding company with diversified midstream energy assets focused on the transportation, storage, and marketing of hydrocarbons and the processing of oilfield water. The company's businesses include a natural gas liquids logistics, transportation and marketing business operating throughout the lower 48 states, a natural gas storage facility in Mississippi, an ethanol terminal in Nevada, crude oil and natural gas liquids trucking businesses in Kansas and Colorado, crude oil gathering, transportation and marketing services, primarily focused in the Mid-Continent, Western and Gulf Coast regions, water treatment, transportation and disposal businesses serving oil and gas producers in Wyoming and Oklahoma, and two asphalt processing, packaging and distribution terminals in Florida. We hold board of directors' observation rights for High Sierra.

The fair value of High Sierra decreased slightly during the quarter. High Sierra has not made cash distributions to its LP and GP unit holders for the last four consecutive quarters. On March 16, 2011, High Sierra closed on a new \$215 million, 3-year committed senior secured credit facility led by BNP Paribas. The new credit facility is expected to provide a level of stability and flexibility to meet the needs of the partnership's existing operations and provide a base of working capital to take advantage of market opportunities. The new facility also financed the buyout of High Sierra's minority partners at Anticline Disposal, resulting in High Sierra owning 100 percent of the membership interests of that business. The closing represents a very positive event for the partnership and we believe this, along with improved operations, should allow High Sierra to return to paying cash distributions.

Recent Developments

On March 1, 2011, the Company paid a distribution in the amount of \$0.10 per common share, for a total of \$914,654. Of this total, the dividend reinvestment amounted to \$88,196.

Portfolio Company Monitoring

Our Adviser monitors each portfolio company to determine progress relative to meeting the company's business plan and to assess the company's strategic and tactical courses of action. This monitoring may be accomplished by attendance at board of directors meetings, ad hoc communications with company management, the review of periodic operating and financial reports, an analysis of relevant reserve information and capital expenditure plans, and periodic consultations with engineers, geologists, and other experts. The performance of each private portfolio company is also periodically compared to performance of similarly sized companies with comparable assets and businesses to assess performance relative to peers. Our Adviser's monitoring activities are expected to provide it with information that will enable us to monitor compliance with existing covenants, to enhance our ability to

make qualified valuation decisions, and to assist our evaluation of the nature of the risks involved in each individual investment. In addition, these monitoring activities should enable our Adviser to diagnose and manage the common risk factors held by our total portfolio, such as sector concentration, exposure to a single financial sponsor, or sensitivity to a particular geography.

As part of the monitoring process, our Adviser continually assesses the risk profile of each of our private investments. We intend to disclose, as appropriate, those risk factors that we deem most relevant in assessing the risk of any particular investment. Such factors may include, but are not limited to, the investment's current cash distribution status, compliance with loan covenants, operating and financial performance, changes in the regulatory environment or other factors that we believe are useful in determining overall investment risk.

Results of Operations

Comparison of the Three Months Ended February 28, 2011 and February 28, 2010

Investment Income: Investment income totaled \$561,786 for the three months ended February 28, 2011. This represents a decrease of \$130,370 as compared to the three months ended February 28, 2010. The decrease in investment income is primarily due to a decrease in the total distributions from investments, offset by a decrease in the amount of such distributions characterized as return of capital.

Net Expenses: Net expenses totaled \$388,523 for the three months ended February 28, 2011. This represents a decrease of \$89,932 as compared to the three months ended February 28, 2010. The decrease is primarily related to an increase in the expense reimbursement from the Adviser and a decrease in interest expense resulting from the elimination of our outstanding leverage.

Distributable Cash Flow: Our portfolio generates cash flow to us from which we pay distributions to stockholders. When our Board of Directors determines the amount of any distribution we expect to pay our stockholders, it reviews distributable cash flow ("DCF"). DCF is distributions received from investments less our total expenses. The total distributions received from our investments include the amount received by us as cash distributions from equity investments, paid-in-kind distributions, and dividend and interest payments. Total expenses include current or anticipated operating expenses, leverage costs and current income taxes on our operating income. Total expenses do not include deferred income taxes or accrued capital gain incentive fees. We do not include in DCF the value of distributions received from portfolio companies which are paid in stock as a result of credit constraints, market dislocation or other similar issues.

We disclose DCF in order to provide supplemental information regarding our results of operations and to enhance our investors' overall understanding of our core financial performance and our prospects for the future. We believe that our investors benefit from seeing the results of DCF in addition to GAAP information. This non-GAAP information facilitates management's comparison of current results with historical results of our operations and with those of our peers. This information is not in accordance with, or an alternative to, GAAP and may not be comparable to similarly titled measures reported by other companies.

The following table represents DCF for the three months ended February 28, 2011 as compared to the three months ended February 28, 2010:

		Three Months Ended		Three
	N			Months Ended
Distributable Cash Flow	Feb	oruary 28, 2011	ary 28, 2011 Fe	
Total from Investments				
Distributions from investments	\$	731,990	\$	1,488,756
Distributions paid in stock		23,367		_
Interest income from investments		135,330		191,431
Dividends from money market mutual funds		190		217
Other income		_		10,392
Total from Investments		890,877		1,690,796
Operating Expenses Before Leverage Costs				
Advisory fees (net of expense reimbursement by Adviser)		234,680		258,268
Other operating expenses		153,843		174,568
Total Operating Expenses, before Leverage Costs		388,523		432,836
Distributable cash flow before leverage costs		502,354		1,257,960
Leverage Costs		_		45,619
Distributable Cash Flow	\$	502,354	\$	1,212,341
Distributions paid on common stock	\$	914,651	\$	1,180,152
Payout percentage for period(1)		182%		97%
DCF/GAAP Reconciliation				
Distributable Cash Flow	\$	502,354	\$	1,212,341
Adjustments to reconcile to Net Investment Income, before Income Taxes:				
Distributions paid in stock(2)		(23,367)		
Return of capital on distributions received from				
equity investments		(305,724)		(998,640)
Net Investment Income, before Income Taxes	\$	173,263	\$	213,701

⁽¹⁾ Distributions paid as a percentage of Distributable Cash Flow.

Distributions: The following table sets forth distributions for the three months ended February 28, 2011 as compared to the three months ended February 28, 2010.

Record Date	Payment Date	Payment Date Amount	
February 18, 2011	March 1, 2011	\$0.10	
February 19, 2010	March 1, 2010	\$0.13	

Net Investment Income: Net investment income after deferred taxes for the three months ended February 28, 2011 was \$112,943 as compared to \$181,007 for the three months ended February 28, 2010. The variance in net investment income is primarily related to decreases in investment income which were offset by decreases in net expenses during the current fiscal periods as described above.

Net Realized and Unrealized Gain: We had net unrealized appreciation (before deferred taxes) of \$303,923 for the three months ended February 28, 2011 as compared to \$2,941,305 for the three months ended February 28, 2010. We had net realized gains (before deferred taxes) of \$373,822 for the three months ended February 28, 2011, as compared to net realized gains (before deferred taxes) of \$1,588,168 for the three months ended February 28, 2010. Net realized gains for the three months ended February 28, 2011 were attributed to a realized gain on the sale of Abraxas Petroleum Corporation shares partially offset by a realized loss on the sale of PostRock Energy Corporation shares.

Liquidity and Capital Resources

We may raise additional capital to support our future growth through equity offerings, rights offerings, and issuances of senior securities to the extent permitted by the 1940 Act and subject to market conditions. We generally may not issue additional common shares at a price below our net asset value (net of any sales load (underwriting discount)) without first obtaining approval of our stockholders and Board of Directors.

²⁾ Distributions paid in stock for the three months ended February 28, 2011 were paid as part of normal operations and are included in DCF.

Contractual Obligations

We do not have any significant contractual payment obligations as of February 28, 2011.

Off-Balance Sheet Arrangements

We do not have any off-balance sheet arrangements that have or are reasonably likely to have a current or future effect on our financial condition, changes in financial condition, revenues or expenses, results of operations, liquidity, capital expenditures, or capital resources.

Borrowings

The Company did not have any borrowings outstanding during the three months ended February 28, 2011.

Critical Accounting Policies

The financial statements included in this report are based on the selection and application of critical accounting policies, which require management to make significant estimates and assumptions. Critical accounting policies are those that are both important to the presentation of our financial condition and results of operations and require management's most difficult, complex or subjective judgments. While our critical accounting policies are discussed below, Note 2 in the Notes to Financial Statements included in this report provides more detailed disclosure of all of our significant accounting policies.

Valuation of Portfolio Investments

We invest primarily in illiquid securities including debt and equity securities of privately-held companies. These investments generally are subject to restrictions on resale, have no established trading market and are fair valued on a quarterly basis. Because of the inherent uncertainty of valuation, the fair values of such investments, which are determined in accordance with procedures approved by our Board of Directors, may differ materially from the values that would have been used had a ready market existed for the investments.

Securities Transactions and Investment Income Recognition

Securities transactions are accounted for on the date the securities are purchased or sold (trade date). Realized gains and losses are reported on an identified cost basis. Distributions received from our equity investments generally are comprised of ordinary income, capital gains and return of capital from the portfolio company. We record investment income and returns of capital based on estimates made at the time such distributions are received. Such estimates are based on information available from each portfolio company and/or other industry sources. These estimates may subsequently be revised based on information received from the portfolio companies after their tax reporting periods are concluded, as the actual character of these distributions are not known until after our fiscal year end.

Federal and State Income Taxation

We, as a corporation, are obligated to pay federal and state income tax on our taxable income. Our tax expense or benefit is included in the Statement of Operations based on the component of income or gains (losses) to which such expense or benefit relates. Deferred income taxes reflect the net tax effects of temporary differences between the carrying amounts of assets and liabilities for financial reporting purposes and the amounts used for income tax purposes.

ITEM 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

Our business activities contain elements of market risk. We consider fluctuations in the value of our equity securities to bour principal market risk. There were no material changes to our market risk exposure at February 28, 2011 as compared to February 28, 2010.

We carry our investments at fair value, as determined by our Board of Directors. The fair value of securities is determined using readily available market quotations from the principal market if available. The fair value of securities that are not publicly traded or whose market price is not readily available is determined in good faith by our Board of Directors. Because there are no readily available market quotations for many of the investments in our portfolio, we value a large portion of our portfolio investments at fair value as determined in good faith by our Board of Directors under a valuation policy and a consistently applied valuation process. Due to the inherent uncertainty of determining the fair value of investments that do not have readily available market quotations, the fair value of our investments may differ significantly from the fair values that would have been used had a ready market quotation existed for such investments, and these differences could be material.

As of February 28, 2011, the fair value of our investment portfolio (excluding short-term investments) totaled \$93,906,701. We estimate that the impact of a 10 percent increase or decrease in the fair value of these investments, net of related deferred taxes, would increase or decrease net assets applicable to common stockholders by approximately \$6,026,932.

Debt investments in our portfolio may be based on floating or fixed rates. As of February 28, 2011, we had no floating rate debt investments outstanding.

We consider the management of risk essential to conducting our businesses. Accordingly, our risk management systems and procedures are designed to identify and analyze our risks, to set appropriate policies and limits and to continually monitor these risks and limits by means of reliable administrative and information systems and other policies and programs.

ITEM 4. CONTROLS AND PROCEDURES

Our management, with the participation of our Chief Executive Officer and Chief Financial Officer, has evaluated the effectiveness of our disclosure controls and procedures (as defined in Rules 13a-15(e) or 15d-15(e) of the Securities Exchange Act of 1934) as of the end of the period covered by this report as required by paragraph (b) of Rule 13a-15 or 15d-15 of the Securities Exchange Act of 1934. Based upon such evaluation, our Chief Executive Officer and Chief Financial Officer concluded that our disclosure controls and procedures were effective and provided reasonable assurance that information required to be disclosed by us in the reports we file or submit under the Securities Exchange Act of 1934 is recorded, processed, summarized and reported within the time periods specified in the SEC rules and forms, and that such information is accumulated and communicated to our management, including our Chief Executive Officer and Chief Financial Officer, as appropriate, to allow timely decisions regarding required disclosure.

There have been no changes in our internal control over financial reporting (identified in connection with the evaluation required by paragraph (d) of Rules 13a-15 or 15d-15 of the Securities Exchange Act of 1934) during the fiscal quarter ended February 28, 2011, that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

PART II — OTHER INFORMATION

ITEM 1. LEGAL PROCEEDINGS

We are not currently subject to any material legal proceeding, nor, to our knowledge, is any material legal proceeding threatened against us.

ITEM 1A. RISK FACTORS

In addition to the other information set forth in this report, you should carefully consider the factors discussed in Part I, "Item 1A. Risk Factors" in our Annual Report on Form 10-K for the fiscal year ended November 30, 2010, which could materially affect our business, financial condition or operating results. The risks described in our Annual Report on Form 10-K are not the only risks facing our Company. Additional risks and uncertainties not currently known to us or that we currently deem to be immaterial also may materially adversely affect our business, financial condition and/or operating results.

ITEM 2. UNREGISTERED SALES OF EQUITY SECURITIES AND USE OF PROCEEDS

We did not sell any securities during the three months ended February 28, 2011 that were not registered under the Securities Act of 1933.

We did not repurchase any of our common shares during the three months ended February 28, 2011.

ITEM 3. DEFAULTS UPON SENIOR SECURITIES

None

ITEM 4. (REMOVED AND RESERVED)

ITEM 5. OTHER INFORMATION

None

ITEM 6. EXHIBITS

F	Exhibit	ibit Description	
1	10.1	Purchase Agreement, dated as of March 6, 2011, by and between Lightfoot Capital Partners, LP, International Industries, Inc., International Resource Partners	
		LP LLC, Kayne Anderson Energy Development Company, and Tortoise Capital Resources Corporation, and James River Coal Company and International	
		Resource Partners GP LLC, as Agent, is filed herewith.	

- 31.1 Certification by Chief Executive Officer pursuant to Exchange Act Rule 13a-14(a), as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002, is filed herewith.
- 31.2 Certification by Chief Financial Officer pursuant to Exchange Act Rule 13a-14(a), as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002, is filed herewith.
- 32.1 Certification by Chief Executive Officer and Chief Financial Officer pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, is furnished herewith.

All other exhibits for which provision is made in the applicable regulations of the Securities and Exchange Commission are not required under the related instruction or are inapplicable and therefore have been omitted.

SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

TORTOISE CAPITAL RESOURCES CORPORATION

Date: April 7, 2011 By: /s/ Terry Matlack

Terry Matlack Authorized Officer and Chief Financial Officer (Principal Financial Officer)

27

PURCHASE AGREEMENT

BY AND BETWEEN

LIGHTFOOT CAPITAL PARTNERS, LP

INTERNATIONAL INDUSTRIES, INC.

INTERNATIONAL RESOURCE PARTNERS GP LLC

KAYNE ANDERSON ENERGY DEVELOPMENT COMPANY

AND

TORTOISE CAPITAL RESOURCES CORPORATION

AND

JAMES RIVER COAL COMPANY

AND

INTERNATIONAL RESOURCE PARTNERS GP LLC

AS AGENT

TABLE OF CONTENTS

ARTICLE	E I. DEFINITIONS	2
1.1	Certain Defined Terms	2
1.2	Other Interpretive Provisions	13
ARTICLE	E II. PURCHASE AND SALE	14
2.1	<u>Purchase Price</u>	14
2.2	The Closing	15
2.3	Deliveries at the Closing	15
2.4	Estimated Closing Adjustments	16
2.5	Closing Adjustments	17
2.6	Payment of the Indebtedness	18
2.7	Escrow Amount and Escrow Fund	19
2.8	Escrow Release	19
ARTICLE	E III. REPRESENTATIONS AND WARRANTIES OF	
IRP GP A	AND RESOURCE PARTNERS RELATED TO THE	
ACQUIRE	ED COMPANIES	22
3.1	<u>Organization</u>	22
3.2	Authorization; Enforceability	22
3.3	<u>Interests</u>	22
3.4	<u>Financial Statements</u>	23
3.5	Absence of Undisclosed Liabilities	23
3.6	No Conflicts or Approvals	24
3.7	Governmental Authorization	24
3.8	Compliance; Permits; Seller Bonds	24
3.9	Proceedings; Governmental Orders	26
3.10	Absence of Certain Changes	27
3.11	Tax Matters	28
3.12	Employee Benefits	30
3.13	Employees; Labor and Employee Relations	31
3.14	Intellectual Property	33
3.15	Contracts	33
3.16	Environmental Matters	35
3.17	<u>Insurance</u>	37
3.18	Personal Property Assets	37
3.19	Real Property	38
3.20	No Brokers' or Other Fees	41
3.21	Affiliate Transactions	41
3.22	<u>Customers and Suppliers</u>	41
3.23	Absence of Certain Business Practices	41
3.24	Full Disclosure	42

i

ARTICLE 1	IIIA. REPRESENTATIONS AND WARRANTIES OF SELLERS	42
3A.1	<u>Organization</u>	42
3A.2	Authorization; Enforceability	42
3A.3	No Conflicts or Approvals	42
3A.4	<u>Interests</u>	42
ARTICLE 1	IV. REPRESENTATIONS AND WARRANTIES OF BUYER	43
4.1	Organization	43
4.2	Authorization; Enforceability	43
4.3	No Conflicts or Approvals	43
4.4	Governmental Authorization	43
4.5	No Brokers' or Other Fees	44
4.6	Permit Blocking	44
	V. EXCLUSIVE REPRESENTATIONS AND WARRANTIES	44
5.1	WARRANTIES EXCLUSIVE	44
	VI. COVENANTS AND AGREEMENTS	45
6.1	Conduct of Business Prior to the Closing	45
6.2	Redemption Redemption	47
6.3	Access and Investigation	47
6.4	Required Approvals; Consents	47
6.5	<u>Further Actions</u>	48
6.6	<u>Financing</u>	48
6.7	<u>Further Assurances</u>	51
6.8	Purchase Price Deductions; Special Indemnity	51
6.9	Guarantees; Seller Bonds	52
6.10	<u>Permits</u>	53
6.11	<u>Confidentiality</u>	54
6.12	Tax Matters	54
6.13	Employee Plans	56
6.14	Antitrust Notification	56
6.15	WARN Act	57
6.16	<u>Schedules</u>	58
6.17	Release	58
6.18	<u>Employees</u>	59
_	VII. CONDITIONS PRECEDENT TO SELLERS'	
	IONS TO CLOSE	59
7.1	Representations and Warranties	59
7.2	<u>Performance</u>	60
7.3	Officer's Certificates	60
7.4	Governmental Approvals	60
7.5	HSR Act	60
7.6	AVS List	60
7.7	Seller Guarantees and Seller Bonds Arrangements	60

7.8	<u>Injunctions</u>	60
7.9	Closing Deliveries	60
	VIII. CONDITIONS PRECEDENT TO BUYER'S	
	IONS TO CLOSE	61
8.1	Representations and Warranties	61
8.2	Performance	61
8.3	Officer's Certificate	61
8.4	Governmental Approvals	61
8.5	Other Approvals	61
8.6	<u>Injunctions</u>	61
8.7	<u>Closing Deliveries</u>	61
8.8	Material Adverse Effect	61
8.9	<u>Permits</u>	62
8.10	Repayment of Indebtedness	62
8.11	Redemption	62
8.12	Environmental Reports	62
8.13	Shared Services Agreement	62
ARTICLE I	IX. TERMINATION	62
9.1	<u>Termination</u>	62
9.2	Procedure and Effect of Termination	63
ARTICLE X	X. INDEMNIFICATION	63
10.1	<u>Indemnification</u>	63
10.2	Treatment of Indemnification Payments	67
ARTICLE X	XI. MISCELLANEOUS	67
11.1	Agent	67
11.2	Fees and Expenses	68
11.3	Governing Law	68
11.4	Amendment	68
11.5	Assignment	68
11.6	Waiver	68
11.7	Notices	69
11.8	Complete Agreement	69
11.9	Counterparts	69
11.10	Publicity	69
11.11	Headings	70
11.12	Severability	70
11.13	Third Parties	70
11.14	Jurisdiction; Service of Process	70
11.15	Further Assurances	70
11.16	Time of the Essence	70
11.17	Construction	70
11.18	Specific Performance	71

11.19	WAIVER OF JURY TRIAL		71
11.20	Consent of Limited Partners		71
Exhibit A I	Escrow Agreement		A-1
		<u>:</u>	

PURCHASE AGREEMENT

This PURCHASE AGREEMENT, dated as of March 6, 2011, is entered into by and between LIGHTFOOT CAPITAL PARTNERS, LP, a Delaware limited partnership ("Lightfoot"), INTERNATIONAL INDUSTRIES, INC., a West Virginia corporation ("International Industries"), INTERNATIONAL RESOURCE PARTNERS GP LLC, a Delaware limited liability company ("IRP GP"), KAYNE ANDERSON ENERGY DEVELOPMENT COMPANY, a Maryland limited partnership ("KED"), and TORTOISE CAPITAL RESOURCES CORPORATION, a Maryland corporation, ("Tortoise", and Lightfoot, International Industries, IRP GP and KED each, a "Seller" and collectively "Sellers"), JAMES RIVER COAL COMPANY, a Virginia corporation ("Buyer"), and IRP GP, solely to accept the duties herein as "Agent," and in its individual capacity as a Seller.

WHEREAS, Lightfoot, International, IRP GP, KED and Tortoise own all the outstanding limited partnership units and general partnership units of International Resource Partners LP, a Delaware limited partnership ("Resource Partners") as set forth on Schedule 3.3;

WHEREAS, certain individuals own limited partnership units of Resource Partners as set forth on Schedule 3.3;

WHEREAS, on or before the Closing Date, Sellers will own all of the outstanding limited partnership units and general partnership units of Resource Partners;

WHEREAS, Resource Partners owns all of the issued and outstanding membership interests of International Resources Holdings I LLC, a Delaware limited liability company ("Holdings I") through itself and IRP WV Corp., a Delaware corporation ("IRP WV");

WHEREAS, Holdings I owns all of the issued and outstanding membership interests of International Resources Holdings II LLC, a Delaware limited liability company ("Holdings II");

WHEREAS, Holdings II owns all of the issued and outstanding membership interests of International Resources, LLC, a West Virginia limited liability company ("Resources"), and Resources in turn owns all of the issued and outstanding membership interests in Hampden Coal Company, LLC, a West Virginia limited liability company ("Hampden"), Rockhouse Creek Development, LLC, a West Virginia limited liability company ("Rockhouse"), Chafin Branch Coal Company, LLC, a West Virginia limited liability company ("Chafin"), Snap Creek Mining, LLC, a West Virginia limited liability company ("Snap Creek"), Logan & Kanawha Coal Co., LLC, a West Virginia limited liability company ("L&K"), and IRP Kentucky LLC, a Kentucky limited liability company ("IRP Kentucky") and IRP Kentucky in turn owns all of the issued and outstanding membership interests in each of Laurel Mountain Resources LLC, a Kentucky limited liability company ("LMR") and Buck Branch Resources LLC, a Kentucky limited liability company ("Buck Branch"); Buck Branch together with Hampden, Rockhouse, Chafin, Snap Creek, L&K, IRP Kentucky, and LMR, collectively, the "Companies"; and the Companies together with Resource Partners, IRP WV, Holdings I, Holdings II and Resources, collectively, the "Acquired Companies");

WHEREAS, Resource Partners through the Acquired Companies, owns certain coal mining assets including permits, coal leases, fee mineral properties, equipment, buildings, preparation facilities, mine areas and complexes, fixtures and other assets related to the business of coal mining ("Mining Assets"); and

WHEREAS, Sellers desire to sell and Buyer desires to purchase all of the Partnership Interests, as defined herein.

NOW, THEREFORE, in consideration of the premises and the representations, warranties, covenants and agreements herein contained and intending to be legally bound hereby, the parties hereto hereby agree as follows:

ARTICLE I

DEFINITIONS

- 1.1 Certain Defined Terms. As used in this Agreement, the following terms shall have the following meanings:
- "Acquired Companies" has the meaning set forth in the recitals to this Agreement.
- "Accounting Principles" has the meaning set forth in Section 2.4(a).
- "Additional Reserves" has the meaning set forth in Section 2.8(b)(i).
- "Adjustment Payment" has the meaning set forth in Section 2.5(f).
- "Affiliate" means, with respect to any specified Person, any other Person that directly, or indirectly through one or more intermediaries, controls, is controlled by, or is under common control with, such specified Person.
- "Affiliated Group" means any affiliated group within the meaning of Section 1504(a) of the Code or any similar group defined under a similar provision of state, local or foreign tax law.
 - "Agent" has the meaning set forth in Section 11.1
 - "Agreement" means this Purchase Agreement (including the Exhibits and Schedules), as amended, modified or supplemented from time to time.
 - "Alternative Financing" has the meaning set forth in Section 6.6(d).
 - "Antitrust Authority" has the meaning set forth in Section 6.14(a).
 - "Antitrust Laws" has the meaning set forth in Section 6.14(a).
 - "Applicable Rate" has the meaning set forth in Section 2.5(f).
 - "Applications" has the meaning set forth in Section 3.8(c).

- "Auditor" means Deloitte LLP, or such other accounting firm as the parties shall mutually agree to select.
- "Audited Financials" has the meaning set forth in Section 3.4(a).
- "Benefit Plan" has the meaning set forth in Section 3.12(a).
- "Bonuses" has the meaning set forth in Section 6.8(b).
- "Buck Branch" has the meaning set forth in the recitals to this Agreement.
- "Business Day" means any day that is not a Saturday, a Sunday or other day on which banks are required or authorized by Law to be closed in the City of Charleston, West Virginia and the City of New York, New York.
 - "Buyer" has the meaning set forth in the first sentence of this Agreement.
- "Buyer Indemnified Taxes" means any and all Taxes together with any costs, expenses or damages (including court and administrative costs and reasonable legal fees and expenses incurred in investigating and preparing for any audit, examination, litigation or other judicial or administrative proceeding) arising out of, in connection with or incident to the determination, assessment or collection of such Taxes (a) imposed on any Acquired Company, or for which it may otherwise be liable, with respect to (i) any Tax period ending on or prior to the Closing Date or (ii) the portion of any Straddle Period (determined in accordance with Section 6.12(a)) ending on the Closing Date, (b) arising out of, in connection with, or related to, a breach of any representation or warranty set forth in Section 3.11 or covenants set forth in Section 6.12, (c) of any member of an affiliated, consolidated, combined or unitary group of which any Acquired Company (or any predecessor thereof) is or was a member on or prior to the Closing Date by reason of Treasury Regulation Section 1.1502-6(a) or any analogous or similar state or local law or (d) of any other Person for which any Acquired Company is or has been liable as a transferee or successor, by contract or otherwise; provided, however, that any such Tax shall not be a Buyer Indemnified Tax to the extent such Tax was included as a liability in the calculation of the Closing Date Net Working Capital.
 - "Buyer Indemnitees" has the meaning set forth in Section 10.1(a).
 - "Buyer Objection Notice" has the meaning set forth in Section 2.5(b).
 - "Buyer Released Parties" has the meaning set forth in Section 6.17(a).
 - "Buyer Review Period" has the meaning set forth in Section 2.5(b).
 - "Buyer's Advisors" has the meaning set forth in Section 6.3.
 - "Cash" has the meaning set forth in Section 2.1(a).
 - "Cash Portion of Purchase Price" has the meaning set forth in Section 2.1(a).

- "Chafin" has the meaning set forth in the recitals to this Agreement.
- "Change of Control Payments" has the meaning set forth in 6.8(b).
- "Claim Amount" has the meaning set forth in Section 2.8(a) below.
- "Claim Notice" has the meaning set forth in Section 2.8(a) below.
- "Closing" has the meaning set forth in Section 2.2.
- "Closing Cash Consideration" has the meaning set forth in Section 2.1(a).
- "Closing Date" has the meaning set forth in Section 2.2.
- "Closing Date Net Working Capital" has the meaning set forth in Section 2.5(a).
- "Closing Date Net Working Capital Statement" has the meaning set forth in Section 2.5(a).
- "Closing Statement" has the meaning set forth in Section 2.5(a).
- "COBRA" has the meaning set forth in Section 3.12(d).
- "Code" means the Internal Revenue Code of 1986, as amended.
- "Commitment Letter" has the meaning set forth in Section 6.6(a).
- "Companies" has the meaning set forth in the recitals to this Agreement.
- "Company Employees" has the meaning set forth in Section 3.12(a).
- "Confidentiality Agreement" has the meaning set forth in Section 6.11.
- "Consent" means any consent, approval, waiver or other authorization of, with or to any Person, or the expiration or termination of the waiting period under any Law that is designed or intended to prohibit, restrict or regulate antitrust, monopolization, restraint of trade or competition, in each case required to permit the consummation of any of the Contemplated Transactions.
 - "Consent Fees" has the meaning set forth in Section 6.8(b)
 - "Contemplated Transactions" means all transactions contemplated by this Agreement.
 - "Contingency Obligations" has the meaning set forth in Section 6.2.
 - "Continuing Employees" has the meaning set forth in Section 6.13.

"Control" (including the terms "controlled by" and "under common control with"), with respect to the relationship between or among two or more Persons, means the possession, directly or indirectly, of the power to direct or cause the direction of the affairs or management of a Person, whether through the ownership of voting securities, by contract or otherwise, including the ownership, directly or indirectly, of securities having the power to elect a majority of the board of directors or similar body governing the affairs of such Person.

"Current Assets" has the meaning set forth in Section 2.4(d).

"Current Liabilities" has the meaning set forth in Section 2.4(e).

"Drop Dead Date" has the meaning set forth in Section 9.1(d).

"Encumbrance" means any security interest, pledge, mortgage, lien, charge, lease, conditional sales agreement, claim, restriction, covenant, easement, right of way, encroachment, servitude, right of first option, right of first refusal, hypothecation, adverse claim of or restriction on ownership, voting, transfer, receipt of income or use, or other encumbrance of any kind.

"Environmental Claim" means any written notice or Proceeding by any Person alleging Liability or potential Liability (including Liability or potential Liability for investigatory costs, cleanup costs, governmental response costs, natural resource damages, fines or penalties) relating to any Environmental Losses or in respect of any Environmental Laws.

"Environmental Law" means any applicable Law relating to remediation, restoration or protection of the environment or natural resources, the storage, treatment, management, generation, transportation, or disposal of any toxic materials or waste material, whether hazardous or not, the conservation of natural resources and resource allocation (including any Law relating to development or exploitation of any natural resource) or exposure of persons or property to Hazardous Materials, including applicable Governmental Approvals pursuant to Environmental Laws and including the Clean Air Act, 42 U.S.C. Section 7401 et seq., the Comprehensive Environmental Response, Compensation and Liability Act, 42 U.S.C. Section 9601 et seq. ("CERCLA"), the Endangered Species Act, 16 U.S.C. Section 1531 et seq., the Federal Land Policy and Management Act, 43 U.S.C. Section 1701 et seq., the Federal Water Pollution Control Act, 33 U.S.C. Section 1251 et seq., the Resource Conservation and Recovery Act, 42 U.S.C. Section 6901 et seq., the Surface Mining Control and Reclamation Act, 30 U.S.C. Section 1201 et seq. ("SMCRA"), and the Toxic Substances Control Act, 15 U.S.C. Section 2601 et seq., each as amended, and any similar state Law.

"Environmental Losses" means Losses arising from a Release of Hazardous Materials or noncompliance with any Environmental Law or Permits required pursuant to any Environmental Law.

"Equipment Leases" has the meaning set forth in Section 3.18(c).

"ERISA" has the meaning set forth in Section 3.12(a).

"ERISA Affiliate" has the meaning set forth in Section 3.12(a).

- "Escrow Agreement" has the meaning set forth in Section 2.7 below.
- "Escrow Amount" has the meaning set forth in Section 2.7.
- "Escrow Fund" has the meaning set forth in Section 2.7 below.
- "Escrow Release Date" has the meaning set forth in Section 2.8(a) below.
- "Estimated Adjustment Amount" has the meaning set forth in Section 2.4(f).
- "Estimated Net Working Capital has the meaning set forth in Section 2.4(a).
- "Estimated Net Working Capital Statement" has the meaning set forth in Section 2.4(a).
- "Final Purchase Price" has the meaning set forth in Section 2.5(e).
- "Financing" has the meaning set forth in Section 6.6(a).
- "FMLA" has the meaning set forth in Section 3.12(d).
- "GAAP" means United States generally accepted accounting principles as in effect from time to time.
- "General Escrow Amount" has the meaning set forth in Section 2.7(i).
- "Governmental Approval" means any Consent of, with or to any Governmental Authority.
- "Governmental Authority" means any United States or other federal, state or local government or other political subdivision thereof, any entity, authority, tribunal, agency or body exercising executive, legislative, judicial, regulatory, fiscal or administrative functions of any such government or political subdivision, and any supranational organization of sovereign states exercising such functions for such sovereign states.
- "Governmental Authorization" means permits, licenses, approvals or other authorizations issued, granted, given or otherwise made available by a Governmental Authority.
- "Governmental Order" means, with respect to any Person, any judgment, order, writ, injunction, decree, stipulation, agreement, determination or award entered or issued by or with any Governmental Authority and binding on such Person.
 - "Hampden" has the meaning set forth in the recitals to this Agreement.
- "<u>Hazardous Materials</u>" means any material or substance defined as a "hazardous substance," "toxic substance," "hazardous waste," "solid waste," "pollutant," "hazardous" or "contaminant" or any other term of similar import under any Environmental Law or any other materials which are regulated or give rise to Liability under Environmental Laws.

- "Holdings I" has the meaning set forth in the first sentence of this Agreement.
- "Holdings II" has the meaning set forth in the first sentence of this Agreement
- "HSR Act" means the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, and the rules and regulations promulgated under the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, and any successor to such statute, rules or regulation.
- "Income Tax" means any Tax based upon, measured by, or calculated with respect to net income or profits (including, but not limited to, any capital gains or similar Tax).
 - "Income Tax Return" means any Tax Return relating to Income Taxes, including any Schedule or attachment thereto, and including any amendment thereof.
- "Indebtedness" means all "indebtedness" for borrowed money of the Acquired Companies, including any notes payable, revolving credit lines, obligations in respect of letters of credit, capital lease obligations, indebtedness of another Person which is guaranteed and any other debt (other than trade accounts payable that are Current Liabilities of the Acquired Companies arising in the ordinary course of business consistent with past practices, except to the extent that such amounts have not been paid when due), including all interest accrued thereon.
 - "Indemnity Termination Date" has the meaning set forth in Section 10.1(c).
 - "Industry" shall mean the steel or coal industry.
- "Intellectual Property" means all (i) patents, (ii) inventions, discoveries, processes, formulae, designs, models, industrial designs, know-how, confidential information, proprietary information and trade secrets, whether or not patented or patentable, (iii) trademarks, service marks, trade names, brand names, trade dress, slogans, logos and internet domain names, (iv) copyrights and other copyrightable works and works in progress, databases and software, (v) all other intellectual property rights and foreign equivalent or counterpart rights and forms of protection of a similar or analogous nature or having similar effect in any jurisdiction throughout the world, (vi) any renewals, extensions, continuations, divisionals, reexaminations or reissues or equivalent or counterpart of any of the foregoing in any jurisdiction throughout the world, and (vii) all registrations and applications for registration of any of the foregoing.
 - "International Industries" has the meaning set forth in the recitals to this Agreement.
 - "International Resource Partners" has the meaning set forth in the recitals to this Agreement.
 - "IRP Administrative" has the meaning set forth in Section 6.18.
 - "IRP GP" has the meaning set forth in the recitals to this Agreement.
 - "IRP Kentucky" has the meaning set forth in the recitals to this Agreement.

"IRP WV" has the meaning set forth in the recitals to this Agreement.

"KED" has the meaning set forth in the recitals to this Agreement.

"Knowledge" means facts or matters that such Person is actually aware of, provided such Person shall have made reasonable inquiry of those Persons who would ordinarily be expected to have knowledge of the matters represented.

"Knowledge of Buyer" means facts or matters that any of the Persons listed on Schedule 1.1(a) are actually aware of, provided that such Persons shall have made reasonable inquiry of those Persons who would ordinarily be expected to have knowledge of the matters represented.

"Knowledge of IRP GP and Resource Partners" means facts or matters that any of the Persons listed on Schedule 1.1(b) are actually aware of, provided that such Persons shall have made reasonable inquiry of those Persons who would ordinarily be expected to have knowledge of the matters represented.

"Latest Balance Sheet" has the meaning set forth in Section 3.4.

"Law" means any applicable Governmental Order or any applicable provision of any constitution, law (including principles of the common law), legally binding directive, treaty, statute, rule, regulation or order of any Governmental Authority.

"Leased Real Property" has the meaning set forth in Section 3.19(a).

"Lenders" has the meaning set forth in Section 6.6(a).

"Liabilities" means any and all debts, claims, commitments, loss contingencies, liabilities and obligations of every kind and description whatsoever, whether direct or indirect, known or unknown, disclosed or undisclosed, matured or unmatured, accrued or unaccrued, absolute or conditional, asserted or unasserted, contingent or otherwise.

"Lightfoot" has the meaning set forth in the recitals to this Agreement.

"L&K" has the meaning set forth in the recitals to this Agreement.

"LMR" has the meaning set forth in the recitals to this Agreement.

"Losses" means any and all claims, losses, damages, fines, penalties and costs (in each case including reasonable out-of-pocket expenses (including reasonable attorneys', accountants', technical consultants', engineers' and experts' fees and expenses)).

"Marshall Miller" means Marshall Miller and Associates, Inc.

"Marshall Miller Report" means the February 17, 2011, Evaluation of Properties Controlled by International Resource Partners LP as of December 31, 2010, by Marshall Miller.

"Material Adverse Effect" or "Material Adverse Change" means, in the case of the definition of Permitted Encumbrances, Article III and Article VIII, any effect or change that, individually or together with any other effect or change, is or would reasonably be expected to be materially adverse to the business operations, properties, assets, liabilities, results of operations, or the condition (financial or otherwise) of the Acquired Companies, taken as a whole; provided that none of the following shall be deemed to constitute, and none of the following shall be taken into account in determining whether there has been, a Material Adverse Effect or Material Adverse Change: (a) any adverse change, event, development, or effect arising from or relating to (1) general business or economic conditions; (2) the Industry generally, including, without limitation, a decline in prices or demand for coal or steel, increases in costs of transportation and raw materials, and labor shortages; (3) national or international political or social conditions, including the engagement by the United States in hostilities, whether or not pursuant to the declaration of a national emergency or war, or the occurrence of any military or terrorist attack upon the United States, or any of its territories, possessions, or diplomatic or consular offices or upon any military installation, equipment or personnel of the United States; (4) financial, banking, or securities markets (including any disruption thereof and any decline in the price of any security or any market index); (5) changes in United States generally accepted accounting principles; (6) changes in Laws, orders or other binding directives issued by any Governmental Authority, including, without limitation, changes in Environmental Laws or Occupational Health and Safety Laws (in each of (1) through (6), only to the extent that such factor does not materially disproportionately affect any of the Acquired Companies relative to their competitors in the Industry); or (7) the taking of any action contemplated by this Agreement and the other Transaction Documents; (b) any existing event, occurrence or circumstances with respect to which Buyer has been provided information or knew or should have known; and (c) any adverse change in or effect on the business of the Acquired Companies that is cured before the earlier of (i) the Closing Date and (ii) the date on which this Agreement is terminated pursuant to Section 9.1 hereof; provided that any failure to meet any financial or other projection or forecast for any period shall not, in and of itself, constitute a Material Adverse Effect or Material Adverse Change. "Material Adverse Effect" or "Material Adverse Change" in the case of Article IV and Article VII means any effect or change that would prevent Buyer from consummating in a timely manner the Contemplated Transactions.

"Material Contracts" has the meaning set forth in Section 3.15(a).

"Material Personal Property" has the meaning set forth in Section 3.18(a).

"Membership Interests" means all of the issued and outstanding membership interests of Holdings I, Holdings II, IRP WV, Resources, Hampden, Snap Creek, Chafin, L&K, Rockhouse, IRP Kentucky, LMR and Buck Branch.

"Mining Assets" has the meaning set forth in the recitals to this Agreement.

"MSHA" means the U.S. Mine Health and Safety Administration.

"Net Working Capital" has the meaning set forth in Section 2.4(c).

"New Commitment Letter" has the meaning set forth in Section 6.6(c).

"Occupational Health and Safety Law" means any Law designed to provide safe and healthful working conditions and to reduce occupational safety and health hazards, and any program, whether governmental or private (including those promulgated or sponsored by industry associations and insurance companies), designed to provide safe and healthful working conditions, including the Federal Mine Safety and Health Act of 1977, the Mine Improvement and New Emergency Response Act of 2006, the Supplemental Mine Improvement and New Emergency Response Act, the Black Lung Benefits Revenue Act of 1977, the Black Lung Benefits Reform Act of 1977, and similar West Virginia state Laws, as amended.

"Ordinary Course of Business" means, the usual, regular and ordinary course of a business consistent with the past practice thereof.

"Organizational Document" means, as to any Person, its constitution, certificate articles of incorporation or articles of organization, its regulations, by-laws or operating agreement or any equivalent documents under the Law of such Person's jurisdiction of incorporation or organization.

"Owned Real Property" has the meaning set forth in Section 3.19(b).

"Partnership Interests" means all of the issued and outstanding limited partnership units and general partnership units of Resource Partners.

"PBGC" has the meaning set forth in Section 3.12(c).

"Permit" means any consent, approval, permit, license, certificate, exemption or other authorization which the Law requires any Acquired Company to hold in order to develop and operate its respective assets and conduct its respective business (including, without limitation, application, any authorization or permit relating to coal mining, preparation, load out or reclamation operations).

"Permits Transfer Period" has the meaning set forth in Section 6.10(c).

"Permitted Encumbrance" means (a) statutory liens securing payments not yet due arising in the Ordinary Course of Business for amounts not yet due (b) encroachments, such imperfections or irregularities of title, non-monetary claims, and rights of way, easements, reservations, covenants, oil and gas leases and other customary restrictions or non-monetary encumbrances, as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect, (c) liens for Taxes not yet due and payable or that are being contested in good faith by appropriate proceedings, (d) any other non-monetary liens that would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect, (e) mechanics', materialmen's or other like liens arising or incurred in the Ordinary Course of Business or by operation of Law but only to the extent that such lien has not been filed of record or, to the Knowledge of Sellers, been threatened to be filed (f) pledges and deposits to secure the performance of bids, trade contracts, leases, surety and appeal bonds, performance bonds and other obligations of a similar nature, in each case in the Ordinary Course of Business of any Acquired Company, (g) liens associated with the Indebtedness, and (h) the terms, covenants, conditions and reservations contained in (1) each deed conveying the Owned Real Property to the applicable Acquired Company, and (2) each Real Property Lease; provided that none of such Encumbrances will individually or in the aggregate prevent or materially interfere with the operation of any Acquired Company's business in the Ordinary Course of Business.

"Person" means any individual, partnership, firm, corporation, association, trust, unincorporated organization, joint venture, limited liability company, Governmental Authority or other entity.

"Post-Signing Event Update" has the meaning set forth in Section 6.16.

"Pre-Signing Event Update" has the meaning set forth in Section 6.16.

"Pre-Transaction Notification Rules" has the meaning set forth in Section 6.14(a).

"Proceeding" means any action, claim, demand, suit, proceeding, arbitration, citation, summons, subpoena, inquiry or investigation of any nature, civil, criminal, regulatory or otherwise, in law or in equity, by or before any Governmental Authority.

"Purchase Price" has the meaning set forth in Section 2.1(a).

"Purchase Price Deductions" has the meaning set forth in Section 6.8(a).

"Real Property" has the meaning set forth in Section 3.19(b).

"Real Property Leases" has the meaning set forth in Section 3.19(a).

"Redeemed Partners" has the meaning set forth in Section 6.2.

"Redemption" has the meaning set forth in Section 6.2.

"Reference Net Working Capital" has the meaning set forth in Section 2.4(b).

"Related Person" means, with respect to a specific Person, any officer, director, employee, agent, shareholder, representative, successor or assign of such Person.

"Release" means any spilling, leaking, pumping, pouring, emitting, emptying, discharging, injecting, escaping, leaching, dumping, or disposing of Hazardous Materials into or upon the environment, including, surface water, ground water, a drinking water supply, land surface or subsurface strata or ambient air (including the abandonment or discarding of barrels, containers, and other closed receptacles containing any Hazardous Material).

"Reserve Dispute" has the meaning set forth in Section 2.8(b)(iii).

"Resource Partners" has the meaning set forth in the recitals to this Agreement.

"Resources" has the meaning set forth in the recitals to this Agreement.

"Rockhouse" has the meaning set forth in the recitals to this Agreement.

"SEC" means the Securities and Exchange Commission.

- "Securities Act" means the Securities Act of 1933, as amended.
- "Seller" has the meaning set forth in the first sentence of this Agreement.
- "Sellers" has the meaning set forth in the first sentence of this Agreement.
- "Seller Bonds" means those deposits, letters of credit, trust funds, bid bonds, performance bonds, reclamation bonds and surety bonds (and all such similar undertakings) set forth on Schedule 1.1(c).
 - "Seller Guarantees" means those guarantees, indemnities, letters of credit, letters of comfort and similar credit obligations set forth on Schedule 1.1(d).
 - "Seller Indemnitees" has the meaning set forth in Section 10.1(b).
 - "Seller Released Claims" has the meaning set forth in Section 6.17(a).
 - "Sellers Claims and Losses" has the meaning set forth in Section 6.17(a).
 - "Snap Creek" has the meaning set forth in the recitals to this Agreement.
 - "Special Escrow Amount" has the meaning set forth in Section 2.7(ii).
 - "Special Indemnity Matters" has the meaning set forth in Section 6.8(c).
 - "Straddle Period" means any Tax period that begins on or before the Closing Date and ends after the Closing Date.
- "Subsidiaries" means any and all corporations, partnerships, limited liability companies and other entities with respect to which a company, directly or indirectly, owns securities having the power to elect a majority of the board of directors or similar body governing the affairs of such entity.
- "Tax" or "Taxes" means (a) any taxes of any kind, including, without limitation, those measured on, measured by or referred to as, income, alternative or add-on minimum, gross receipts, escheat, capital, capital gains, sales, use, ad valorem, franchise, profits, license, privilege, transfer, withholding, payroll, employment, social security, excise, severance, stamp, occupation, premium, value added, goods and services, property, environmental or windfall profits taxes, customs, duties or similar fees, assessments or charges of any kind whatsoever, including any interest, penalty or addition thereto, whether disputed or not, and (b) any Liability for the payment of any amounts of the type described in clause (a) of this definition as a result of being a member of an affiliated, consolidated, combined or unitary group for any period, as a result of any tax sharing, tax indemnity or tax allocation agreement, arrangement or understanding, or as a result of being liable for another Person's taxes as a transferee or successor, by contract or otherwise.
- "<u>Tax Return</u>" means any return, report, declaration, form, election letter, statement or other information or document required to be or prepared by a Person, filed with any Governmental Authority with respect to Taxes, including, but not limited to, any schedule or attachment thereto or amendment thereof.

- "Tortoise" has the meaning set forth in the recitals to this Agreement.
- "Transaction Documents" means, collectively, the Agreement and all certificates and other documents executed and delivered at the Closing.
- "Transfer Applications" has the meaning set forth in Section 6.10(a).
- "Transfer Taxes" has the meaning set forth in Section 6.12(b).
- "Unaudited Financials" has the meaning set forth in Section 3.4(a).
- "Unpaid Indebtedness" has the meaning set forth in Section 2.5(a).
- "Unpaid Transaction Expenses" has the meaning set forth in Section 2.5(a).
- "Update" has the meaning set forth in Section 6.16.
- "WARN Act" means the federal Worker's Adjustment and Retraining Notification Act, as amended, and any similar state law relating to plant closing or layoffs.
- "WVOMH&S" means the West Virginia Office of Mine Health and Safety.

1.2 Other Interpretive Provisions.

- (a) The words "hereof," "herein" and "hereunder" and words of similar import when used in this Agreement refer to this Agreement as a whole (including any Exhibits or Schedules hereto) and not to any particular provision of this Agreement, and all Article, Section, Schedule and Exhibit references are to this Agreement unless otherwise specified. Exhibits, Schedules, annexes or appendices to any document shall be deemed incorporated by reference in such document. The words "include," "includes" and "including" shall be deemed to be followed by the phrase "without limitation." The meanings given to terms defined herein shall be equally applicable to both the singular and plural forms of such terms. Whenever the context may require, any pronoun shall include the corresponding masculine, feminine and neuter forms. All references to "dollars" or "\$" shall be deemed references to the lawful money of the United States of America. Reference to "days" shall mean calendar days, unless the term "Business Days" is used.
- (b) The disclosure of any matter on a Schedule shall not be deemed to be an admission or representation as to the materiality of the matter so disclosed. Any matter disclosed on a Schedule pursuant to any Section of this Agreement shall be deemed to have been disclosed for purposes of another Section or Sections of this Agreement if the relevance or applicability of such disclosure to the subject matter of such other Section or Sections is reasonably apparent on the face of such disclosure.

ARTICLE II

PURCHASE AND SALE

2.1 Purchase Price.

- (a) On the terms and subject to the conditions set forth in this Agreement, in exchange for the Partnership Interests, Buyer shall pay to Sellers the aggregate purchase price (the "Purchase Price") of (i) Four Hundred Seventy-Five Million Dollars (\$475,000,000) (the 'Cash Portion of the Purchase Price') minus (ii) the Purchase Price Deductions. The Purchase Price shall be subject to adjustment at and after the Closing in accordance with Section 2.4 and Section 2.5. At the Closing, Buyer shall pay to Sellers the Purchase Price increased or decreased for any Estimated Adjustment Amount pursuant to Section 2.4(f) and an amount reflecting the Acquired Companies' cash, in the aggregate, on the Closing Date as reflected on the consolidated balance sheet of Resource Partners in cash (the "Cash" and the Purchase Price, as increased or decreased by the Estimated Adjustment Amount and the Cash, is referred to as the "Closing Cash Consideration"). The Closing Cash Consideration shall be payable at the Closing as follows:
 - (i) Buyer shall deliver the Escrow Amount to the Escrow Agent pursuant to the terms and conditions of Section 2.7; and
- (ii) Buyer shall pay to Sellers the balance of the Closing Cash Consideration (being the Closing Cash Consideration less the Escrow Amount) by wire transfer of immediately available United States funds, in the percentage per Seller as set forth on Schedule 2.1(a)(ii), to accounts designated by Agent for the benefit of Sellers in written instructions given to Buyer at least two (2) Business Days prior to the Closing. Prior to the payment to Sellers of the balance of the Closing Cash Consideration, such Closing Cash Consideration will additionally be reduced by the amounts estimated by Resource Partners, IRP GP and Agent necessary to pay fees and expenses of Sellers including those incurred in connection with the transactions contemplated hereby, which amounts Agent shall certify in writing to Buyer at least one (1) Business Day prior to the Closing, and Buyer shall pay such amounts by wire transfer of immediately available United States funds to accounts designated by Agent for the benefit of Sellers.
- (b) As promptly as practicable following the execution and delivery of this Agreement, and in any case prior to the Closing, Buyer and Sellers shall use reasonable efforts to agree upon the allocation of the Purchase Price (and all other capitalizable costs) among the assets of the Acquired Companies, which the parties shall adjust to reflect any adjustments made to the Purchase Price in accordance with Section 2.5(f). Any such allocation shall be intended to comply with the allocation method required by Section 1060 of the Code. The parties shall cooperate in complying with all requirements of Section 1060 of the Code and the regulations thereunder, and the allocation shall be adjusted only if and to the extent necessary to comply with such requirements. Buyer and Sellers agree that they will not take nor will they permit any of their Affiliates or Related Persons to take, for income Tax purposes, any position inconsistent with any such agreed-upon allocation, if any; provided, however, that (a) Buyer's total cost for the assets of each Acquired Company may differ from the total amount allocated hereunder to reflect Buyer's transaction costs other than the Purchase Price, and (b) the amount realized by Sellers may differ from the total amount allocated hereunder to reflect Sellers' transaction costs that reduce the amount realized for income tax purposes.

Under no circumstance shall Buyer be liable to any Seller or other Person for any action, omission or delay of the Agent or any other recipient in dealing with or distributing any payment or any portion of the Purchase Price or of any other payment or performance made by or on behalf of Buyer in accordance herewith or any other Transaction Document, to or as prescribed by the Agent. Nothing contained in Section 2.1(a)(ii) providing for allocation of the Purchase Price hereunder in accordance with Schedule 2.1(a)(ii) shall be deemed to limit the foregoing.

2.2 The Closing. Unless this Agreement shall have been terminated pursuant to Article IX, subject to the satisfaction or waiver of the conditions set forth in Articles VII and VIII, the closing (the "Closing") of the Contemplated Transactions shall take place at the offices of Jackson Kelly PLLC, 1600 Laidley Tower, Charleston, West Virginia 25301, or at such other place as may be agreed upon by Agent and Buyer, effective at 12:01 a.m. on the third Business Day after all of such conditions have been satisfied or waived by the party or parties entitled to waive the same, or at such other date and time as may be agreed upon by Sellers and Buyer (the "Closing Date"); provided that, if Buyer is in compliance with its obligations contained in Section 6.6 but has not consummated financing to complete the Contemplated Transactions, then the Closing Date may be extended beyond the time when all conditions have been satisfied or waived but only up to and including June 1, 2011, unless mutually extended in writing by Buyer and Agent.

2.3 Deliveries at the Closing

- (a) At the Closing, Agent shall deliver or cause to be delivered, as applicable, to Buyer the following:
 - (i) an instrument of assignment evidencing the transfer of the Partnership Interests in a form reasonably acceptable to Buyer;
- (ii) copies of the resolutions of Lightfoot, International Industries and IRP GP, authorizing and approving this Agreement, the other Transaction Documents and the Contemplated Transactions, certified by an officer of Sellers to be true and complete and in full force and effect and unmodified as of the Closing Date;
 - (iii) the officer's certificate required by Section 8.3;
- (iv) a copy of the certificate of existence, as applicable, of each Acquired Company certified as true and correct by the Secretary of State of the State of West Virginia, the State of Delaware and the Commonwealth of Kentucky (as the case may be);
- (v) resignations, effective as of the Closing, of each member of the board of directors or board of managers, as applicable, and of each officer, of each of the Acquired Companies;

- (vi) all minute books and members' interest or stock records of the Acquired Companies;
- (vii) with respect to each Seller, an opinion regarding authority from outside counsel for such Seller in form and substance reasonably acceptable to Buyer; and
- (vii) all other documents, instruments and writings required to be delivered by Buyer at or prior to the Closing pursuant to this Agreement or otherwise required in connection herewith.
 - (b) At the Closing, Buyer shall execute and deliver or cause to be executed and delivered, as applicable, to Agent the following:
 - (i) the Closing Cash Consideration, less the Escrow Amount, by wire transfer of immediately available funds as specified in Section 2.1(a)(ii);
- (ii) copies of the resolutions of Buyer authorizing and approving this Agreement, the Transaction Documents and the Contemplated Transactions certified by an officer of Buyer to be true and complete and in full force and effect and unmodified as of the Closing Date;
 - (iii) the officer's certificate required by Section 7.3;
- (iv) a certificate (dated not less than five (5) Business Days prior to the Closing) of the Secretary of State of the State of Virginia as to the good standing of Buyer; and
- (v) all other documents, instruments and writings required to be delivered by Agent at or prior to the Closing pursuant to this Agreement or otherwise required in connection herewith.

2.4 Estimated Closing Adjustments.

- (a) No later than the fifth (Sh) Business Day prior to the Closing Date, Agent shall deliver to Buyer a statement (the "<u>Estimated Net Working Capital Statement</u>") setting forth Agent's good faith estimate of Net Working Capital of the Acquired Companies as of the Closing Date ("<u>Estimated Net Working Capital</u>") and Agent's calculation thereof in reasonable detail, calculated in accordance with GAAP consistently applied, except as described in <u>Schedule 2.4(a)</u> (the "<u>Accounting Principles</u>")
 - (b) "Reference Net Working Capital" means an amount equal to \$18,500,000.
- (c) "Net Working Capital" means the amount, if any, by which Current Assets of the Acquired Companies are greater or less than Current Liabilities of the Acquired Companies as calculated pursuant to the Accounting Principles.

- (d) "Current Assets" means with respect to the Acquired Companies, their accounts receivable, coal ready for shipment, parts and supplies inventories, deposits, prepaid expenses and other current assets calculated on a consolidated basis and in accordance with the Accounting Principles. Cash, cash equivalents, returned bond deposits and debt of any kind due from Affiliates will be excluded from Current Assets.
- (e) "Current Liabilities" means, with respect to the Acquired Companies, their accounts payable, accrued expenses and other current liabilities calculated on a consolidated basis and in accordance with the Accounting Principles. Current Liabilities shall not include current maturities of Indebtedness, accrued interest expense or accrued expenses related to derivative instruments.
- (f) "Estimated Adjustment Amount" means the amount by which the Estimated Net Working Capital exceeds the Reference Net Working Capital, or the amount by which the Reference Net Working Capital exceeds the Estimated Net Working Capital.

2.5 Closing Adjustments.

- (a) <u>Closing Date Net Working Capital</u>. Within sixty (60) days after the Closing Date, Agent shall deliver to Buyer a statement (the "<u>Closing Date Net Working Capital Statement</u>") setting forth Agent's calculation of Net Working Capital as of and including the Closing Date (the "<u>Closing Date Net Working Capital</u>"), together with an audit report of Resource Partners' independent accountants certifying as to the accuracy of Agent's calculations. The Closing Date Net Working Capital Statement is the "<u>Closing Statement</u>". The Closing Statement shall also include a description of (i) all Indebtedness of the Acquired Companies as of and including the Closing Date set forth on <u>Schedule 2.6</u> and not paid by Sellers pursuant to Section 2.6, except to the extent included in Current Liabilities of the Acquired Companies for purposes of calculating Net Working Capital ("<u>Unpaid Indebtedness</u>"), and (ii) all fees and expenses of Sellers and the Acquired Companies not paid by Sellers pursuant to Section 11.2, except to the extent included in Current Liabilities of the Acquired Companies for purposes of calculating Net Working Capital ("<u>Unpaid Transaction Expenses</u>"). The Closing Statement shall be prepared and all determinations therein made in accordance with the Accounting Principles and shall be consistent with the manner in which the Reference Net Working Capital Statement was prepared.
- (b) Review of Buyer. Buyer shall have thirty (30) days after the delivery to Buyer of the Closing Statement (the "Buyer Review Period") in which to notify Agent in writing (the "Buyer Objection Notice") of any good faith disagreement with the Closing Statement, setting forth in reasonable detail (i) the items or amounts with which Buyer disagrees and the basis for such disagreement, and (ii) Buyer's proposed corrections to the Closing Statement. Notwithstanding the timely delivery of a Buyer Objection Notice, Buyer on the one hand, or Agent on the other hand, as applicable, shall make any and all payments as to amounts not in dispute in accordance with Section 2.5(f) prior to the resolution of the Buyer Objection Notice pursuant to Section 2.5(c). If Buyer does not deliver a Buyer Objection Notice within the Buyer Review Period, Buyer shall be deemed to agree in all respects with the Closing Statement and the items and amounts reflected thereon shall be final and binding upon Buyer and Sellers for purposes of Section 2.5.

- (c) Review by Auditor. If Buyer gives Agent a timely Buyer Objection Notice to the Closing Statement, then Agent and Buyer shall use their reasonable best efforts to resolve any such objection. If a written resolution is not obtained within thirty (30) days after the Agent has received the Buyer Objection Notice, the issue(s) in dispute will be submitted to the Auditor for resolution, which resolution will be limited to determining Closing Date Net Working Capital, Unpaid Indebtedness and Unpaid Transaction Expenses, and any other accounting matters specifically referred to the Auditor by the mutual agreement of the Agent and Buyer. The determination of the Auditor shall be set forth in a written notice delivered, as promptly as practicable after the issue(s) in dispute have been submitted to the Auditor, to the Buyer and Agent by the Auditor and will be final, binding and conclusive on the parties. Buyer and Sellers shall each bear fifty percent (50%) of the fees and expenses of the Auditor for such determination. Buyer and Agent will use all reasonable efforts to cause the Auditor to render its decision as promptly as practicable, including without limitation by promptly complying with all reasonable requests by the Auditor for information, books, records and similar items.
- (d) <u>Cooperation</u>. Buyer, on the one hand, and Sellers, on the other hand, shall provide reasonable cooperation and assistance to each other and to the Auditor in the preparation of the Closing Statement and in the conduct of the reviews referred to in this Section 2.5, including providing reasonable access to the books, records, work papers and personnel of the Acquired Companies.
- (e) <u>Final Purchase Price</u>. The "<u>Final Purchase Price</u>" means an amount equal to the Purchase Price, plus the amount, if any, by which Closing Date Net Working Capital exceeds the Reference Net Working Capital, or minus the amount, if any, by which the Reference Net Working Capital exceeds the Closing Date Net Working Capital.
- (f) Adjustment Payment. Within three (3) Business Days after the calculation of Closing Date Net Working Capital becomes final pursuant to Section 2.5(b) or (c): (i) Buyer shall pay to Agent, for the benefit of Sellers, by wire transfer of immediately available funds to an account designated by Agent, an amount equal to the amount, if any, by which the Final Purchase Price exceeds the Closing Cash Consideration, or (ii) Sellers shall pay to Buyer, by wire transfer of immediately available funds to an account designated by Buyer, an amount equal to the sum of (A) an amount equal to the amount, if any, by which the Closing Cash Consideration exceeds the Final Purchase Price, plus (B) an amount equal to the sum of the Unpaid Indebtedness, plus the Unpaid Transaction Expenses. The amount of the payment pursuant to clause (i) or clause (ii) of this Section 2.5(f) being referred to as the "Adjustment Payment", in either case, (x) together with interest on the Adjustment Payment at the Applicable Rate from and including the Closing Date to, but excluding, the date of such payment and (y) giving effect to any amounts paid pursuant to Section 2.4(f). As used in this Agreement, the term "Applicable Rate" means a rate per annum equal to the "prime rate" as set forth from time to time in The Wall Street Journal "Money Rates" column. Buyer and Sellers agree that, unless otherwise required by Law, any payments made pursuant to this Section 2.5 shall be treated as an adjustment to the Final Purchase Price for all Tax purposes.
- 2.6 <u>Payment of the Indebtedness</u>. Prior to or at the Closing, Sellers shall pay the Indebtedness set forth on Schedule 2.6 in full and shall deliver to Buyer satisfactory evidence that they have done so. Prior to the Closing, Sellers shall use their commercially reasonable efforts to cause each holder of Indebtedness listed on Schedule 2.6 to enter into an agreement to release or authorize the release of any Encumbrances on any of the Acquired Companies' assets securing such Indebtedness upon payment in full of such Indebtedness and shall cause such agreement to be delivered to Buyer.

2.7 Escrow Amount and Escrow Fund. At the Closing, Buyer shall pay to the Escrow Agent (i) \$20,000,000 (the "General Escrow Amount"), and (ii) \$10,000,000 (the "Special Escrow Amount", and collectively with the General Escrow Amount, the "Escrow Amount") (the "Escrow Fund") in cash payable by wire transfer of immediately available United States funds for deposit in an Escrow Account in accordance with the terms and conditions of the Escrow Agreement to be entered into by and among Buyer, Agent, on behalf of Sellers, and the Escrow Agent in the form attached hereto as Exhibit A (the "Escrow Agreement"). The Escrow Fund shall be held and distributed by the Escrow Agent in accordance with the terms and conditions of this Agreement and the Escrow Agreement.

2.8 Escrow Release.

(a) General. Subject to the terms and provisions of this Section 2.8 and the Escrow Agreement, fourteen (14) months following the Closing (the Escrow Release Date"), the Escrow Agent shall disburse to Sellers a portion of the Escrow Fund in an amount equal to the General Escrow Amount (as reduced by any amounts previously disbursed to Sellers or Buyer pursuant to Section 2.8(b), Section 2.8(c)(iii), Section 10.1 and the Escrow Agreement) in accordance with the terms of the Escrow Agreement. In the event, however, that Agent has received, on or before the Escrow Release Date, a notice (a "Claim Notice") submitted in good faith by Buyer that the Escrow Agent may be required to disburse all or a portion of the Escrow Amount (such claimed amount, the "Claim Amount") to Buyer pursuant to Section 10.1(a) or Sections 2.7 or 2.8, then the portion of the Escrow Amount subject to such Claim Notice shall continue to be held by the Escrow Agent until the Claim Amount with respect thereto has been resolved. As soon as any dispute with respect to any such Claim Amount has been resolved in accordance with the terms of the Escrow Agreement, the Escrow Agent shall be instructed to disburse such portion of the General Escrow Amount in the Escrow Fund, if any, that is required to be disbursed to Buyer pursuant to Section 10.1(a) or Sections 2.7 or 2.8 in connection with such Claim Amount, and the Escrow Agent shall disburse the entire remaining portion of the General Escrow Amount, if any, to Sellers.

(b) Reserve Claims.

- (i) Since the issuance of the Marshall Miller Report, the Acquired Companies have continued and will continue to prove metallurgical reserves not classified in the Marshall Miller Report as proven metallurgical reserves ("Additional Reserves") through drilling and/or adding additional acreage.
- (ii) Up to \$5,000,000 of the amounts remaining in the General Escrow Amount will be released to Buyer if at least 10 Million tons of Additional Reserves are not proven by the Escrow Release Date. The portion of such \$5,000,000 (less drilling expenses paid pursuant to Section 2.8(b)(iv)) to be released to Buyer shall be equal to the ratio of (A) 10 Million minus the number of tons of Additional Reserves proven prior to the Escrow Release Date, to (B) 10 Million tons.

- (iii) Coal will be classified as "metallurgical" and as "reserves" under customary coal industry standards or otherwise to the reasonable satisfaction of Buyer, provided that the resulting washed quality of coal must meet such standards for metallurgical coal. Any dispute between the parties as to whether mineral interests constitute "reserves" or "Additional Reserves" or "Additional Reserves" shall be finally settled by Marshall Miller. These Additional Reserves are to be established by drilling reserves covered by existing leases of the Acquired Companies or under new leases, agreements or other arrangements that are in active negotiation by the Acquired Companies as of the date of this Agreement, rather than under new leases not being pursued as of the date of this Agreement. In the event that Agent has received, on or before the Escrow Release Date, a notice submitted in good faith by Sellers that there is a dispute regarding Additional Reserves (a "Reserve Dispute"), then the portion of the General Escrow Amount subject to such Reserve Dispute shall continue to be held by the Escrow Agent until the Reserve Dispute has been resolved.
- (iv) Up to \$1,000,000 of the General Escrow Amount shall also be available to fund drilling and other reasonable out-of-pocket expenses that are incurred by the Acquired Companies in connection with any efforts to prove such Additional Reserves. Such funds shall be released to Buyer from the General Escrow Amount promptly upon its delivery to Escrow Agent of reasonable documentation that such expenses have been incurred for such purpose. In no event shall the General Escrow Amount be paid for expenses incurred for drilling after the date that 10 Million tons of Additional Reserves are proven. Any dispute between the parties as to whether any expenditures qualify for reimbursement to Buyer or the Acquired Companies pursuant to this Section 2.8(b)(iv) shall be finally settled by Marshall Miller.
- (v) Both parties will use their best efforts to cause such Additional Reserves to be proven by the Acquired Companies, including instituting an active drilling program to prove such Additional Reserves; provided that such efforts by Buyer shall not include making any cash payments not reimbursed out of the General Escrow Amount or entering into any agreement or arrangement that Buyer concludes in good faith is not in the best interests of the Acquired Companies to make or enter into. After the Closing Date, Buyer shall provide Agent with regular updates of the status of its drilling program with respect to the Additional Reserves, including providing any data reasonably requested by Agent.

(c) Mine 8 Related Claims.

- (i) All fines and legal fees of Buyer and/or the Acquired Companies related to MSHA or WVOMH&S violations that took place at the Acquired Companies' Rockhouse Mine 8 ("Mine 8") prior to Closing, whether disclosed to Buyer or not("Prior Violations"), will be paid from the Special Escrow Amount (first dollar out not subject to any deductible).
- (ii) All claims for damages suffered by Buyer and/or the Acquired Companies that are related to the Prior Violations also will be paid from the Special Escrow Amount (first dollar out not subject to any deductible). Damages for these purposes shall include, but not be limited to, \$20,000 per day that Mine 8 is designated by MSHA to be on a Potential Pattern of Violation (PPOV) list. Any other damages that can be proven by Buyer shall be in addition to the per diem amount.

- (iii) If, prior to the Escrow Release Date, Mine 8 is designated on a Pattern of Violation (POV) list by MSHA that is attributable to the Prior Violations, then all claims for damages suffered by Buyer and/or the Acquired Companies that are related to the Prior Violations may at Buyer's option be paid out of the General Escrow Amount before proceeding against the Special Escrow Amount. Damages for these purposes shall include, but not be limited to, \$20,000 per day that Mine 8 is designated by MSHA to be on a POV list. Any other damages that can be proven by Buyer shall be in addition to this per diem amount.
- (iv) In the event of a dispute regarding claims by Buyer for damages under this Section 2.8(c), the parties shall use their reasonable best efforts to resolve any such disputes. If a written resolution is not obtained within thirty (30) days after either party provides written notice of a dispute, the issue(s) in dispute will be submitted to Norwest Corporation for resolution, which resolution shall be limited to determining such issue(s) in dispute; provided that the \$20,000 per day liquidated damages amount shall not be subject to dispute or arbitration. The determination of Norwest Corporation shall be set forth in a written notice delivered, as promptly as practicable after the issue(s) in dispute have been submitted to Norwest Corporation, to Buyer and Agent by Norwest Corporation and will be final, binding and conclusive on the parties. The losing party will pay all fees and expenses of such arbitration including legal fees for the prevailing party. Buyer and Agent will use reasonable best efforts to cause Norwest Corporation to render its decision as promptly as practicable, including without limitation by promptly complying with all reasonable requests by Norwest Corporation for information, books, records and similar items.
- (v) The Special Escrow Amount shall remain in place and not be released to Agent until such time as all legal proceedings and fines relating to the Prior Violations have been finally resolved (such date being referred to as the "Final Release Date"). Until the Final release Date, claims may continue to be made against the Special Escrow Amount as provided hereunder. Upon the Final Release Date, the Escrow Agent shall disburse to Sellers any remaining portion of the Special Escrow Amount.
- (vi) Buyer shall provide and shall cause its outside counsel to the Acquired Companies to provide prompt and regular updates to Agent regarding material developments in connection with the Prior Violations, including the status of any negotiations and proceedings.

ARTICLE III

REPRESENTATIONS AND WARRANTIES OF IRP GP AND RESOURCE PARTNERS RELATED TO THE ACQUIRED COMPANIES

IRP GP and Resource Partners hereby jointly and severally represent and warrant to Buyer, as of the date hereof and as of the Closing Date, other than representations and warranties that expressly speak as of the specific date and time (which need only be true and correct in all respects or true and correct in all material respects as applicable, as of such date and time), as follows:

- 3.1 <u>Organization</u>. Each of the Acquired Companies is duly organized and validly existing under the Laws of its jurisdiction of organization. Each of the Acquired Companies has the requisite organizational power and authority to own, lease and operate its assets and to carry on its business as now being conducted and is duly qualified or licensed to do business and is in good standing in the jurisdictions in which the ownership of its property or the conduct of its business requires such qualification or license, except where the failure to be so qualified or licensed (a) would not reasonably be expected, individually or in the aggregate, to prevent any of the Acquired Companies from consummating the Contemplated Transactions or (b) would not reasonably be expected, individually or in the aggregate, to have a Material Adverse Effect. Resource Partners has delivered to Buyer true and complete copies of the Organizational Documents of each of the Acquired Companies.
- 3.2 <u>Authorization</u>: <u>Enforceability</u>. Resource Partners has the requisite organizational power and authority to execute and deliver this Agreement and the other Transaction Documents and to perform its obligations hereunder and thereunder. The execution and delivery of this Agreement and the other Transaction Documents by Resource Partners, and the performance of its obligations hereunder and thereunder, has been duly authorized by all necessary action on the part of Resource Partners, and, upon such authorization, no other proceedings or actions are necessary to authorize and consummate this Agreement, the other Transaction Documents or the Contemplated Transactions. This Agreement has been, and upon the Closing Date the other Transaction Documents will be, duly executed and delivered by Resource Partners, and, assuming due authorization, execution and delivery by Buyer, constitute the valid and binding agreements of Resource Partners and are enforceable against Resource Partners in accordance with their terms, except as may be limited by applicable bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium and other similar Laws relating to or affecting creditors' rights generally and general equitable principles (whether considered in a proceeding in equity or at law).
- 3.3 Interests. The authorized, issued and outstanding Membership Interests of the Acquired Companies are set forth on Schedule 3.3, together with the identity of the members of each of the Acquired Companies, as applicable. On the Closing Date, Resource Partners will be the record and beneficial owner and holder of the Membership Interests, in each case free and clear of all Encumbrances. All of the outstanding equity securities of each Acquired Company have been duly authorized and validly issued and are fully paid and nonassessable. Other than the Organizational Documents, there are no contracts relating to the issuance, sale, or transfer of any equity securities or other securities of any Acquired Company. Except as set forth on Schedule 3.3, there are no outstanding warrants, options, agreements, convertible or exchangeable securities or other commitments pursuant to which any Acquired Company is or may become obligated to issue or sell any shares of equity securities or other securities, and there are no equity securities of any Acquired Company reserved for issuance for any purpose.

3.4 Financial Statements.

- (a) IRP GP has delivered to Buyer: (i) the financial statements of Resource Partners as of December 31, 2010, 2009 and 2008, audited by PriceWaterhouseCoopers LLP, which include true and correct information associated with the respective balance sheets and statements of operations and cash flows of the Acquired Companies (the "Audited Financials"), and (ii) if available, on the date even herewith, the unaudited financial statements for the month ended January 31, 2011, including the unaudited consolidated balance sheet as of January 31, 2011 (the "Latest Balance Sheet"), and the related statements of operations and cash flows for the period then ended, which includes the respective balance sheets and statements of operations and cash flows of the Acquired Companies (the "Unaudited Financials", and such Audited Financials and Unaudited Financials, collectively "Financial Statements"). The Financial Statements fairly present in all material respects the consolidated financial position of the Acquired Companies and the results of their operations and cash flows as of the dates and for the periods indicated in accordance with GAAP consistently applied; provided, however, that the Unaudited Financials are prepared and derived from the consolidated financial statements of Resources Partners and (i) lack footnotes and other presentation items that may apply for a stand-alone business and (ii) are subject to certain normal year end adjustments.
- (b) To the Knowledge of IRP GP and Resource Partners, there (i) are no significant deficiencies or material weakness in the design or operation of internal controls which are reasonably likely to adverse affect in any material respect the ability of the Acquired Companies considered as a whole to record, process, summarize and report financial data, (ii) is no fraud, whether or not material, that involves management or other employees who have a significant role in any Acquired Company's internal controls, (iii) Resource Partners' transactions are executed in accordance with management's general or specific authorization; (iv) Resource Partners' transactions are recorded as necessary to permit the preparation of financial statements in conformity with GAAP and to maintain accountability for assets; and (v) the amount recorded for assets on the respective books and records of Resource Partners is compared with the existing assets at reasonable intervals in connection with the preparation of the annual audits of Resource Partners' consolidated financial statements and appropriate action is taken with respect to any differences.
- (c) Since January 1, 2008, none of the Acquired Companies nor any of their respective Subsidiaries nor, to the Knowledge of IRP GP and Resource Partners, any director, officer, employee, auditor, accountant or representative of any Acquired Company or any of its respective Subsidiaries has received or otherwise had or obtained Knowledge of any material complaint, allegation, assertion or claim, whether written or oral, regarding the accounting or auditing practices, procedures, or methodologies of any Acquired Company or its respective Subsidiaries or their respective internal accounting controls, including any material complaint, allegation, assertion, or claim that any Acquired Company or any of its respective Subsidiaries has engaged in questionable accounting or auditing practices.
- 3.5 <u>Absence of Undisclosed Liabilities</u>. Except as reflected in the Latest Balance Sheet or on <u>Schedule 3.5</u>, no Acquired Company has any Liability of any nature whatsoever (direct or indirect, matured or unmatured, absolute, accrued, or contingent), except as would not be material to any of the Acquired Companies, that would be required by GAAP to be included in the Latest Balance Sheet or reflected in footnotes to a balance sheet prepared in accordance with GAAP as consistently applied or that would prevent Sellers or any of the Acquired Companies from consummating the Contemplated Transactions.

3.6 No Conflicts or Approvals. Except as set forth on Schedule 3.6 or Schedule 3.7, the execution, delivery and performance by Sellers of this Agreement and the other Transaction Documents and the consummation by Sellers of the Contemplated Transactions does not and will not (a) violate, conflict with or result in a breach by Sellers or any of the Acquired Companies of their respective Organizational Documents, (b) require any Consent or other action by any Person, cause or permit the termination, cancellation, acceleration or other change of any right or obligation or violate, conflict with or result in a material breach of, or constitute a default by Sellers or any of the Acquired Companies (with or without notice or lapse of time) under any Material Contract, Permit or Equipment Lease to which any of the Acquired Companies or any or or result in a material breach of, or constitute a default by Sellers or any of the Acquired Companies (with or without notice or lapse of time) under any Real Property Lease requiring lessor's consent for change of control of Resource Partners to which any of the Acquired Companies or any Acquired Companies' properties or assets may be bound, (d) assuming compliance with applicable requirements under the HSR Act, violate, conflict with, or result in a material breach of any Governmental Order or Law applicable to Sellers or any of the Acquired Companies or any of their respective properties or assets or (e) result in the creation of any Encumbrance, other than Permitted Encumbrances, upon any of the properties or assets of any of the Acquired Companies.

3.7 Governmental Authorization. Assuming compliance with applicable requirements under the HSR Act, the execution, delivery and performance by Sellers and each of the Acquired Companies of this Agreement and the other Transaction Documents and the consummation of the Contemplated Transactions, except for the actions and filings listed on Schedule 3.7, the filings required for the transfer of control of the Permits set forth on Schedule 3.8(b) (i) require no action by, or in respect of, any Governmental Authority or any Governmental Approval, and any such action or filing as to which the failure to make or obtain would not be material to any of the Acquired Companies or prevent Sellers or any of the Acquired Companies from consummating the Contemplated Transactions, (ii) will not give any Governmental Authority the right to challenge any portion of the Contemplated Transactions or exercise any remedy or obtain any relief under any Law to which Sellers or any Acquired Company is subject, and (iii) will not contravene, conflict with or result in a violation of any of the terms or requirements of, or give any Governmental Authority the right to revoke, withdraw, suspend, cancel, not renew, terminate or materially modify, any Permit of an Acquired Company.

3.8 Compliance; Permits; Seller Bonds

(a) Each of the Acquired Companies has conducted its respective business and developed and operated its assets in compliance in all material respects with all Permits, except as set forth on Schedule 3.8(a).

- (b) Each Acquired Company possesses all material Permits necessary to own, lease and operate its assets and lawfully conduct its business consistent in all material respects with past practices. Schedule 3.8(b) contains an accurate and complete list and summary description of all such Permits, in each case setting forth the name of the granter, the name of the grantee, and the expiration and renewal date of, and the amount of bonding obligation with respect to each Permit. Except as set forth on Schedule 3.8(b), each such Permit has been validly issued and is in full force and effect and is final and, to the Knowledge of IRP GP and Resource Partners, no event has occurred that constitutes or, with the giving of notice, the passage of time, or both, would constitute a material default by any Acquired Company or any other Person under any such Permit. All fees and other payments due and owing in connection with such Permits have been paid in full and in a timely manner so as to prevent any lapse, revocation or material breach thereof.
- (c) There are no applications for permits to be issued by Governmental Authorities to any Acquired Company other than those set forth or Schedule 3.8(c) (the "Applications"), and an Acquired Company will, on grant of any of such Applications, hold a legal or beneficial title to the interest in each such application as set forth on Schedule 3.8(c). Each Application has been made in accordance with applicable Law. No Acquired Company has received any written communication from the applicable permitting authority with respect to any Application that (i) indicates that such Application has been denied or (ii) requests any Acquired Company to provide additional information with respect to such Application.
- (d) The Acquired Companies have posted all deposits, letters of credit, trust funds, bid bonds, performance bonds, reclamation bonds and surety bonds (and all such similar undertakings) required to be posted in connection with their operations. The Seller Bonds are sufficient to conduct the business of the Acquired Companies consistent with past practices. Except as disclosed on Schedule 3.8(d): (i) each of the Acquired Companies is and has been in compliance in all material respects with all Seller Bonds applicable to it; and (ii) the operation of each Company's coal mining and processing operations and the state of reclamation with respect to Seller Bonds are "current" or in "deferred status" regarding reclamation obligations and otherwise are and have been in compliance in all material respects with all applicable mining, reclamation and other analogous Laws, except where noncompliance would not interfere in any material respect with the ability of any Acquired Company to continue to operate its assets and conduct its business as currently conducted and would not prevent Sellers or any Acquired Company from consummating the Contemplated Transactions.
- (e) Each Acquired Company is, and at all times has been since June 12, 2007, in material compliance with all Laws that are or were applicable to it or to the use or operation of its assets, including Occupational Health and Safety Laws.
- (f) To the Knowledge of IRP GP and Resource Partners, no event has occurred or circumstance exists that (with or without notice or lapse of time) (i) may constitute or result in a material violation by any Acquired Company of any Law applicable to it or its business or assets, (ii) constitute or result directly or indirectly in a violation of or a failure to comply with any material term or requirement of, or result directly or indirectly in the revocation, withdrawal, suspension, cancellation, non-renewal, termination of or modification to, or result in the imposition of any condition on, any Permit listed or required to be listed on Schedule 3.8(b), or (iii) may give rise to any obligation on the part of any Acquired Company to undertake, or to bear all or any portion of the cost of, any remedial action of a material nature.

- (g) All applications required to have been filed for the renewal of the Permits listed or required to be listed on Schedule 3.8(b) have been duly filed on a timely basis with the appropriate Governmental Authority, and all other filings required to have been made with respect to such Permits have been duly made on a timely basis with the appropriate Governmental Authority.
- (h) Except as set forth in Schedule 3.8(h), no Acquired Company has received, at any time since June 12, 2007, any notice or other written communication or, to the Knowledge of IRP GP and Resource Partners, any oral communication from any Governmental Authority or any other Person regarding (i) any actual, alleged, possible or potential violation of, or failure to comply in any material respect with, any Law or failure to comply with any term or requirement of any Permit, (ii) any actual, proposed, possible, or potential revocation, withdrawal, suspension, cancellation, non-renewal, termination of, or modification to any Permit, or (iii) any actual, alleged, possible, or potential obligation on the part of an Acquired Company to undertake or to bear all or any portion of the cost of, any remedial action in excess of \$100,000.
- (i) The execution, delivery and performance by the Sellers of this Agreement and the other Transaction Documents and the consummation of the Contemplated Transactions does not and will not violate, conflict with or result in a breach by the Sellers or any Acquired Company of any Permit or Application, or result in any termination, modification or non-renewals thereof.

3.9 Proceedings; Governmental Orders.

- (a) Except as set forth on Schedule 3.9(a), there is no pending Proceeding:
 - (i) that has been commenced by or against any Acquired Company, its properties or other assets, or otherwise relates to or may affect an Acquired Company;
 - (ii) that challenges, or that may have the effect of preventing, delaying, making illegal, or otherwise interfering with any of the Contemplated Transactions; or
 - (iii) that is in respect of any Permit.

To the Knowledge of IRP GP and Resource Partners, other than as set forth on Schedule 3.9(a), (1) no such Proceeding has been threatened (whether orally or in writing), and (2) no event has occurred or circumstance exists that may give rise to or serve as a basis for the commencement of any such Proceeding. Resource Partners has delivered to Buyer copies of all pleadings, non-privileged correspondence, and other non-privileged documents relating to each Proceeding listed on Schedule 3.9(a). Except as set forth on Schedule 3.9(a), none of such Proceedings, if adversely determined, would reasonably be expected to involve, individually or in the aggregate, an award of damages in excess of \$250,000.

(h)	Excent	as set	forth	on Sc	hedule	39	(h).

- (i) there is no Governmental Order relating to any Acquired Company or that any Acquired Company is subject to;
- (ii) no Related Person of any Acquired Company is subject to any Governmental Order that would impair or prevent the Consummated Transaction or that would impair or to the Knowledge of IRP GP and Resource Partners prevent Buyer's use, conduct or activity relating to the performance of the business of the Acquired Companies after Closing in substantially the same manner as the Acquired Companies have conducted such business prior to Closing;
- (iii) each Acquired Company is and at all times has been in full compliance with all of the terms and requirements of each Governmental Order to which it has been subject, and no event or circumstance exists that may constitute or result in (with or without notice or lapse of time) a violation of or failure to comply with any Governmental Order to which any Acquired Company is subject; and
- (iv) no Acquired Company has received any notice or other communication (whether oral or written) from any Governmental Authority or any other Person regarding any actual, alleged, possible, or potential violation of, or failure to comply with, any term or requirement of any Governmental Order to which any Acquired Company is bound.
- (c) Except as set forth in Schedule 3.9(c), to the Knowledge of IRP GP and Resource Partners, there are no existing claims by or disputes between any Acquired Company and Persons owning, controlling or occupying lands or reality adjourning or near any of the real property and or reserves of such Acquired Company or regarding adverse possession, the location of boundary lines, encroachments, miner rights, subsidence, water quality or quantity, flood damage, blasting damage, trespass, waste, transportation of coal or other materials, nuisances or other similar matter.
 - 3.10 Absence of Certain Changes. Except as disclosed in Schedule 3.10 or as otherwise contemplated by this Agreement, since December 31, 2010:
- (a) there has not been any condition, change or event specific to the Acquired Companies which has had, or would reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect;
 - (b) the business of each Acquired Company has been conducted only in the Ordinary Course of Business;
- (c) the assets and properties of the Acquired Companies have been operated and maintained in the Ordinary Course of Business and there has not been any Material Adverse Effect;
- (d) there has not been any material damage, destruction or loss (whether or not covered by insurance) affecting any material portion of the assets or properties of the Acquired Companies;

- (e) there has not been any sale, transfer, lease, abandonment, or other disposal of any material property or equipment of the Acquired Companies, except in the Ordinary Course of Business; and
- (f) none of the Sellers or the Acquired Companies has taken (or caused or permitted any of the Acquired Companies to take) any action which, if it had been taken by any of the Sellers or any of the Acquired Companies after the date hereof, would have required Buyer's approval or waiver under Section 6.1(b)(iii).

3.11 <u>Tax Matters</u>. Except as set forth in <u>Schedule 3.11</u>:

- (a) All Tax Returns required to be filed by or on behalf of each Acquired Company prior to the Closing Date (separately or as part of a consolidated, combined or unitary group) have been or shall be timely filed (subject to permitted extensions applicable to such filing), all such Tax Returns were materially correct and complete in all respects and all Taxes due (whether or not such Taxes are shown or required to be shown on a Tax Return) have been or shall be paid within the prescribed period or any extension thereof, other than Taxes that are being contested in good faith for which adequate accruals or reserves have been established on the Balance Sheet.
- (b) There are no Encumbrances relating to Taxes encumbering any of the assets of the Acquired Companies, the Partnership Interests or the Membership Interests to be sold pursuant to this Agreement.
- (c) Other than in connection with routine extensions of time for filing Tax Returns, none of the Acquired Companies have granted any extension or waiver of the statute of limitations period applicable to any Tax or Tax Return, or agreed to any extension of time with respect to a Tax assessment or deficiency, which period has not yet expired.
- (d) No claim has ever been made by a Governmental Authority in a jurisdiction where any of the Acquired Companies do not file Tax Returns that any of them are or may be subject to taxation in that jurisdiction.
- (e) None of the Acquired Companies are a party to, are bound by or have any obligation under, any Tax sharing agreement or similar contract or arrangement or any indemnification agreement or any other agreement that obligates them to make any payment computed by reference to Taxes, taxable income or taxable losses of any other Person.
- (f) Since June 12, 2007, none of the Acquired Companies (i) have been a member of an Affiliated Group filing a consolidated federal Income Tax Return (other than the Affiliated Group of which they are currently members) or (ii) have any Liability for the Taxes of any Person (other than the Acquired Companies) under Treasury Regulation Section 1.1502-6 (or any similar provision of state, local of foreign law), as transferees or successors, by contract, or otherwise.
- (g) None of the Acquired Companies have distributed stock of another Person, or have had their stock distributed by another Person, in a transaction that was purported or intended to be governed in whole or in part by Section 355 or 361 of the Code.

- (h) Resource Partners is, and has been since its formation, classified as a partnership, within the meaning of Treasury Regulation Section 301.7701-3 (and similar provisions of state and local Tax Law) and Resource Partners does not own, directly or indirectly, any interest in an entity that is not an entity disregarded from its owner within the meaning of Treasury Regulation Section 301.7701-3 (and similar provisions of state and local Tax Law).
 - (i) Resource Partners has in effect a valid election under Section 754 of the Code (and similar applicable provisions of state and local Tax Law).
- (j) There is no contract or agreement, plan or arrangement by any of the Acquired Companies covering any Person that, individually or collectively, could give rise to the payment of any amount that would not be deductible by any of the Acquired Companies by reason of Section 280G of the Code.
- (k) Each of the Acquired Companies has timely withheld and paid over to the appropriate taxing authority all Taxes that it is required to withhold from amounts paid or owing to any employee, shareholder, creditor or other Person and has properly completed and timely filed all Forms W-2 and 1099 (and similar forms) required with respect thereto.
- (l) No foreign, federal, state or local Tax audits or assessments or administrative or judicial proceedings are ongoing, pending or to the Knowledge of IRP GP, Resource Partners or James Thornburg, former Tax Manager of International Industries and current Tax Manager of Resource Partners, threatened with respect to the Sellers or any Acquired Company, and neither Sellers nor any Acquired Company has received from a Governmental Authority any request for information related to any Tax matter, and neither the Sellers nor any Related Person (or employee responsible for Tax matters) of the Acquired Companies expects any authority to assess any additional Taxes for or with respect to any period for which Tax Returns have been filed.
- (m) Neither the Sellers nor any of the Acquired Companies is or has been a party to any "reportable transaction," as defined in Section 6707A(c)(1) of the Code and Treasury Regulation Section 1.6011-4(b).
- (n) Each of the Acquired Companies is classified as an entity disregarded from its owner within the meaning of Treasury Regulation Section 301.7701-3 (and similar provisions of state and local Tax law) and (ii) no Acquired Companies owns, directly or indirectly, any interest in an entity that is not an entity disregarded from its owner within the meaning of Treasury Regulation Section 301.7701-3 (and similar provisions of state and local Tax Law).

3.12 Employee Benefits.

(a) <u>Schedule 3.12(a)</u> sets forth a list of each employee benefit plan (as defined in Section 3(3) of the Employee Retirement Income Security Act of 1974, as amended ("<u>ERISA</u>")), each employment agreement, and each other plan, fund, agreement, or arrangement providing bonuses, incentive compensation, profit sharing benefits, pension benefits, compensation, deferred compensation, equity compensation, welfare benefits, fringe benefits, severance payments, post-retirement benefits, scholarships, disability benefits, sick leave pay, vacation pay, commissions, payroll practices, retention payments, change in control, or other benefits, maintained or contributed to by, or with respect to which there is or could be any Liability or obligation of, any Acquired Company or any trade or business, whether or not incorporated (an "<u>ERISA Affiliate</u>"), which together with any Acquired Company would be deemed a "single employer" within the meaning of Section 4001(b)(1) of ERISA (each a "<u>Benefit Plan</u>" and collectively, the "<u>Benefit Plans</u>"), for the benefit of current or former employees, officers, directors, independent contractors and consultants of the Acquired Companies (<u>Company Employees</u>").

(b) Except as set forth in Schedule 3.12(b), with respect to each Benefit Plan: (i) if intended to qualify under Section 401(a) of the Code, such plan has received a current, favorable determination letter from the applicable Governmental Authority stating that it so qualifies and that its trust is exempt from taxation under Section 501(a) of the Code and nothing has occurred since the date of such determination that could adversely affect such qualification or exempt status; (ii) such plan has been administered in all material respects in accordance with its terms and applicable Law; (iii) no disputes are pending, or, to the Knowledge of IRP GP and Resource Partners, threatened that give rise to or might reasonably be expected to give rise to material Liability on the part of the Acquired Companies; (iv) no non-exempt prohibited transaction (within the meaning of Section 406 of ERISA) has occurred that gives rise to or would reasonably be expected to give rise to Liability on the part of the Acquired Companies; (v) all contributions, reserves or premium payments required to be made or accrued to or with respect to such plan as of the date hereof (taking into account any extensions of time for the making of such contributions) have been timely made or accrued in full and all contributions, reserves or premium payments required to be made or accrued between the date hereof and the Closing will be made or accrued; (vi) Resource Partners has provided or made available to Buyer complete and correct copies of each such Benefit Plan and any related trust or other funding agreement, any amendment, any summary plan descriptions, the most recent annual report (Form 5500) and the most recent actuarial valuation with respect thereto; (vii) each Benefit Plan which after Closing covers any employee of an Acquired Company may be amended or terminated unilaterally by an Acquired Company at any time without Liability other than for benefits accrued or incurred prior to the date of such amendment or termination; and (viii) each

- (c) Except as set forth in Schedule 3.12(c) and except for Liabilities for premiums due to the Pension Benefit Guaranty Corporation (PBGC") in the ordinary course, no Liability has been incurred by the Acquired Companies or any ERISA Affiliate under or pursuant to Title IV of ERISA, relating to its or their employee benefit plans, and no event, transaction or condition has occurred that has resulted in any such Liability to the Acquired Companies or any ERISA Affiliate or any employee benefit plan of the Acquired Companies or any ERISA Affiliate. No Acquired Company and no ERISA Affiliate currently maintains, contributes to or participates in, nor at any time in the past, has maintained, contributed to, participated in, or had any obligation to maintain, contribute to, or otherwise participate in, any plan that is (i) subject to the provisions of Title IV of ERISA, (ii) a "multiple employer plan" within the meaning of Section 4001(a)(3) of ERISA, (iii) a "multiple employer" plan within the meaning of Section 412 of the Code, or (iv) a multiple employer welfare arrangement within the meaning of Section 3(40) of ERISA. Except as set forth on Schedule 3.12(c) or contemplated by this Agreement, the execution of this Agreement and the consummation of the Contemplated Transactions will not (either alone or upon the occurrence of any additional or subsequent event) give rise to any Liability for any employee benefits, including Liability for severance pay, unemployment compensation, termination pay or withdrawal Liability, or accelerate the time of payment or vesting or increase the amount of compensation or benefits due to any Company Employee.
- (d) The Acquired Companies have complied with the applicable requirements of Part 6 of Subtitle B of Title I of ERISA and Section 4980B of the Code ("COBRA"), and will comply with all COBRA obligations arising in connection with the Contemplated Transactions, and the Acquired Companies are not subject to any Liability as a result of any failure to administer or operate any "group health plan" (as defined in COBRA and/or as defined in 45 Code of Federal Regulations Section 160.103) in compliance with COBRA and/or the applicable requirements of the Health Insurance Portability and Accountability Act of 1996 and the Regulations promulgated thereunder. Schedule 3.12(d) identifies each Company Employee who has experienced a "Qualifying Event" (as defined in COBRA) with respect to a Benefit Plan who has elected or is eligible to elect "Continuation Coverage" (as defined in COBRA) and whose maximum period for Continuation Coverage required by COBRA has not expired. Schedule 3.12(d) also lists the name of each Covered Employee who is on a leave of absence (whether or not pursuant to the Family and Medical Leave Act of 1993, as amended ("FMLA")) and is receiving or entitled to receive health coverage under an Employee Benefit Plan, whether pursuant to FMLA, COBRA or otherwise.
- (e) No Benefit Plan provides and no promises have been made to provide medical, surgical, hospitalization, death or similar benefits (whether or not insured) for current or former employees, directors, officers, consultants, independent contractors, contingent workers, or leased employees (or any of their dependents, spouses or beneficiaries) for periods extending beyond their retirement or other termination of service, other than continuation coverage mandated by applicable Law and only to the extent required under such applicable Law.

3.13 Employees; Labor and Employee Relations.

(a) Schedule 3.13(a) lists all current employees of each of the Acquired Companies as of the Closing Date, and for each such employee: (i) job position; (ii) job location; (iii) classification as full-time or part-time; (iv) classification as exempt or non-exempt under applicable state or federal overtime regulations; (v) hourly rate of compensation or base salary (as applicable); (vi) total 2009 and 2010 compensation; (vii) target incentive compensation for 2011 (including commission and/or bonus, as applicable); (viii) any other annual compensation or allowance; (ix) vacation and paid time off accrual rate; (x) accrued but unused vacation and paid time off; (xi) last pay review date; (xii) hours of work per week in 2010 (for non-exempt and part-time employees); (xiii) visa type (if any); (xiv) the commencement date of employment and any breaks in service; and (xv) the entity by which such employee is employed. Schedule 3.13(a) also discloses all open workers' compensation claims against any Acquired Company. The records of each Acquired Company accurately reflect in all material respects, the employment or service histories of its employees, independent contractors, contingent workers and leased employees, including their hours of service, and all such records are maintained in a usable electronic form, and each person who is classified on such records as an employee or independent contractor is properly so classified. All employees are employees at-will, terminable without notice and without penalty except as otherwise provided in the employment agreements listed on Schedule 3.12(a).

- (b) Each Acquired Company has complied in all material respects and is presently in compliance in all material respects with all statutes, laws, ordinances, rules or regulations, or any orders, rulings, decrees, judgments or arbitration awards of any court, arbitrator or any government agency relating to employment, equal opportunity, nondiscrimination, immigration, wages, employee exemption status, hours, benefits, collective bargaining, the payment of social security and similar taxes, income tax withholding, occupational safety and health, and/or privacy rights of employees. Each Acquired Company currently, and for each of the prior three (3) years, has completed Form I-9s on file with respect to each of its employees. No key employee or group of employees of any Acquired Company has indicated any plans to terminate employment with the Acquired Company before or after the Closing.
- (c) With respect to each Company Employee, the applicable Acquired Company: (i) has withheld and reported all amounts required by law or by agreement to be withheld and reported with respect to wages, salaries and other payments since June 12, 2007 preceding the Closing Date, (ii) is not liable for any arrears of wages, severance pay or any Taxes or any penalty for failure to comply with any of the foregoing since June 12, 2007 preceding the Closing Date, and (iii) is not liable for any payment to any trust or other fund governed by or maintained by or on behalf of any governmental authority, with respect to unemployment compensation benefits, social security or other benefits or obligations for employees (other than routine payments to be made in the normal course of business and consistent with past practice). No Acquired Company has any Liability with respect to any misclassification of any person as (x) an independent contractor rather than as an employee, or with respect to any employee leased from another employer, or (y) an employee exempt from state or federal overtime regulations.
- (d) Except as set forth in Schedule 3.13(d), (i) none of the Acquired Companies are presently a party to any written collective bargaining agreement in respect of the Company Employees, (ii) there is no unfair labor practice charge or comparable or analogous complaint pending before any Governmental Authority or, to the Knowledge of IRP GP and Resource Partners, threatened against any Acquired Company, (iii) there is no, nor has there been in the past three (3) years any, written grievance, order, dispute, arbitration hearing, or arbitration award pending or filed or, to the Knowledge of IRP GP and Resource Partners, threatened against any Acquired Company, (iv) none of the Acquired Companies are in breach of any order by a Governmental Authority relating to the Company Employees, (v) within the past three (3) years, there has been no labor strike, slowdown, work stoppage, or lockout in effect, or, to the Knowledge of IRP GP and Resource Partners, threatened against or otherwise affecting any Acquired Company or the Company Employees at any of their respective mines, (vi) none of the Acquired Companies are a party to, or otherwise bound by, any consent decree, order, award with, or citation by, any Governmental Authority relating to employees or employment practices, (vii) there is no actual or, to the Knowledge of IRP GP and Resource Partners, alleged breach of the material terms of any contractual obligations governing the employment of the Company Employees which would reasonably be likely to lead to the interference in any material respect with the conduct of the business of any of the Acquired Companies as currently conducted, and (viii) as of the date hereof, there is no formal legal proceeding (i.e. petition for certification of representation) to organize Company Employees which is pending or, to the Knowledge of IRP GP and Resource Partners, threatened.

3.14 Intellectual Property. Schedule 3.14 lists all patents, patent applications, material unregistered trademarks, registered trademarks, copyright applications and registrations, and internet domain names, owned by or registered in the name of any Acquired Company or otherwise used in the conduct of the business of any Acquired Company. Except (a) as set forth in Schedule 3.14, or (b) as would not interfere in any material respect with the conduct of the business of any of the Acquired Companies as currently conducted, one or more of the Acquired Companies legally or beneficially owns or has the right to use, transfer and license all Intellectual Property necessary for the conduct of the business of the Acquired Companies as it is currently conducted. To the Knowledge of IRP GP and Resource Partners, the operation of the business of any of the Acquired Companies does not infringe or otherwise violate any Intellectual Property of any other Person and, to the Knowledge of IRP GP and Resource Partners none of the Intellectual Property owned by any of the Acquired Companies is being infringed by any other Person.

3.15 Contracts.

- (a) <u>Schedule 3.15(a)</u> sets forth a complete list of each of the following contracts to which any Acquired Company is a party or by which any of them is bound as of the date of this Agreement (collectively, the "<u>Material Contracts</u>"):
- (i) any option, purchase and sale contract or lease (whether real or personal property) providing for annual payments of \$150,000 or more or that cannot be terminated on not more than thirty (30) days' notice without payment by any Acquired Company of any penalty;
- (ii) contracts involving the annual expenditure by any Acquired Company of more than \$150,000 in any instance for the purchase of materials, goods, supplies, equipment or services, excluding any such contracts that are terminable by the Acquired Companies without penalty on not more than thirty (30) days' notice;
- (iii) contracts providing for payments to any Acquired Company of more than \$150,000 in any instance for the sale of natural gas, materials, goods, supplies, equipment or services, excluding any such contracts that are terminable by the Acquired Companies without penalty on not more than thirty (30) days' notice;
 - (iv) contracts involving the annual expenditure by any Acquired Company of more than \$150,000 for the purchase, sale, transportation or storage of coal;

- (v) any agreement relating to Indebtedness for borrowed money or the deferred purchase price of property (whether incurred, assumed, guaranteed or secured by any asset), including indentures, mortgages, loan agreements, security agreements, or other agreements for the incurrence of debt, other than (A) trade accounts payable incurred in the Ordinary Course of Business and (B) any such agreement relating to indebtedness owed to Sellers or any of their Affiliates to be repaid on or before the Closing Date or owed to any Acquired Company;
 - (vi) partnership, limited liability company, joint venture agreements or other agreements involving a sharing of profits or expenses by any Acquired Company;
- (vii) any agreement under which (A) any Person (including any Seller) has directly or indirectly guaranteed any liabilities or obligations of any Acquired Company (other than any such guarantee by any other Acquired Company) or (B) any Acquired Company has, directly or indirectly, guaranteed any liabilities or obligations of any other Person (including any Seller but excluding any other Acquired Company);
- (viii) any agreement prohibiting or limiting the ability of any Acquired Company to engage in any business activity or compete with any Person or prohibiting or limiting the ability of any Person to compete with any Acquired Company;
- (ix) any agreement relating to the acquisition or disposition of any business (whether by merger, sale of stock, sale of assets or otherwise), including any contract under which any Acquired Company will have Liabilities after the date of this Agreement relating to the acquisition or sale of any business enterprise;
- (x) distributor, dealer, sales agency, marketing or similar contracts under which any Acquired Company is obligated to pay after the date of this Agreement an amount in excess of \$100,000 during any calendar year;
- (xi) any other contract providing that any Acquired Company will receive future payments aggregating more than \$100,000 per annum or \$500,000 in the aggregate prior the expiration of such contract;
- (xii) any contract with any current or former officer, director or employee of any Acquired Company or any of the Sellers involving annual consideration or payments in excess of \$150,000, including offer letters with respect to employment scheduled to begin after the date hereof;
- (xiii) any consulting or similar agreement with an independent contractor providing for (A) annual payments by any Acquired Company in excess of \$100,000 or (B) aggregate payments by any Acquired Company of \$250,000, excluding any such contracts that are terminable by the Acquired Companies without penalty on not more than thirty (30) days notice;
 - (xiv) any outstanding power-of-attorney empowering any Person not a current employee of any Acquired Company to act on behalf of any Acquired Company;

- (xv) any employee collective bargaining agreement with any labor union or employees covering former, current or future employees of any Acquired Company or work done, being done or to be done in the future by any Acquired Company;
 - (xvi) any contract mining agreement; and
 - (xvii) any material agreement, commitment, arrangement or plan not made in the Ordinary Course of Business.
- (b) Each Material Contract is a valid and binding agreement of each Acquired Company which is a party thereto and, to the Knowledge of IRP GP and Resource Partners, each of the other parties thereto, enforceable by or against such Acquired Company and, to the Knowledge of IRP GP and Resource Partners, each of such other parties thereto in accordance with its terms, except as such enforceability may be limited by applicable bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium and other similar Laws relating to or affecting creditors' rights generally and general equitable principles (whether considered in a proceeding in equity or at law). Resource Partners has heretofore delivered to Buyer true and complete copies of all such written Material Contracts. Except as set forth in Schedule 3.15(b), none of the rights of the Acquired Companies under the Material Contracts have been assigned (including by an absolute assignment of rents or contracts) or collaterally assigned, assigned for the purpose of granting security, or are affected by any security interest or similar encumbrance. Except as set forth in Schedule 3.6, none of the Material Contracts require consent to consummate the Contemplated Transactions, whether by operation of law or otherwise.
- (c) Except as set forth on Schedule 3.15(c), (i) the applicable Acquired Company is, and at all times has been, in compliance in all material respects with all applicable terms and requirements of each Material Contract, (ii) to the Knowledge of IRP GP and Resource Partners, each other Person that has had any obligation or Liability under any Material Contract is, and at all times has been, in material compliance with all applicable terms and requirements of such Material Contract, (iii) to the Knowledge of IRP GP and Resource Partners no event has occurred or circumstance exists that (with or without notice or lapse of time) may contravene, conflict with, or result in a violation or breach of, or give the Acquired Companies, or any other Person the right to declare a default or exercise any remedy under, or to accelerate the maturity or performance of, or to cancel, terminate, or modify, any Material Contract, and (iv) no Acquired Company has been given or received from any Person at any time since January 1, 2009, any written notice or other written communication or, to the Knowledge of IRP GP and Resource Partners, oral notice or other oral communication regarding any actual, alleged, possible, or potential violation or breach of, or default under, any Material Contract.
 - 3.16 Environmental Matters. Except as set forth on Schedule 3.16:

(i) each of the Acquired Companies is in compliance in all material respects with all Environmental Laws and any Permits required pursuant to any Environmental Law and has disclosed to Buyer on <u>Schedule 3.16</u> any material non-compliance of any Acquired Company of any Environmental Laws and Permits which has occurred since June 12, 2007:

- (ii) none of the Acquired Companies are currently subject to any written, or to the Knowledge of IRP GP and Resource Partners oral, Environmental Claim or written, or to the Knowledge of IRP GP and Resource Partners oral, notice of any threatened Environmental Claim, resulting from or arising under or pursuant to any Environmental Law, that would be material to any Acquired Company, regarding or resulting from activities of any Acquired Company, the business of any Acquired Company or any property or assets currently owned, operated or used by any Acquired Company;
- (iii) none of the Acquired Companies have entered into, have agreed to, or are subject to, any Governmental Order under any Environmental Law regarding any Acquired Company or any property or assets currently or formerly owned, operated or used by any Acquired Company, that would interfere in any material respect with the ability of such Acquired Company to continue to operate its assets and conduct its business as currently conducted or would result in any Liability to any Acquired Company;
- (iv) no property currently owned or operated by any Acquired Company (A) is listed or, to the Knowledge of IRP GP and Resource Partners, proposed for listing on the CERCLA National Priorities List or CERCLIS list or any similar Governmental Authority's list of sites at which remedial action is or may be necessary or (B) to the Knowledge of IRP GP and Resource Partners, contains asbestos or asbestos-containing materials, or polychlorinated biphenyls that is in a condition constituting a violation of Environmental Law, would reasonably be expected to result in material Liability under Environmental Law or is in a damaged or friable condition; and
- (v) to the Knowledge of IRP GP and Resource Partners, there is no block or punitive legal limitation on any Acquired Company's ability to apply for or obtain any Permit under any Environmental Law;
- (vi) no Acquired Company has received any written citation, directive, inquiry, notice, Governmental Order, warning or other written communication or, to the Knowledge of IRP GP and Resource Partners, oral citation, directive, inquiry, notice, Governmental Order, warning or other communication that relates to Hazardous Materials or any alleged, actual, or potential violation or failure to comply with any Environmental Law, or of any alleged, actual, or potential obligation to undertake or bear the cost of any Environmental Losses;
 - (vii) no Acquired Company has experienced any material Environmental Losses since January 1, 2008;
- (viii) since January 12, 2010, there has been no material Release nor is there any threat of material Release at or from any equipment, real property, leaseholds, or other interests currently or formerly owned, leased or operated by LMR, or at any other location where any Hazardous Materials were generated, manufactured, refined, transferred, produced, imported, used, or processed from, by, for or to LMR;
- (ix) since June 12, 2007, there has been no material Release nor is there any threat of material Release at or from any equipment, real property, leaseholds, or other interests currently or formerly owner, leased or operated by any Acquired Company (other than LMR), or at any other location where any Hazardous Materials were generated, manufactured, refined, transferred, produced, imported, used, or processed from, by, for or to any Acquired Company (other than LMR); and

(x) to the Knowledge of IRP GP and Resource Partners, there has been no environmental investigation, study, audit, test, review or other analysis conducted since June 12, 2007, with respect to the activities conducted by any of the Acquired Companies on Real Property required under a Permit or any Environmental Law that has not been delivered or made available to Buyer prior to the date hereof.

3.17 <u>Insurance</u>. <u>Schedule 3.17</u> lists all insurance policies held in the name of any of the Acquired Companies covering the assets and operations of any Acquired Company and any Company Employees as of the date hereof, specifying the policies' numbers, terms, the insurer, amount of coverage, annual premiums and type of insurance (including, but not limited to details regarding Broad-Form Employers Liabilities (Mandolidis) coverage and Federal Black Lung Workers Compensation since January 1, 2000). Resource Partners has delivered to Buyer true and complete copies of all insurance policies listed on <u>Schedule 3.17</u>. Except as set forth on<u>Schedule 3.17</u>, all such policies are in full force and effect, on an occurrence basis, all premiums due thereon have been paid and, where applicable, each Acquired Company has complied in all material respects with the provisions of such policies and have not received any notice from any of its insurance brokers or carriers that such broker or carrier has cancelled or terminated coverage or will not be willing or able to renew their existing coverage or reserved rights with respect to any outstanding claim. All insurance policies not held in the name of any Acquired Company but which cover the assets and operations of the business of any Acquired Company and any Company Employees are in full force and effect (subject to changes made in the Ordinary Course of Business that will not materially reduce the coverage thereunder) and will remain in full force and effect until the Closing, at which time, coverage thereunder will be discontinued with respect to the Acquired Companies and the business of the Acquired Companies. As of the date hereof, there are no claims outstanding under any insurance policy described in this Section 3.17 and relating to the business of any Acquired Company, and, to the Knowledge of IRP GP and Resource Partners, no event has occurred which might give rise to any claim in excess of \$100,000. There is no claim by any Acquired Company pending under any su

3.18 Personal Property Assets.

(a) Schedule 3.18 sets forth a true and complete list of all the material items of machinery, equipment, vehicles, and other tangible personal property now owned or leased by any Acquired Company (the "Material Personal Property"), which list represents all such property used in connection with the businesses of the Acquired Companies and is sufficient to operate the businesses of the Acquired Companies consistent with past practices. Except as set forth on Schedule 3.18, each Acquired Company has good and marketable title to, or holds by a valid and existing lease or license for, all the Material Personal Property, except with respect to assets disposed of in the Ordinary Course of Business since such date, free and clear of all Encumbrances, other than Permitted Encumbrances. Each item of the Material Personal Property is in working order and repair (normal Industry wear and tear excepted), has been operated and maintained in the Ordinary Course of Business and consistent with prudent coal mining industry standards and is in suitable and adequate condition for use consistent with past practices and prudent coal mining industry practices.

(b) Except to the extent reserved for in the Financial Statements, the accounts receivable of each Acquired Company (i) are valid and genuine and (ii) have arisen solely out of bona fide sales and deliveries of goods, performances of services and other business transactions in the Ordinary Course of Business.

(c) Schedule 3.18(c) contains a true and complete list of all of the leases, instruments and other agreements and arrangements pursuant to which any Acquired Company leases or rents any Material Personal Property to or from any other Person other than between or among two (2) or more of the Acquired Companies (collectively, the "Equipment Leases"). Resource Partners has delivered to Buyer available true and complete copies of all of the Equipment Leases. Each Equipment Lease is a valid and binding agreement of each Acquired Company which is a party thereto in accordance with its terms, except as such enforceability may be limited by applicable bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium and other similar Laws relating to or affecting creditor's rights generally and general equitable principles (whether considered in a Proceeding in equity or at law). Except as set forth on Schedule 3.18(c), (i) the applicable Acquired Company is, and at all times has been, in compliance in all material respects with all terms and requirements of each Equipment Lease; (ii) to the Knowledge of IRP GP and Resource Partners, any other Person that has had any obligation or Liability under any Equipment Lease is, and at all times has been, in compliance in all material respects with all terms and requirements of such Equipment Lease, (iii) no event has occurred or circumstance exists that (with or without notice or lapse of time) may contravene, conflict with, or result in a violation or breach of, or give the Acquired Companies, or any other Person the right to declare a default or exercise any remedy under, or to accelerate the maturity or performance of, or cancel, terminate, or modify, any Equipment Lease, and (iv) no Acquired Company has been given or received from any Person at any time since January 1, 2009, any written notice or other communication or, to the Knowledge of IRP GP and Resource Partners, any oral notice or other communication regarding any actual, alleged, possible, or potential vio

3.19 Real Property

(a) <u>Leased Properties</u>. <u>Schedule 3.19(a)(i)</u> lists all material real property and water rights, and other material interests in land, including coal, mining, exploration and surface rights, easements, rights of way and options, leased or subleased by any Acquired Company (the "<u>Leased Real Property</u>") and all of the leases and subleases related thereto (the "<u>Real Property Leases</u>"). Resource Partners has delivered or made available to Buyer true and complete copies of all Real Property Leases. With respect to each Real Property Lease and except as otherwise specified on <u>Schedule 3.19(a)(ii)</u> or where the failure of any of the following to be true and correct would not reasonably be expected to interfere in any material respect with the respective abilities of the Acquired Companies to continue to operate their respective assets and conduct their respective businesses as currently conducted:

- (i) each Real Property Lease is a valid and binding agreement of each Acquired Company that is a party thereto and, to the Knowledge of IRP GP and Resource Partners, each of the other parties thereto, enforceable by or against such Acquired Company and, to the Knowledge of IRP GP and Resource Partners, each of such other parties thereto in accordance with its terms, except as such enforceability may be limited by applicable bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium and other similar Laws relating to or affecting creditors' rights generally and general equitable principles (whether considered in a proceeding in equity or at law);
- (ii) (A) none of the Acquired Companies are in material default under any such Real Property Lease and, to the Knowledge of IRP GP and Resource Partners, no event has occurred which, with the passage of time or expiration of any grace period, would constitute a default of any Acquired Company's obligations under such Real Property Lease, (B) to the Knowledge of IRP GP and Resource Partners, no other party to any such Real Property Lease is in default thereunder and (C) none of the Acquired Companies have received a written or, to the Knowledge of IRP GP and Resource Partners, other notice of default with respect to any such Real Property Lease;
 - (iii) except for Permitted Encumbrances, no such Real Property Lease has been mortgaged, deeded in trust or subjected to an Encumbrance;
- (iv) with regard to Leased Real Property, to the Knowledge of IRP GP and Resource Partners an Acquired Company has adequate rights of ingress and egress to such Leased Real Property and all building structures, facilities, fixtures and other improvements thereon;
- (v) with regard to Leased Real Property, no Acquired Company owes any brokerage or other commissions with respect to any such Real Property Lease for which adequate reserves have not been established on the Latest Balance Sheet; and
- (vi) other than Permitted Encumbrances, there are no other matters that, to the Knowledge of IRP GP and Resource Partners, would adversely affect the rights of any Acquired Company to the Leased Real Property.
- (b) <u>Owned Properties</u>. <u>Schedule 3.19(b)(i)</u> lists all material real property and other interests in land, including coal, mining, exploration and surface rights, easements, rights of way and options, owned by any of the Acquired Companies (the "<u>Owned Real Property</u>", and collectively with the Leased Real Property, the "<u>Real Property</u>"), and the record title owner of the Owned Real Property. Resource Partners has delivered to Buyer true and complete copies of all deeds, and all title insurance policies, title insurance, abstracts and other evidence of title (if any) in the possession of any Acquired Company relating to the Owned Real Property. With respect to each such parcel of the Owned Real Property, except as otherwise specified on <u>Schedule 3.19(b)(ii)</u> or where the failure of any of the following to be true and correct would not reasonably be expected to be materially adverse to the use of such Owned Real Property:

- (i) the identified owner has adequate rights of ingress and egress with respect to, and marketable fee simple title to the parcel of the Owned Real Property, free and clear of any Encumbrances, except for Permitted Encumbrances;
- (ii) there are no pending or, to the Knowledge of IRP GP and Resource Partners, threatened condemnation proceedings, lawsuits, or administrative actions or to the Knowledge of IRP GP and Resource Partners other material matters affecting adversely the current use, occupancy, or value of the Owned Real Property;
- (iii) to the Knowledge of IRP GP and Resource Partners, no material Owned Real Property serves any adjoining property for any purpose inconsistent with the use of the Owned Real Property;
- (iv) to the Knowledge of IRP GP and Resource Partners, there are no leases, subleases, licenses, concessions, or other agreements, written or oral, granting to any Person the right of use or occupancy of any portion of the Owned Real Property;
- (v) no tenant or other Person in possession of any Owned Real Property has any right to purchase or holds any right of first refusal to purchase any such properties; and
 - (vi) other than Permitted Encumbrances, there are no other matters that would adversely affect the title of any Acquired Company.
- (c) All material buildings, structures and other improvements located upon the Real Property, including all material components thereof, are in working order and repair (normal Industry wear and tear excepted), have been operated and maintained in the Ordinary Course of Business and are in suitable and adequate condition for use consistent with past practices except as would not reasonably be expected to have a Material Adverse Effect.
- (d) None of the Acquired Companies have received any written, or to Knowledge of IRP GP and Resource Partners oral, notice of claims, with respect to the Real Property, that any Acquired Company has mined any coal that it did not have the right to mine or mined any coal in such reckless or imprudent fashion as to give rise to any claims for waste or trespass and no facts exist upon which such a claim could be based. Resource Partners has made available to Buyer the most recent complete and, to the Knowledge of IRP GP and Resource Partners, correct version of each of the following items to the extent such items are (i) in the possession of any Acquired Company, (ii) relate to or affect the Real Property, including the coal reserves, coal ownership, mining conditions, mines, and mining, and (iii) material to the use or operation of any assets of any Acquired Company: geological data, reserve data, existing mine maps, surveys, core hole logs and associated data, coal measurements, coal samples, lithologic data, coal reserve calculations or reports, washability analyses or reports, mine plans, mining permit applications and supporting data, engineering studies and all other books and records, information, maps, reports and data.

- (e) There does not exist any pending or, to the Knowledge of IRP GP and Resource Partners, any threatened condemnation action or any other Governmental Action undertaken to enforce any land use that would reasonably be expected to materially affect or materially impair the present use or operation of the Real Property and none of the Real Property has suffered any material damage by fire or other casualty.
- 3.20 No Brokers' or Other Fees. Except for those Persons listed on Schedule 3.20, whose fees and expenses will be paid by Sellers, no Person has acted directly or indirectly as a broker, finder, investment banker or financial advisor in connection with the Contemplated Transactions, and no Person is entitled to any fee or commission or like payment in connection with the Contemplated Transactions based upon any agreement, arrangement or other understanding made by or on behalf of Sellers.
- 3.21 Affiliate Transactions. Except as set forth on Schedule 3.21, since January 1, 2008, there have been no transactions or agreements between any Acquired Company, on the one hand, and any of its Related Persons (including Sellers) or any of their Affiliates, on the other hand, including loans, services agreements, consulting agreements, employment agreements, or business arrangements. None of the Related Persons (including Sellers) or any of their respective Affiliates has entered into or engaged in any transaction with any Acquired Company other than transactions entered into in the Ordinary Course of Business and that are on an arm's length basis.
- 3.22 <u>Customers and Suppliers.</u> <u>Schedule 3.22</u> contains (a) a list of the top ten (10) customers of the Acquired Companies on a consolidated basis (determined on the basis of revenues) for each of the last two (2) fiscal years and (b) a list of the top ten (10) suppliers of the Acquired Companies on a consolidated basis (determined on the basis of cost of items purchased) for fiscal year 2010. No customer set forth on <u>Schedule 3.22</u> has ceased or materially reduced its purchases from or use of the services of any Acquired Company since December 31, 2010, or to the Knowledge of IRP GP and Resource Partners, has threatened to cease or materially reduce such purchases or use after the date hereof. No supplier set forth on <u>Schedule 3.22</u> has ceased or materially reduced its supply of materials or goods or provision of services to any Acquired Company since December 31, 2010, or to the Knowledge of IRP GP and Resource Partners, has threatened to cease or materially reduce such supply or provision after the date hereof. To the Knowledge of IRP GP and Resource Partners, no such customer or supplier is currently in, or threatened with, bankruptcy or insolvency.
- 3.23 Absence of Certain Business Practices. No Acquired Company, nor any of their Related Persons, or any Person acting on behalf of any Acquired Company has, directly or indirectly, within the past two (2) years given or agreed to give any gift or similar benefit to any customer, supplier, governmental employee or other Person who is or may be in a position to help or hinder the respective business of such Acquired Company (or assist such Acquired Company in connection with any actual or proposed transaction) that (a) might subject any Acquired Company to any material damage or material penalty assessed by any Governmental Authority, (b) if not given in the past might have had a Material Adverse Effect, (c) if not continued in the future, might have a Material Adverse Effect or (d) might subject any Acquired Company to a Proceeding by any Governmental Authority.

3.24 Full Disclosure. To the Knowledge of IRP GP and Resource Partners, the representations and warranties made by IRP GP and Resource Partners in this Agreement and the certificates to be delivered pursuant to Section 2.3(a)(ii) and Section 8.3 of this Agreement do not contain any untrue statement of material fact or omit to state a material fact necessary to make any of them in light of the circumstances in which they were made, not misleading.

ARTICLE IIIA

REPRESENTATIONS AND WARRANTIES OF SELLERS

Each Seller individually represents and warrants, severally, but not jointly, to Buyer and only to the extent that the following representations and warranties pertains to such Seller and its ownership interest in Resource Partners, as of the date hereof and as of the Closing Date, other than representations and warranties that expressly speak as of the specific date and time (which need only be true and correct in all respects or true and correct in all material respects as applicable, as of such date and time), as follows:

- 3A.1 Organization. Seller is duly organized, validly existing and in good standing under the Laws of its jurisdiction of organization.
- 3A.2 <u>Authorization</u>; <u>Enforceability</u>. Seller has the requisite organizational power and authority to execute and deliver this Agreement and the other Transaction Documents and to perform its obligations hereunder and thereunder. The execution and delivery of this Agreement and the other Transaction Documents by Seller, and the performance of its obligations hereunder and thereunder, has been duly authorized by all necessary company action on the part of Seller, and, upon such authorization, no other proceedings or actions are necessary to authorize and consummate this Agreement, the other Transaction Documents or the Contemplated Transactions. This Agreement has been, and upon the Closing Date the other Transaction Documents will be, duly executed and delivered by Seller, and, assuming due authorization, execution and delivery by Buyer, constitute the valid and binding agreements of Seller and are or will be enforceable against Seller in accordance with their terms, except as may be limited by applicable bankruptey, insolvency, fraudulent conveyance, reorganization, moratorium and other similar Laws relating to or affecting creditors' rights generally and general equitable principles (whether considered in a proceeding in equity or at law).
- 3A.3 No Conflicts or Approvals. Except as set forth on Schedule 3.6, the execution, delivery and performance of this Agreement and the other Transaction Documents and the consummation of the Contemplated Transactions does not and will not (a) violate, conflict with or result in a breach by Seller of its Organizational Documents, (b) require any filing with or permit, Consent or approval of, or the giving of any notice to any Person, or (c) assuming compliance with applicable requirements under the HSR Act, violate, conflict with, or result in a material breach of any Governmental Order or Law applicable to Seller.
- 3A.4 <u>Interests</u>. The authorized, issued and outstanding Partnership Interests are set forth on Schedule 3A.4, together with the identity of the partners. On the Closing Date, Seller will be the record and beneficial owner and holder of the Partnership Interests shown on <u>Schedule 3A.4</u> as being owned by it, free and clear of all Encumbrances. Seller is not a party to any option, warrant, purchase right, or other contract or commitment (other than this Agreement) that could require Seller to sell, transfer or otherwise dispose of Seller's Partnership Interests.

ARTICLE IV

REPRESENTATIONS AND WARRANTIES OF BUYER

Buyer hereby represents and warrants to Sellers, as of the date hereof and as of the Closing Date, as follows:

- 4.1 <u>Organization</u>. Buyer is a corporation duly organized, validly existing and in good standing under the Laws of its jurisdiction of its formation. Buyer has the requisite organizational power and authority to own, lease and operate its assets and to carry on its business as now being conducted and is duly qualified or licensed to do business and is in good standing in the jurisdictions in which the ownership of its property or the conduct of its business requires such qualification or license, except where the failure to be so qualified or licensed would not reasonably be expected, individually or in the aggregate, to have a Material Adverse Effect.
- 4.2 <u>Authorization</u>; <u>Enforceability</u>. Buyer has the requisite organizational power and authority to execute and deliver this Agreement and the other Transaction Documents and to perform its obligations hereunder and thereunder. The execution and delivery of this Agreement and the other Transaction Documents by Buyer and the performance of its obligations hereunder and thereunder has been duly authorized by all necessary organization on the part of such party and, upon such authorization, no other organization or partnership proceedings or actions are necessary to authorize or consummate this Agreement, the other Transaction Documents or the Contemplated Transactions. This Agreement and the other Transaction Documents have been duly executed and delivered by Buyer and, assuming due authorization, execution and delivery by Sellers, constitute the valid and binding agreements of Buyer and are enforceable against Buyer in accordance with their terms, except as may be limited by applicable bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium and other similar Laws relating to or affecting creditors' rights generally and general equitable principles (whether considered in a proceeding in equity or at law).
- 4.3 No Conflicts or Approvals. The execution, delivery and performance by Buyer of this Agreement and the other Transaction Documents and the consummation by Buyer of the Contemplated Transactions does not (a) violate, conflict with or result in a breach by Buyer of its Organizational Documents, or (b) violate or result in a breach of any Governmental Order or Law applicable to Buyer or any of its properties or assets.
- 4.4 <u>Governmental Authorization</u>. Assuming compliance with the applicable requirement of the HSR Act and compliance with any requirements of the Securities Act, or the Securities and Exchange Act of 1934, as amended, and any other United States state or federal securities laws, the execution, delivery and performance by Buyer of this Agreement and the other Transaction Documents and the consummation of the Contemplated Transactions, require no action by or in respect of, or any Governmental Approval, except with any such action or filing as to which the failure to make or obtain would not be material to Buyer and its Subsidiaries or prevent Buyer from consummating the Contemplated Transactions.

- 4.5 No Brokers' or Other Fees. Except for those Persons listed on Schedule 4.5, whose fees and expenses will be paid by Buyer, no Person has acted directly or indirectly as a broker, finder, investment banker or financial advisor in connection with the Contemplated Transactions, and no Person is entitled to any fee or commission or like payment in connection with the Contemplated Transactions based upon any agreement, arrangement or other understanding made by or on behalf of Buyer.
- 4.6 <u>Permit Blocking</u>. To the Knowledge of Buyer, there is no block or punitive legal limitation on the ability of Buyer or any of its Affiliates to apply for or obtain any Permit under any Environmental Law.

ARTICLE V

EXCLUSIVE REPRESENTATIONS AND WARRANTIES

5.1 WARRANTIES EXCLUSIVE. Except for the express representations and warranties set forth in Article III, Buyer acknowledges that Sellers are making no (and Buyer specifically disclaims that it is relying upon or had relied upon any) representation or warranty, express or implied, at law or in equity, in respect of the Acquired Companies or any of their respective assets, Liabilities or operations, and any such other representations or warranties are hereby expressly disclaimed. Buyer acknowledges that it has inspected and is knowledgeable of the coal reserves that are included with the Real Property. Accordingly, except for the representations and warranties in Article III related to title of the Acquired Companies to the coal reserves otherwise existing at Law or in equity, notwithstanding anything to the contrary in Article III, the coal reserves that are included with the Real Property are accepted by Buyer on an "AS IS" basis, there being no warranties or representations, either express or implied, including but not limited to mineability of the coal, merchantability of the coal, fitness for a particular purpose, quality of the coal or quantity of the coal. With respect to any projection or forecast delivered by or on behalf of Sellers or any Affiliate to Buyer as part of the express representations and warranties set forth in Article III, Buyer acknowledges that (w) there are uncertainties inherent in attempting to make such projections and forecasts, (x) it is familiar with such uncertainties, (y) it is taking full responsibility for making its own evaluation of the adequacy and accuracy of all such projections and forecasts so furnished to it, and (z) it shall have no claim against any Person with respect thereto.

ARTICLE VI

COVENANTS AND AGREEMENTS

6.1 Conduct of Business Prior to the Closing

- (a) Sellers covenant and agree that Sellers will, and will require each Acquired Company to, except (i) as expressly contemplated by this Agreement, (ii) as disclosed in Schedule 6.1(a) or (iii) with the prior written consent of an authorized officer of Buyer, from the date hereof to Closing, (A) conduct the business of each Acquired Company in the Ordinary Course of Business, (B) use commercially reasonable efforts to preserve intact the current business organization of each Acquired Company, keep available the services of the current officers, employees, and agents of each Acquired Company, and maintain the relations and good will with suppliers, customers, landlords, creditors, employees, agents, and others having business relationships with each such Acquired Company, and (C) maintain and renew, if necessary, all insurance policies listed on Schedule 3.17 and all insurance policies not held in the name of any Acquired Company but which cover the assets and operations of the business of any Acquired Company and any Company Employees.
- (b) Without limiting the foregoing, Sellers covenant and agree that they will not, and will require each Acquired Company not to, except (i) as expressly contemplated by this Agreement, (ii) as disclosed in <u>Schedule 6.1(b)</u> or (iii) with the prior written consent of an authorized officer of Buyer, from the date hereof to Closing,
- (A) change any Acquired Company's authorized or issued membership interests; grant any option or right to purchase membership interest of any Acquired Company; issue any security convertible to a membership interest; grant any registration rights in such membership interests; or sell, pledge, purchase, redeem, retire, or otherwise acquire any membership interests;
 - (B) amend the Organizational Documents of any Acquired Company;
- (C) pay or increase any bonuses, salaries, severance or other compensation to any stockholder, director, officer, or employee; adopt or amend any Benefit Plan or Employee Arrangement, or enter into any employment, severance, or similar contract with any director, officer, or employee (except in the Ordinary Course of Business);
 - (D) cause a material change in the accounting methods or practices, collection policies, pricing policies or payment policies used by any Acquired Company;
- (E) make any Tax election or settle or compromise any federal, state, local or foreign income Tax Liability, or waive or extend the statute of limitations in respect of any such Taxes;
- (F) (i) make any loan, or redeem or purchase any equity interests of an Acquired Company, transfer any asset, or enter into or agree to enter into any transaction to, with or for the benefit of any Acquired Company or any of their Affiliates (or agree, whether in writing or otherwise, to do any of the foregoing) or (ii) make any loans, advances or capital contributions to, or invest in, any other Person, except for employee advances for expenses in the Ordinary Course of Business;

(G) sell, transfer, license, sublicense, lease, rent or otherwise dispose of or encumber any of the Leased Real Property, Owned Real Property or Material
Personal Property pertaining to the businesses of the Acquired Companies, other than the (i) sale of coal, other inventory and obsolete equipment, (ii) lease, sublease or entering
into other contractual relationships by and among an Acquired Company and one or more other Acquired Companies or third parties relating to rights of way, surface rights or
similar rights of use, (iii) purchase of leased equipment under existing lease options that are disclosed on Schedule 3.18(c), or (iv) lease of additional equipment in the Ordinary
Course of Business under existing master leases that are disclosed on Schedule 3.18(c);

- (H) cancel any debts owed to or waive any material claims or rights of the Acquired Companies;
- (I) fail to make any capital expenditure or commitment in the Ordinary Course of Business or as planned as of the date hereof, or make or commit to make any capital expenditure or commitment outside the Ordinary Course of Business not planned as of the date of this Agreement;
 - (J) recognize any labor union or other employee representative, or enter into any collective bargaining agreement;
- (K) incur, assume or guarantee any Indebtedness other than such Indebtedness that is repaid in full prior to Closing, or enter into any swap or other off-balance sheet transaction for its own account, or enter into any economic arrangement having the economic effect of any of the foregoing;
- (L) amend or terminate any Material Contract, Equipment Lease or Real Property Lease or enter into any agreement that would be considered a Material Contract, Equipment Lease or Real Property if entered into prior to the data hereof, or waive or assign any material right under any Material Contract, Equipment Lease or Real Property Lease, including grants of Intellectual Property rights by assignment, license or otherwise;
- (M) satisfy any liabilities or compromise or settle (including by entering into any consent decree) any lawsuit or claim (including any lawsuit or claim involving a Governmental Authority) unless such satisfaction or compromise includes the payment in full by the applicable Acquired Company of all financial obligations arising therefrom;
 - (N) fail to pay accounts payable and other obligations, when they become due and payable;
- (O) merge or consolidate with, or agree to merge or consolidate with, or purchase the assets of, or otherwise acquire, any business, business organization or division thereof, or any Person (other than the purchase of assets from suppliers and vendors in the Ordinary Course of Business);
 - (P) fail to maintain in full force and effect insurance comparable in amount and scope of coverage to insurance now carried by it; or

(Q) agree, whether in writing or by oral agreement, to do any of the foregoing.

- (c) Until such time, if any, as this Agreement is terminated pursuant to Article IX, Sellers will not, and will cause their respective representatives not to, directly or indirectly, solicit, initiate, or encourage any inquiries or proposals from, discuss or negotiate with, provide any non-public information to, or consider the merits of any unsolicited inquiries or proposals from, any Person (other than Buyer) relating to any transaction involving the sale of any Acquired Company, or any of the membership interests of any Acquired Company, substantially all of the assets of any Acquired Company, or any merger, consolidation, business combination, or similar transaction involving any Acquired Company. Sellers, and each of their respective representatives, shall immediately notify Buyer regarding any contact between Sellers, and/or their respective representatives and any other Person regarding any offer, proposal or related inquiry unless such notification could violate any existing confidentiality agreement to which an Acquired Company is a party.
- 6.2 <u>Redemption</u>. Prior to the Closing, Resource Partners shall redeem all the limited partnership units issuable or outstanding under Resource Partners' Restricted Unit Plan for a price equal to such holder's (the "<u>Redeemed Partners</u>") pro rata share of the Purchase Price to which such Redeemed Partner would have been entitled pursuant to this Agreement had such Redeemed Partner held such limited partnership units immediately prior to the Closing (the "<u>Redemption</u>") and satisfy or enter into an agreement pursuant to which Agent on behalf of Sellers shall be solely responsible for satisfying any actual or contingent obligations under Resource Partners' Long-Term Incentive Plan (the "<u>Contingency Obligations</u>"). Resource Partners shall enter into a redemption agreement with each Redeemed Partner and shall enter into an agreement with each Person entitled to any Contingency Obligation, each in a form reasonably satisfactory to Buyer which agreements shall include, among other things, a release by such Redeemed Partner or such Person entitled to any Contingency Obligations of Resource Partners, Buyer and their Affiliates in a form reasonably satisfactory to Buyer.
- 6.3 Access and Investigation. Between the date of this Agreement and the Closing Date, Sellers will, and will cause each Acquired Company and its Related Persons to, (a) afford Buyer and its Related Persons and prospective lenders and their Related Persons (collectively, "Buyer's Advisors") reasonable access to each Acquired Company's personnel, properties (including subsurface testing), contracts, books and records, and other documents and data, (b) furnish Buyer and Buyer's Advisors with copies of all such contracts, books and records, and other existing documents and data as Buyer may reasonably request, and (c) furnish Buyer and Buyer's Advisors with such additional financial, operating, and other data and information as Buyer may reasonably request.

6.4 Required Approvals; Consents.

(a) As promptly as practicable after the date of this Agreement, Sellers will, and will cause each Acquired Company to, make all filings required by Law to be made by them in order to consummate the Contemplated Transactions except for filings to be made pursuant to Section 6.10 and Section 6.14 which shall be governed by those sections. Between the date of this Agreement and the Closing Date, Sellers will, and will cause each Acquired

Company to, (i) cooperate with Buyer with respect to all filings that Buyer makes pursuant to Section 6.4(b), and (ii) cooperate with Buyer in obtaining and use its commercially reasonable efforts to secure all Consents and Governmental Approvals required as a result of the Contemplated Transactions identified in Schedule 3.6 and Schedule 3.7 provided that Seller's commercially reasonable efforts shall not include making any cash payments to any party from which a consent is sought other than the Consent Fees.

- (b) As promptly as practicable after the date of this Agreement, in consultation with Sellers, Buyer will, and will cause each of its Related Persons to, make all filings required by Law to be made by them in order to consummate the Contemplated Transactions except for filings to be made pursuant to Section 6.10 and Section 6.14 which shall be governed by those sections. Between the date of this Agreement and the Closing Date, Buyer will, and will cause each Related Person to, (i) cooperate with Sellers with respect to all filings that Sellers make pursuant to Section 6.4(a), and (ii) cooperate with Sellers in obtaining all Consents and Governmental Approvals required as a result of the Contemplated Transactions identified in Schedule 3.6 and Schedule 3.7.
- (c) To the extent permitted by Law, Buyer and Sellers shall each cause their respective counsel to supply to the other party copies of all correspondence, filings or written communications by such party or its Affiliates with any Governmental Authority or staff members thereof, with respect to the Contemplated Transactions.
- 6.5 <u>Further Actions</u>. Between the date of this Agreement and the Closing Date, Sellers and Buyer will use their respective commercially reasonable efforts to take, or cause to be taken, all appropriate action, and do, or cause to be done, all things necessary, proper or advisable under applicable Laws or otherwise to consummate and make effective the Contemplated Transactions as promptly as practicable, and cooperate with each other to do so, including without limitation, obtaining as promptly as practicable all Consents and Government Approvals necessary in order to consummate the Contemplated Transactions, all in accordance with the provisions of this Agreement.

6.6 Financing.

- (a) Buyer has received, accepted and agreed to a commitment letter, dated March 6, 2011 (the "Commitment Letter") from the lenders party thereto (collectively, the "Lenders") relating to the commitment of the Lenders to provide the debt financing required to consummate the Contemplated Transactions. The debt financing required to consummate the Contemplated Transactions, whether obtained pursuant to the arrangements contemplated by the Commitment Letter or through substitute permanent financing arrangements which may involve public or private offerings of debt or equity securities, is collectively referred to in this Agreement as the "Financing". A complete and correct copy of the executed Commitment Letter has been provided to Agent.
- (b) Buyer shall use its reasonable best efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things reasonably necessary, proper or advisable to consummate and obtain the Financing on the terms and conditions described in the Commitment Letter, including (i) taking actions to enforce its rights under the Commitment Letter and (ii) using its reasonable best efforts to (A) negotiate definitive agreements with respect thereto on the terms and conditions contained therein or on other terms no more adverse to Buyer in any material respect, (B) satisfy on a timely basis all conditions applicable to it in such definitive agreements and (C) consummate the Financing at or prior to the Closing but in no event later than June 1, 2011.

- (c) Buyer may terminate and replace the Commitment Letter with a New Commitment Letter or may amend the Commitment Letter to add lenders, lead arrangers, bookrunners, syndication agents or similar entities who had not executed the Commitment Letter as of the date hereof, so long as (i) the addition of such lenders, lead arrangers, bookrunners, syndication agents or similar entities would not reasonably be expected to prevent or materially delay the consummation of the Contemplated Transactions and (ii) the Arranger (as defined in the Commitment Letter) shall remain obligated and committed to fund its financing commitments under the Commitment Letter (as in effect prior to any such amendment) on the terms and conditions of, and subject to the assignment provisions set forth in, the Commitment Letter (as in effect prior to any such amendment, replacement, supplement or modification of the Commitment Letter in accordance with this Section 6.6(c), the term "Commitment Letter" shall mean the Commitment Letter as so amended, supplemented or modified or any New Commitment Letter in replacement of the Commitment Letter, as the case may be, and, in the event that Buyer obtains Alternative Financing in accordance with Section 6.6(c), the term "Commitment Letter" shall mean the commitment letter (as amended, replaced, supplemented or modified in accordance with this Section 6.6(c)) related to the Alternative Financing. The term "New Commitment Letter" shall mean the instrument replacing the existing Commitment Letter that (A) is on terms not materially less beneficial to Buyer than those set forth in the Commitment Letter; (B) does not involve any material conditions precedent to funding the Financing that are not contained in the Commitment Letter; and (C) would not reasonably be expected to prevent or materially delay the consummation of the Contemplated Transactions.
- (d) In the event any portion of the Financing becomes unavailable on the terms and conditions contemplated in the Commitment Letter (as such terms may be modified or adjusted in accordance with the terms of, and within the limits of, the "flex" provisions contained in the Commitment Letter or accompanying fee letter) for any reason, Buyer shall use its reasonable best efforts to obtain, as promptly as practicable but no later than the Drop Dead Date, financing from alternative sources in an amount sufficient to consummate the Contemplated Transactions and that would not (i) involve any material conditions to funding the Financing that are not contained in the Commitment Letter; (ii) reasonably be expected to prevent or materially delay the consummation of the Contemplated Transactions; and (iii) include terms that would be materially adverse when compared to the terms of the Commitment Letter (an alternative financing meeting the foregoing requirements shall be an "Alternative Financing"). In the event that Alternative Financing shall be obtained pursuant to this Section 6.6(d), Buyer shall comply with the covenants in Section 6.6(e) with respect to such Alternative Financing.
- (e) Buyer shall give Sellers prompt written notice of any material breach by any party of the Commitment Letter (or commitments for any Alternative Financing obtained in accordance with this Section 6.6(e)) of which Buyer becomes aware, any termination of the Commitment Letter (or commitments for any Alternative Financing obtained in accordance with this Section 6.6(e)) or any condition precedent of the Financing (or any Alternative Financing obtained in accordance with this Section 6.6(e)) which Buyer expects will not be satisfied at the Closing, as applicable.

- (f) At Buyer's request, Sellers shall cause the Acquired Companies, and its or their representatives, to use their reasonable best efforts to (x) cooperate with Buyer to complete the Financing to the extent that information relating to, or the participation by members of management of, the Acquired Companies is reasonably necessary in connection therewith; and (y) provide all information and assistance that is customarily provided in financings comparable to the proposed Financing. Without limiting the generality of the foregoing, Sellers shall cause the Acquired Companies to:
- (i) furnish to Buyer as promptly as practicable, upon Buyer's request from time to time, (1) all Financial Statements, (2) all other financial statements required by the Commitment Letter (or commitments for any Alternative Financing obtained in accordance with Section 6.6(e)) with respect to the Acquired Companies within the time periods specified therein, including information reasonably requested by Buyer to assist in the preparation of pro forma financial statements and the projected consolidated financial statements of Buyer and (3) all financial statements, business and other financial data, audit reports and other information regarding the Acquired Companies of the type that would be required by Regulation S-X and Regulation S-K promulgated under the Securities Act for a registered public offering of debt or equity securities of Buyer (to the extent such information in clauses (2) or (3) currently exists and, in the event such information does not exist, Sellers will assist in obtaining such information at Buyer's expense);
- (ii) use its reasonable best efforts to furnish to Buyer, as promptly as reasonably practicable, other information with respect to the Acquired Companies reasonably requested by Buyer in writing (1) so that Buyer may satisfy the applicable SEC requirements for public capital markets offerings and (2) of the type and form customarily included in offering documents used in private placements under Rule 144A under the Securities Act;
- (iii) use its reasonable best efforts to furnish to Buyer, as promptly as reasonably practicable, information relating to the Acquired Companies reasonably requested by Buyer in writing that is customarily used in the preparation of customary offering or information documents or rating agency, lender presentations or road shows relating to the Financing;
- (iv) upon reasonable advance notice and at Buyer's expense, participate in investor presentations, road shows, due diligence sessions and rating agency presentations in connection with the Financing, including participation by senior management of the Acquired Companies with appropriate seniority and expertise;
 - (v) cooperate reasonably with Buyer's Financing sources' due diligence, to the extent customary and reasonable and at Buyer's expense;

- (vi) assist at Buyer's expense with the preparation, including participating in drafting sessions, of any confidential information memorandum, offering memorandum, prospectus, registration statement or other offering document (including any amendments or supplements thereto and an additional version of any confidential information memorandum for use by "public-side" lenders) or rating agency presentation used by Buyer in connection with the Financing;
- (vii) request required consents of accountants for use of their reports in any materials relating to the Financing where the inclusion of such reports is necessary and request the delivery of customary comfort letters and customary representation letters;
- (viii) use its reasonable best efforts to facilitate pledging collateral of the Acquired Companies in connection with any debt portion of the Financing, including the execution and delivery of customary pledge and security agreements and other customary certificates and documents prepared by Buyer, as may be reasonably requested by Buyer (provided that any obligations contained in such documents shall be effective no earlier than as of the Closing);
 - (ix) provide authorization letters to the Financing sources authorizing the distribution of the Acquired Companies' information to prospective lenders;
- (x) use commercially reasonable efforts to assist Buyer and the Financing sources in benefiting from the Acquired Companies' existing lending and investment banking relationships;
 - (xi) take such corporate actions (subject to the occurrence of the Closing) as are reasonably necessary to complete the Financing; and
 - (xii) use reasonable best efforts to facilitate the execution and delivery (at the Closing) of definitive documents related to the Financing.

Sellers hereby consent to the use of the Acquired Companies' logos in connection with the Financing; provided that such logos are used solely in a manner that is not intended to or reasonably likely to harm or disparage the Acquired Companies or the Sellers or the reputation or goodwill of the Acquired Companies or the Sellers.

- 6.7 <u>Further Assurances</u>. Following the Closing, Sellers and Buyer will, and each will cause each of its respective Related Persons to, from time to time, execute and deliver such additional instruments, documents, conveyances or assurances and take such other actions as shall be necessary, or otherwise reasonably required by Sellers, on the one hand, and Buyer, on the other hand, as applicable, to confirm and assure the rights and obligations provided for in this Agreement and render effective the consummation of the Contemplated Transactions.
 - 6.8 Purchase Price Deductions; Special Indemnity.
 - (a) Sellers hereby agree that the Cash Portion of the Purchase Price shall be reduced by the amounts set forth or <u>Schedule 6.8(a)</u> (the "<u>Purchase Price Deductions</u>").

- (b) Sellers hereby covenant and agree that, prior to the Closing, Sellers shall pay to the extent known and quantifiable (i) any and all performance bonuses earned by any Company Employee through the Closing Date and not reflected in Current Liabilities ("Bonuses"); (ii) any and all "change of control" or similar payments to which any Company Employee is entitled as a result of the Contemplated Transactions (but only to the extent such right is not waived in writing by such Company Employee) ("Change of Control Payments"); and (iii) one-half of any cost or expense incurred by Sellers, Buyer or the Acquired Companies to induce a third party to that certain lease agreements listed under Item 3 of Schedule 6.8(c) to consent to the assignments of such lease and other similar fees related to such lease and arising from the Contemplated Transactions (the "Consent Fees").
- (c) Sellers hereby covenant and agree that, from and after the Closing, Sellers shall indemnify each Buyer Indemnitee, to the extent provided in Section 10.1(a), for any Losses of such Buyer Indemnitee with respect to, whether disclosed to Buyer or not, any Bonus, Change of Control Payment, Consent Fees or other matter set forth on Schedule 6.8(c) (the "Special Indemnity Matters") arising on or after the Closing Date, not included in the Purchase Price Deductions pursuant to Section 6.8(a) or not paid by Sellers pursuant to Section 6.8(b).

6.9 Guarantees; Seller Bonds.

- (a) <u>Sellers' Guarantees</u>. Prior to and after the Closing, Buyer shall cooperate with Sellers and their Affiliates and shall use its commercially reasonable efforts, which shall include preparation and submission of documents, provision of information and response to requests, to cause Buyer and/or one or more Affiliates of Buyer to be substituted in all respects for Sellers or any Affiliate of Sellers, in respect of all Liabilities under the Seller Guarantees, and to cause Sellers and all Affiliates of Sellers to be fully released and discharged with respect thereto, in each case, as of the Closing Date; provided that, Buyer's commercially reasonable efforts shall not include making any cash payments to any party from which a release is sought or modifying or amending any agreement with any party from which a release is sought in a manner that is adverse to the Acquired Companies or Buyer. Without limiting the generality of the foregoing, with respect to such Seller Guarantees that are the result of Sellers being an indemnitor under any of Seller Bonds, Buyer may post a replacement surety bond for the bond of the Company in the manner set forth in Section 6.9(b) in lieu of being substituted for Sellers. With respect to any Seller Guarantees which are not fully released and discharged prior to the Closing Date, Buyer shall continue to use commercially reasonable efforts to replace such Seller Guarantees and/or to cause Sellers and all Affiliates of Sellers to be fully released and discharged therefrom effective as of the Closing Date.
- (b) <u>Bonds</u>. With respect to the Seller Bonds, Buyer shall post, within five (5) Business Days following the Closing Date, a replacement surety bond for the benefit of the beneficiaries thereof in form and substance and issued by a surety company that is satisfactory to such beneficiaries. Each such replacement surety bond so provided by the Buyer shall provide on its face that it "supersedes and replaces" the applicable Seller Bond as of the Closing Date. At or prior to the Closing, Buyer shall deliver, or cause to be delivered, to Sellers, such other documents as may reasonably be requested by Sellers in order to permit Sellers to effect the full release and discharge of Sellers and all Affiliates of Sellers as of the Closing Date. Buyer shall reimburse Sellers on a monthly basis for all reasonable costs and expenses incurred by Sellers or the Affiliates of Sellers with respect to all such bonds that remain outstanding following the Closing Date to the extent such costs and expenses accrue on or after the Closing Date.

- (c) <u>Indemnification After Closing</u>. Buyer shall indemnify and hold Sellers and all Affiliates of Sellers harmless from and against, and pay and reimburse Sellers and all Affiliates of Sellers for, any and all Losses that they may incur arising or relating to the Seller Guarantees or Seller Bonds for events occurring on or after the Closing Date. Any payment required to be made by Buyer under this Section shall be made within ten (10) Business Days after Buyer's receipt of written notice from Sellers or any Affiliate of Sellers describing in reasonable detail the amount owing hereunder.
- (d) <u>Refund of Premiums</u>. All funds in respect of premiums that are refunded in respect of Seller Guarantees or Seller Bonds on account of the replacement thereof by Buyer pursuant to this Section shall be for the account of Sellers and their Affiliates. Buyer shall pay over or cause the Acquired Companies to pay over to Sellers any such refunds received by any Acquired Company after the Closing promptly upon receipt of such refunds.
- (e) Required Efforts. Notwithstanding the foregoing, in no event shall any party hereto be required to or otherwise file any lawsuit in order to cause the replacement and/or release and discharge of any Seller Guaranty or Seller Bond. After the Closing, Buyer will not, and will not permit any Acquired Company to, renew, extend, amend or supplement any loan, contract, lease or other obligation underlying any guarantee with respect to which Sellers and their Affiliates are not fully released and discharged as of the Closing Date in any manner that would extend or increase the Liability of Sellers or any Affiliate of Sellers without providing Sellers with evidence satisfactory to Sellers that Sellers and their Affiliates have been fully released and discharged.
- (f) <u>Approval of Replacement Bonds</u>. So long as Buyer provides adequate credit support for the replacement of each Seller Bond pursuant to this Section 6.9 on or before the Closing Date, any delay in the release of a Seller Bond or a bond being replaced at the discretion of Buyer in lieu of replacing Sellers as an indemnitor for a Seller Bond shall not constitute a breach of the covenant of Buyer to post such replacement surety bond by the Closing Date.

6.10 <u>Permits</u>.

(a) Pending Transfers: Documents. Within five (5) Business Days following the Closing Date, Buyer shall submit (i) all filings necessary to cause the applicable Governmental Authorities to transfer to Buyer the Permits (the "Transfer Applications"); and (ii) any and all replacement bonds or other financial security, if applicable, required to complete the transfer of the Permits and replace the Seller Bonds. The parties acknowledge that some Governmental Authorities may not require a permit transfer but may require updated ownership and control information perhaps with a bond replacement and/or the issuance of a new license or mine number and that any reference herein to a permit transfer shall also refer to bond replacement and/or ownership and control information or issuance of a new license or mine number. Prior to filing, Buyer shall deliver to Agent copies of the Transfer Applications. Sellers will cooperate with Buyer and take all actions necessary to complete all necessary forms to effect the Transfer Applications and to effect Buyer's ability to operate under Sellers' permits during the Permits Transfer Period described below.

- (b) <u>Post-Filing Obligations</u>. Buyer shall diligently pursue and use its commercially reasonable efforts to expeditiously secure the transfer of all the Permits within the Permits Transfer Period, which shall include, but is not limited to, cooperating with the relevant Governmental Authorities, participating in the public comment process, and working with interested parties and neighboring landowners. In the event Buyer fails to secure transfer of (i) all Permits (other than Permits for which there is a pending action) within three (3) months from the Closing Date and (ii) all Permits for which there is a pending action within twelve (12) months from the Closing Date, then Buyer shall pay to Agent, for the benefit of Sellers the amount of any and all reasonable costs, expenses or damages incurred by Sellers or their Affiliates resulting from such failure, provided however, that if such failure is caused by Sellers' actions (including any failure by Sellers to take any necessary action), then Sellers shall pay to Buyer the amount of any and all reasonable costs, expenses or damages incurred by Buyers or their Affiliates (including the Acquired Companies following the Closing) resulting from such failure.
- (c) <u>Violations</u>. During the period of time from the Closing Date until the applications for the transfer of control of the Acquired Companies has been accepted by the relevant Governmental Authorities ("<u>Permits Transfer Period</u>"), if Sellers receive a notice of violation under any of the Permits, it will give Buyer and the applicable Acquired Company (as owned by Buyer) prompt notice thereof. During the Permits Transfer Period, Buyer and the applicable Acquired Company (as owned by Buyer) shall use their commercially reasonable efforts to timely cure such violation. If Sellers reasonably determine that Buyer or the applicable Acquired Company (as owned by Buyer) will not cause such violation to be cured in a timely fashion during the Permits Transfer Period, they shall have the right to cure, or cause to be cured, such violation and be reimbursed by Buyer or the applicable Acquired Company (as owned by Buyer) for all reasonable costs incurred in curing such violation.
- 6.11 <u>Confidentiality</u>. The provisions of that certain letter agreement by and between Resource Partners and Buyer dated December 14, 2010 (the <u>Confidentiality Agreement</u>"), shall continue to apply to the parties as though all such parties were original signatories thereto. If this Agreement is terminated for any reason whatsoever, the parties shall continue to be bound by the terms of the Confidentiality Agreement and the Buyer shall return to the Company all tangible embodiments (and all copies) of "Evaluation Material" (as defined in the Confidentiality Agreement) that are in its possession.

6.12 Tax Matters.

- (a) <u>Proration of Straddle Period</u>. In the case of Taxes that are payable with respect to any Straddle Period, the portion of any such Tax that is attributable to the portion of the period ending on the Closing Date shall be:
- (i) in the case of Taxes that are either (A) based upon or related to income or receipts, or (B) imposed in connection with any sale or other transfer or assignment of property (real or personal, tangible or intangible), deemed equal to the amount that would be payable if the Tax years of each of the Acquired Companies (and each partnership in which any of the Acquired Companies is a partner) ended with (and included) the Closing Date; and

- (ii) in the case of Taxes that are imposed on a periodic basis with respect to the assets of any Acquired Company, deemed to be the amount of such Taxes for the entire period (or, in the case of such Taxes determined on an arrears basis, the amount of such Taxes for the immediately preceding period), multiplied by a fraction the numerator of which is the number of calendar days in the portion of the period ending on the Closing Date and the denominator of which is the number of calendar days in the entire period.
- (b) <u>Transfer Taxes</u>. All excise, sales, use, transfer (including real property transfer or gains), stamp and other similar taxes, together with any interest, additions or penalties with respect thereto and any interest in respect of such additions or penalties, resulting directly from the Contemplated Transactions (the "<u>Transfer Taxes</u>"), if any, shall be borne equally by Buyer, on the one hand, and Sellers, on the other hand. If Sellers, on the one hand, or Buyer, on the other hand, pay any Transfer Taxes then the other party shall reimburse the paying party for one half of such Transfer Taxes actually paid. Any Tax Returns that must be filed in connection with Transfer Taxes shall be prepared and filed when due by the party primarily or customarily responsible under the applicable local law for filing such Tax Returns, and such party shall use reasonable commercial efforts to provide such Tax Returns to the other party at least ten (10) days prior to the due date for such Tax Returns.

(c) Cooperation on Tax Matters.

- (i) The Buyer and each of the Acquired Companies, on the one hand, and the Sellers, on the other hand, shall cooperate fully, as and to the extent reasonably requested by the other party, in connection with the filing of Tax Returns and any audit, litigation or other proceeding with respect to Taxes. Such cooperation shall include the retention and, upon the other party's request, the provision of records and information that are reasonably relevant to any such audit, litigation or other proceeding and making employees available on a mutually convenient basis to provide additional information and explanation of any material provided hereunder. The Buyer and each of the Acquired Companies, on the one hand, and the Sellers, on the other hand, agree to retain all books and records with respect to Tax matters pertinent to the Acquired Companies relating to any taxable period beginning before the Closing Date until the expiration of the statute of limitations (and, to the extent notified by the other party, any extensions thereof) of the respective taxable periods, and to abide by all record retention agreements entered into with any taxing authority.
- (ii) The Buyer and the Sellers further agree, upon request, to use their commercially reasonable efforts to obtain any certificate or other document from any Governmental Authority or any other Person as may be necessary to mitigate, reduce or eliminate any Tax that could be imposed (including with respect to the Contemplated Transactions).

6.13 Employee Plans. Buyer shall, or shall cause one of its Affiliates to, provide each individual who is employed by any of the Acquired Companies on the Closing Date and who continues employment with Buyer or any of its Affiliates (including the Acquired Companies after Closing) after the Closing Date (the "Continuing Employees") with a position providing base pay that is at least equal to the base pay provided to such Continuing Employees by the applicable Acquired Company immediately prior to the Closing Date. Nothing in this Section 6.13, however, will limit the right of Buyer to terminate or suspend employment of any Continuing Employee after the Closing, subject only to the provisions of Section 6.15. For a period of at least twelve (12) months beginning on the Closing Date, the Buyer shall provide, or cause to be provided, to Continuing Employees, employee benefits and programs that, individually or in the aggregate, are substantially similar either to the Employee Benefit Plans maintained or contributed to by the Acquired Companies (as the case may be) for employees of the Acquired Companies as of the date of this Agreement or to employee benefit plans and programs provided by the Buyer to similar situated employees of the Buyer on the date as of which the substantially similar comparison is made. Continuing Employees shall receive full credit for their services with the Acquired Companies for purposes of eligibility to participate and vesting in benefits under any employee benefit plans and programs of the Buyer and its controlled group of corporations in which they are eligible to participate; provided, that no credit shall be given for purposes of benefit accrual or entitlement to severance benefits or if the service credit would result in a duplication of benefits. For the entire calendar year of 2011, Continuing Employees shall be permitted to take earned but not taken vacation time-off as provided in Sellers' vacation time-off policy. Buyer shall provide the Continuing Employees and their eligibl

6.14 Antitrust Notification.

(a) As promptly as reasonably practicable following the execution of this Agreement, Buyer and Agent, on behalf of Sellers shall make all pre-transaction notification filings required under the HSR Act (which shall be made no later than seven (7) Business Days after the date of this Agreement) and any other filing under any statute, law, ordinance, rule or regulation designed to prohibit, restrict or regulate actions for the purpose or effect of preventing monopolization, restraints of trade or abusing a dominant position (collectively, "Antitrust Laws") of any country, state, province or jurisdiction of competent jurisdiction requiring a notification and observation of a waiting period or requiring prior approval before consummating the Contemplated Transactions and the other Transaction Documents ("Pre-Transaction Notification Rules"). Each of Buyer and Sellers shall: (i) cooperate fully with each other and shall furnish to the other such necessary information and reasonable assistance as the other may reasonably request in connection with its determination as to whether and which Pre-Transaction Notification Rules are necessary or advisable and the preparation of any filings under any Pre-Transaction Notification Rules; (ii) keep the other party reasonably informed of any communication received by such party from, or given by such party to any Governmental Authority charged with enforcing, applying, administering, or investigating any Antitrust Law, including the United States Federal Trade Commission, the United States Department of Justice, any attorney general of any state of the United States or any other competition authority of any jurisdiction ("Antitrust Authority"), and of any communication received or given in connection with any proceeding by a private party related to the Antitrust Laws, in each case, regarding this Agreement and the other Transaction Documents and in a manner that protects attorney-client or attorney work product privilege; and (iii) permit the other party to review and incorporate the other party's reasonable comments in any communication given by it to any Governmental Authority or in connection with any proceeding by a private party related to Antitrust Laws with any other Person, in each case, regarding the Contemplated Transactions and in a manner that protects attorney-client or attorney work product privilege. Unless otherwise agreed and without limiting the obligations stated in this Section 6.14(a), Buyer and Sellers shall each use their respective commercially reasonable efforts to ensure the prompt expiration of any applicable waiting period under any Pre-Transaction Notification Rules and approval by any relevant Antitrust Authority. Further, without limiting the obligations stated in this Section 6.14(a), Buyer and Sellers shall each use their respective commercially reasonable efforts to respond promptly to and comply with any request for information any Governmental Authority regarding the Contemplated Transactions or filings under any Pre-Transaction Notification Rules. Neither party hereto shall initiate, or participate in any meeting or discussion with any Governmental Authority with respect to any filings, applications, investigation, or other inquiry regarding the Contemplated Transactions or filings under any Pre-Transaction Notification Rules without giving the other party reasonable prior notice of the meeting or discussion and, to the extent permitted by the relevant Governmental Authority, the opportunity to attend and participate (which, at the request of any party, shall be limited to outside antitrust counsel only).

- (b) The parties shall take reasonable efforts to share information protected from disclosure under the attorney-client privilege, work product doctrine, joint defense privilege or any other privilege pursuant to this section so as to preserve any applicable privilege.
- (c) No party shall take any action with the intention to or that could reasonably be expected to hinder or delay the obtaining of clearance or any necessary approval of any antitrust authority under any Pre-Transaction Notification Rule or Antitrust Law or the expiration of the required waiting period under the Pre-Transaction Notification Rules or any other Antitrust Laws.
- 6.15 WARN Act. Notwithstanding the provisions of Section 6.13, Sellers agree that prior to the Closing, and Buyer agrees that for ninety (90) days after the Closing, Sellers or Buyer shall not, and shall not allow any of the Acquired Companies to, take any act or allow any failure to act that would be a violation of any WARN Act. Sellers covenant and agree that, from the date hereof through the Closing, Sellers shall, and shall cause each of the Acquired Companies, not to terminate more than 30 employees of the Acquired Companies, in the aggregate, without notice to and the consent of Buyer, which consent shall not be unreasonably withheld.

6.16 Schedules. At all times prior to the Closing, Sellers shall provide Buyer with written notification, which notification may be in the form of a written update to the schedules (each, an "Update"), of any change, fact, event, occurrence or other information of any kind whatsoever, whether arising before or after the date of this Agreement, which, except as provided in this Section 6.16, causes or constitutes a breach of any representation or warranty made by Sellers or that would make the timely satisfaction of any of the conditions set forth in Article VII impossible. Any Update shall specify each portion of any representation and warranty or condition affected by such Update. Sellers shall provide to Buyer such information as Buyer may reasonably request in order to review any matter disclosed in any Update. In connection with determining the accuracy of any of the representations and warranties made by Sellers in this Agreement for the purposes of Article X, (a) the delivery of an Update with respect to any event, fact or matter existing but not disclosed in an applicable schedule as of the date of this Agreement (a "Pre-Signing Event Update") shall not cure or otherwise affect or be deemed a waiver of any breach of a representation or warranty made as of the date hereof and such Pre-Signing Event Update shall not qualify any such representation and warranty as of the date of this Agreement, and (b) the delivery of an Update with respect to any event, fact or matter first arising after the date of this Agreement (a "Post-Signing Event Update") shall be deemed to supplement or amend the schedules solely for the purpose of qualifying the portions of the representations and warranties specified in the Post-Signing Event Update as the same are made at and as of the Closing as if such Post-Signing Event Update had been incorporated into the schedules theretofore in effect; provided, however, after delivery of a Post-Signing Event Update to Buyer, Buyer shall have the right to terminate this Agreement upon written notice to Sellers unless Sellers agree in the Post-Signing Event Update that such Post-Signing Event Update shall not be deemed to effect or to qualify any representation or warranty for purposes of making the representations and warranties herein at and as of the Closing provided that if Seller does not agree and Buyer does not exercise its right to terminate and the Closing occurs, Buyer Indemnitee's right to indemnification shall be deemed to have been waived. Except as provided in the immediately preceding sentence, neither the delivery of an Update nor any Closing thereafter shall be deemed a waiver of any Buyer Indemnitee's right to indemnification pursuant to Section 10.1(a) for any breach or nonfulfillment by Sellers of any covenant or agreement contained in this Agreement with respect to the subject matter of any Update or for any breach of any representation or warranty made at and as of the date of this Agreement. In connection with determining the accuracy of any of the representations and warranties made by Sellers in this Agreement for the purposes of determining whether any of the conditions set forth in Article VII has been satisfied (including the condition in Section 7.1), all Updates, including Post-Signing Event Updates, shall be disregarded. If an Update is delivered within three (3) Business Days of the Closing, Buyer shall be entitled, at its option, to delay the Closing by three (3) Business Days. Each Update shall specify whether it is a Pre-Signing Event Update or a Post-Signing Event Update.

6.17 Release.

(a) As of the Closing, Sellers, in their individual capacity and on behalf of their respective Affiliates, and to the extent Sellers have the authority, any Related Person of any of the foregoing and all Persons claiming by, through, for or under Sellers on behalf of Sellers, hereby irrevocably and unconditionally release, settle, cancel, discharge and acknowledge to be fully and finally satisfied any and all claims, demands, rights, actions, causes of action, debts, accounts, covenants, contracts, agreements, promises, damages, costs, reimbursements, compensation, liabilities and expenses, including attorney's fees, of any and every kind, nature or description whatsoever, known or unknown, at law or in equity (including with respect to the allocation among Sellers of the proportionate amount of consideration received pursuant to the Contemplated Transactions) other than those that Sellers may have against Buyer arising under this Agreement or any other Transaction Document (subject to the terms of this Agreement) (collectively, "Sellers Claims and Losses"), which Sellers or any of their respective Related Parties, or any other Person claiming by, through, for or under Sellers on behalf of Sellers, may have had or may now have or assert against (i) Buyer and its Affiliates, (ii) any of the Acquired Companies (collectively with Buyer and its Affiliates, "Buyer Released Parties"), which are on account of any matter whatsoever attributed to the period, or arising during the period, from the beginning of time through and including and ending at the Closing (all Sellers Claims and Losses released in this Section 6.14 being referred to as the "Sellers Released Claims").

- (b) Sellers agree that neither Sellers nor their Related Parties, nor anyone claiming under, through or for them or on their behalf will bring, file, institute, prosecute, maintain, participate in, or recover upon, either directly or indirectly, or encourage or benefit from the institution of any suit, charge, administrative proceeding, investigation, or action at law or in equity against the Buyer Released Parties, or any of them, in or before any court, agency or forum, state or federal, or otherwise, for or relating to any of the Sellers Released Claims. Sellers agree that this release may be pleaded by the Buyer Released Parties as a counterclaim or cross-claim to or as a defense in bar or abatement of any Sellers Released Claim. Notwithstanding anything herein to the contrary, this Section 6.17 shall not preclude Sellers nor their Related Parties, nor anyone claiming under, through or for them from asserting any claim against the Buyer Released Parties or any of them resulting from any third-party claim against Sellers or such persons for actions of the Acquired Companies or any of the Buyer Related Parties or any claims under Article X.
- 6.18 Employees. Prior to the Closing, Sellers shall cause (i) each of the officers and employees of IRP Administrative Services, LLC ('IRP Administrative')' who spend a substantial portion of their business time performing services on behalf of the Acquired Companies rather than IRP Administrative to become officers or employees of one or more of the Acquired Companies, (ii) each of the employment agreements that any such officer or employee is a party to as set forth on Schedule 6.18 to be assigned by IRP Administrative to the applicable Acquired Company and (iii) each such officer or employee to consent to such assignment of his employment agreement, agree that such assignment is not included within the definition of "Good Reason" as defined in his or her employment agreement and waive any rights to any "change of control" or similar payments such officer or employee may have under such employment agreement from Buyer or any of the Acquired Companies. For purposes of this Agreement, any officer or employee of IRP Administrative Services that will become an officer or employee of one or more of the Acquired Companies prior to the Closing shall be deemed a "Company Employee."

ARTICLE VII

CONDITIONS PRECEDENT TO SELLERS' OBLIGATIONS TO CLOSE

Sellers' obligation to effect the Closing under this Agreement is subject to the satisfaction, at or prior to the Closing, of each of the following conditions (any of which may be waived by Agent in whole or in part):

7.1 Representations and Warranties. The representations and warranties of Buyer in this Agreement that are qualified by materiality shall be true and correct in all respects and the representations and warranties of Buyer in this Agreement that are not qualified by materiality shall be true and correct in all material respects as of the Closing Date, other than representations and warranties that expressly speak as of a specific date or time (which need only be true and correct in all respects or true and correct in all material respects, as applicable, as of such date or time), without giving effect to any supplement to the schedule.

- 7.2 <u>Performance</u>. Buyer shall have performed and complied in all material respects with all agreements and obligations required by this Agreement to be so performed or complied with by it at or prior to the Closing.
- 7.3 Officer's Certificates. Buyer shall have delivered to Agent a certificate, dated as of the Closing Date and executed by the chief executive officer of Buyer, certifying to the fulfillment of the conditions specified in Sections 7.1 and 7.2.
- 7.4 Governmental Approvals. All material Governmental Approvals with respect to the Contemplated Transactions shall have been obtained and all conditions relating to such Governmental Approvals shall have been satisfied.
- 7.5 HSR Act. Any required waiting period (and any extension thereof) applicable to the Contemplated Transactions under the HSR Act and antitrust laws shall have expired or been terminated.
- 7.6 <u>AVS List</u>. Buyer shall not be (i) listed on the Applicant Violation System maintained by the Office of Surface Mining of the United States Department of the Interior or any counterpart system maintained by each of the State of West Virginia or the Commonwealth of Kentucky, or (ii) blocked from obtaining any Permits by any applicable Governmental Authority.
- 7.7 <u>Seller Guarantees and Seller Bonds Arrangements</u> All Seller Guarantees and Seller Bonds listed on <u>Schedule 7.7</u> shall have been released or adequate credit support for the replacement of Seller Bonds shall have been provided by Buyer as described in Section 6.9(f).
- 7.8 Injunctions At the Closing there shall not be in effect any Law or Governmental Order which restrains, prohibits or declares illegal the consummation of the Contemplated Transactions and no Governmental Authority of competent jurisdiction shall have instituted or threatened a Proceeding seeking to impose any such restraint or prohibition which would have a Material Adverse Effect.
 - 7.9 Closing Deliveries Buyer shall have delivered in all material respects all of the closing deliveries required by Section 2.3(b).

ARTICLE VIII

CONDITIONS PRECEDENT TO BUYER'S OBLIGATIONS TO CLOSE

Buyer's obligations to effect the Closing under this Agreement is subject to the satisfaction, at or prior to the Closing, of each of the following conditions (any of which may be waived by Buyer in whole or in part).

- 8.1 Representations and Warranties. The representations and warranties of Sellers in this Agreement that are qualified by materiality shall be true and correct in all respects and the representations and warranties of Sellers in this Agreement that are not qualified by materiality shall be true and correct in all material respects as of the Closing Date, other than representations and warranties that expressly speak as of a specific date or time (which need only be true and correct in all respects or true and correct in all material respects, as applicable, as of such date or time), without giving effect to any supplement to the schedule.
- 8.2 <u>Performance</u>. Sellers shall have performed and complied in all material respects with all agreements and obligations required by this Agreement to be performed or complied with by it at or prior to the Closing.
- 8.3 Officer's Certificate. Sellers shall have delivered to Buyer a certificate, dated as of the Closing Date and executed by the chief executive officer of each Seller, certifying to the fulfillment of the conditions specified in Sections 8.1 and 8.2 to the Knowledge of such officer.
- 8.4 <u>Governmental Approvals</u>. All material Governmental Approvals with respect to the Contemplated Transactions shall have been obtained and all conditions relating to such Governmental Approvals shall have been satisfied, including the expiration or termination of any waiting period (and any extension thereof) applicable to the Contemplated Transactions under the HSR Act and any other antitrust laws.
- 8.5 Other Approvals. All Consents from Persons other than Governmental Authorities with respect to the Contemplated Transactions listed on Schedule 3.6 shall have been obtained.
- 8.6 <u>Injunctions</u>. At the Closing there shall not be in effect any Law or Governmental Order which restrains, prohibits or declares illegal the consummation of the Contemplated Transactions and no Governmental Authority of competent jurisdiction shall have instituted or threatened a Proceeding seeking to impose any such restraint or prohibition which would have a Material Adverse Effect.
 - 8.7 Closing Deliveries. Sellers shall have delivered, or caused to be delivered, in all material respects, all of the Closing deliveries required by Section 2.3(a).
- 8.8 Material Adverse Effect. At any time on or after the Latest Balance Sheet Date there shall not have occurred and be continuing a change in the condition of the Acquired Companies, taken as a whole, that has resulted in a Material Adverse Effect.

- 8.9 <u>Permits</u>. For each Permit that is not transferred at Closing in accordance with Section 6.10, Sellers shall have fully complied with all of the terms and obligations of Section 6.10(b).
- 8.10 Repayment of Indebtedness. Sellers shall have repaid all the Indebtedness listed on Schedule 2.6 as to be repaid at or before Closing and shall have delivered to Buyer evidence of such payment in form and substance satisfactory to Buyer.
 - 8.11 Redemption. The Redemption shall have been completed and the Contingency Obligations satisfied in accordance with the provisions of this Agreement.
- 8.12 Environmental Reports. Sellers shall have caused each Acquired Company to have completed, to the reasonable satisfaction of Buyers, each of the matters identified in the Modified Phase I Environmental Site Assessment Report for Mine Operations Associated with International Resource Partners West Virginia and Kentucky Operations February 2011, under the heading "The following housekeeping issues and recommendations were also noted during the site assessment."
- 8.13 <u>Shared Services Agreement</u>. Holdings II and International Industries shall have executed and delivered a First Amendment to the Shared Services Agreement between them dated as of June 12, 2007, giving Holdings II the right to terminate the Shared Services Agreement at any time, with or without cause, at its sole discretion.

ARTICLE IX

TERMINATION

- 9.1 <u>Termination</u>. This Agreement may, by notice given prior to or at the Closing, be terminated:
- (a) by either Buyer or Agent if a material breach of any provision of this Agreement has been committed by the other party (including, in the case of termination by Buyer, the Agent or Sellers) and such breach has not been waived by the party to waive such condition;
- (b) (i) by Buyer if any of the conditions in Article VIII has not been satisfied in all material respects as of the Closing Date or if satisfaction of such a condition is or becomes impossible (other than through the failure of Buyer to comply with its obligations under this Agreement) and Buyer has not waived such condition on or before the Closing Date; or
- (ii) by Agent, if any of the conditions in Article VII has not been satisfied in all material respects as of the Closing Date or if satisfaction of such a condition is or becomes impossible (other than through the failure of Sellers or Agent to comply with their obligations under this Agreement) and Agent have not waived such condition on or before the Closing Date;
 - (c) by mutual consent of Buyer and Agent; or

- (d) by either Buyer or Agent if the Closing has not occurred (other than through the failure of any party seeking to terminate this Agreement (including in the case of Agent, Sellers) to comply fully with its obligations under this Agreement) on or before September 1, 2011 (the "<u>Drop Dead Date</u>"), or such later date as the parties may agree upon.
- 9.2 <u>Procedure and Effect of Termination</u>. Each party's right of termination under Section 9.1 is in addition to any other rights it may have under this Agreement or otherwise, and the exercise of a right of termination will not be an election of remedies. If this Agreement is terminated pursuant to Section 9.1, all further obligations of the parties under this Agreement will terminate, except that the obligations in Sections 6.11 and 11.2 will survive; provided, however, that if this Agreement is terminated by a party because of the breach of the Agreement by the other party or because one or more of the conditions to the terminating party's obligations under this Agreement is not satisfied as a result of the other party's failure to comply with its obligations under this Agreement, the terminating party's right to pursue all legal remedies (including damages and attorney's fees) available at law or equity, including the remedy of specific performance as set forth in Section 11.18, will survive such termination unimpaired.

ARTICLE X

INDEMNIFICATION

10.1 Indemnification.

- (a) <u>Indemnification by Sellers</u>. Subject to the limitations set forth in this Section 10.1, Sellers agree to indemnify, defend and hold Buyer and its respective Affiliates and their respective officers, directors, partners, members, stockholders, employees, agents, representatives, successors and permitted assigns (collectively, the "<u>Buyer Indemnitees</u>"), harmless from and in respect of any and all Losses, net of the proceeds from any insurance policies or other third party reimbursement for, or any tax benefit Buyer actually receives as a result of, such loss, and net of any Losses incurred by Seller Indemnitees and not previously paid pursuant to which they are entitled to indemnification under Article X that they may incur arising solely out of or related to:
- (i) any breach of any representation or warranty made by Sellers, Resource Partners or Agent (in each case without regard to any "Material Adverse Effect" or other "materiality" qualifier contained therein) in this Agreement, the Schedules, or any other Transaction Document delivered by Sellers or Agent pursuant to this Agreement;
- (ii) any breach of any representation or warranty made by Sellers or Resource Partners (in each case without regard to any "Material Adverse Effect" or other "materiality" qualifier contained therein) in this Agreement as if such representation or warranty were made on and as of the Closing Date;
- (iii) any breach by Sellers, Resource Partners or Agent of any covenant or obligation of Sellers, Resource Partners or Agent in this Agreement or any other Transaction Document delivered pursuant to this Agreement;

(iv) any claim by any Person for brokerage or finder's fees or commissions or similar payments based upon any agreement or understanding alleged to have be	een
made by any such Person with Sellers or any Acquired Company (or any Person acting on their behalf) in connection with any of the Contemplated Transactions;	

- (v) the Redemption and any Contingency Obligations;
- (vi) any Buyer Indemnified Taxes; or
- (vii) any Special Indemnity Matter.
- (b) <u>Indemnification by Buyer</u>. Subject to the limitations set forth in this Section 10.1, Buyer agrees to indemnify, defend and hold Sellers and their Affiliates and their respective officers, directors, partners, members, stockholders, employees, agents, representatives, successors and permitted assigns (collectively, the "<u>Seller Indemnitees</u>"), harmless from and in respect of any and all Losses, net of the proceeds from any insurance policies or other third party reimbursement for, or any tax benefit Sellers actually receives as a result of, such loss, that they may incur arising solely out of or relating to:
- (i) any breach of any representation or warranty made by Buyer (in each case without regard to any "materiality" qualifier contained therein) in this Agreement, the Schedules, or any other Transaction Document delivered by Buyer pursuant to this Agreement;
- (ii) any breach of any representation or warranty made by Buyer (in each case without regard to any "materiality" qualifier contained therein) in this Agreement as if such representation or warranty were made on and as of the Closing Date;
 - (iii) any breach by Buyer of any covenant or obligation of Buyer in this Agreement or any other Transaction Document delivered pursuant to this Agreement; or
- (iv) any claim by any Person for brokerage or finder's fees or commissions or similar payments based upon any agreement or understanding alleged to have been made by any such Person with Buyer or any Affiliate (or any Person acting on their behalf) in connection with any of the Contemplated Transactions;

provided, however, that Buyer shall have no obligation to indemnify any Seller Indemnitee to the extent that Sellers are required to indemnify the Buyer Indemnitees or would be so required but for the limitations set forth in Section 10.1(d).

(c) <u>Survival of Representations</u>, <u>Warranties and Indemnities</u>. The representations, warranties, covenants, undertakings and agreements of the parties contained in or made pursuant to this Agreement or in any instrument delivered pursuant hereto, and the rights of the parties to seek indemnification with respect thereto, shall survive the Closing for a period of fourteen (14) months from the Closing Date, except that the representations and warranties in Section 3.3 (Interests) and Section 3A.4 (Interests) shall survive indefinitely (for each type of indemnification claim, the "<u>Indemnity Termination Date</u>"). Any claim for indemnification under this Agreement which is made in good faith and in a writing specifying the factual basis of the claim in reasonable detail to the extent then known by the claiming party prior to the expiration of the applicable Indemnity Termination Date shall survive such expiration until mutually resolved or otherwise determined hereunder, as applicable, and the applicable Indemnity Termination Date for all purposes hereunder shall automatically be extended with respect to such claim (but not any other claims) until such claim is so mutually resolved or otherwise determined hereunder. Any claim not so made prior to the expiration of the applicable Indemnity Termination Date shall be deemed to have been waived.

- (i) The maximum aggregate amount of indemnifiable Losses under Article X arising out of or resulting from the causes that may be recovered from Sellers shall not exceed the amounts then remaining of the General Escrow Amount in the Escrow Fund; provided that, notwithstanding anything in this Section 10.1(d) to the contrary, in no event will the limitations set forth in this Section 10.1(d)(i) apply with respect to any breach of any representation, warranty, or covenant or obligation pursuant to Section 3.3 (Interests) and Section 3A.4 (Interests); provided, however, that no Seller shall be liable for the amount of idemnifiable Losses exceeding, on an aggregate basis, the portion of the Purchase Price received by such Seller;
- (ii) No party shall be entitled to any recovery unless and until the total of all claims brought by such party for indemnity or damages pursuant to this Section 10.1 exceeds \$500,000 and such party then shall be entitled to recover only the amount by which such claims for indemnity or damages exceed \$500,000; provided that, notwithstanding anything in this Section 10.1(d) to the contrary, in no event will the limitations set forth in this Section 10.1(d)(ii) apply (x) in the event of fraud, willful misconduct or bad faith, (y) with respect to any breach of any representation, warranty or covenant or obligation pursuant to Section 3.3 (Interests) and Section 3A.4 (Interests), and (z) with respect to any claim for indemnification pursuant to Section 10.1(a)(iv), (v), (vi), or (vii); and
- (iii) The remedies set forth in this Article X and Sections 2.7 and 2.8 and the specific performance remedy referenced in Section 11.18 shall provide the sole and exclusive remedies arising from this Agreement.

(e) Procedure for Indemnification - Third Party Claims.

(i) Promptly after receipt by an indemnified party under Section 10.1(a) or Section 10.1(b) of notice of the commencement of any proceeding against it, such indemnified party will, if a claim is to be made against an indemnifying party under such Section, give notice to the indemnifying party of the commencement of such claim, but the failure to notify the indemnifying party will not relieve the indemnifying party of any Liability that it may have to any indemnified party, except to the extent that the indemnifying party demonstrates that the defense of such action is prejudiced by the indemnifying party's failure to give such notice.

- (ii) If any proceeding referred to in Section 10.1(e)(i) is brought against an indemnified party and it gives notice to the indemnifying party of the commencement of such proceeding, the indemnifying party will be entitled to participate in such proceeding and, to the extent that it wishes (unless (A) the indemnifying party is also a party to such proceeding and the indemnified party determines in good faith that joint representation would be inappropriate, or (B) the indemnifying party fails to provide reasonable assurance to the indemnified party of its financial capacity to defend such proceeding and provide indemnification with respect to such proceeding), to assume the defense of such proceeding with counsel satisfactory to the indemnified party and, after notice from the indemnifying party to the indemnified party of its election to assume the defense of such proceeding, the indemnifying party will not, as long as it diligently conducts such defense, be liable to the indemnified party under this Section 10 for any fees of other counsel or any other expenses with respect to the defense of such proceeding, in each case subsequently incurred by the indemnified party in connection with the defense of such proceeding, other than reasonable costs of investigation. If the indemnifying party assumes the defense of a proceeding, (1) it will be conclusively established for purposes of this Agreement that the claims made in that proceeding are within the scope of and subject to indemnification; and (2) no compromise or settlement of such claims may be effected by the indemnifying party without the indemnified party's consent unless (x) there is no finding or admission of any violation of Laws or any violation of the rights of any Person and no effect on any other claims that may be made against the indemnified party, and (y) the sole relief provided is monetary damages that are paid in full by the indemnifying party of the commencement of any proceeding and the indemnifying party does not, within ten (10) days after
- (iii) Notwithstanding the foregoing, if an indemnified party determines in good faith that there is a reasonable probability that a proceeding may adversely affect it or its Affiliates other than as a result of monetary damages for which it would be entitled to indemnification under this Agreement, the indemnified party may, by notice to the indemnifying party, assume the exclusive right to defend, compromise, or settle such proceeding, but the indemnifying party will not be bound by compromise or settlement effected without its consent (which may not be unreasonably withheld).
- (f) <u>Procedure for Indemnification Other Claims</u>. A claim for indemnification for any matter not involving a third-party claim may be asserted by notice to the party from whom indemnification is sought.
- (g) <u>Punitive Damages</u>. Notwithstanding any other provision of this Agreement to the contrary, no indemnifying party shall be liable to an indemnified party for any punitive damages, special damages, indirect damages, or similar items except to the extent that a Buyer Indemnitee has become liable to any third party for such punitive damages, special damages, indirect damages, or similar items resulting from a third party claim for which indemnification is provided for in Section 10.1(a).

10.2 <u>Treatment of Indemnification Payments</u>. All indemnification payments made under this Agreement shall be treated by Buyer and Sellers as an adjustment to the Purchase Price.

ARTICLE XI

MISCELLANEOUS

11.1 Agent. IRP GP ("Agent") is hereby appointed as agent and attorney-in-fact for each Seller, and through the Closing, for each Acquired Company, for and on behalf of Sellers and the Acquired Companies, to act as Agent for Sellers and the Acquired Companies under this Agreement and each other Transaction Document, and to give and receive notices and communications, to agree to, negotiate, enter into settlements and compromises of, and demand arbitration and comply with orders of courts and awards of arbitrators with respect to claims for indemnification, to authorize delivery to Buyer or any Buyer Indemnitee of any payment hereunder, and to take all actions necessary or appropriate in the judgment of Agent for the accomplishment of the foregoing in accordance with the terms and provisions of this Agreement and each other Transaction Document. Notices or communications to or from Agent hereunder shall constitute notice to or from each Seller or, at or prior to the Closing, the Acquired Companies. Sellers and the Acquired Companies hereby agree that the appointment of Agent pursuant to this Section 11.1 shall be irrevocable except as otherwise provided herein or by nonwaivable provisions of Applicable Law. Notwithstanding the foregoing, such appointment shall be automatically revoked as to the Acquired Companies as of the completion of the Closing. Any decision, act, consent or instruction of Agent relating to this Agreement or any other Transaction Document shall constitute a decision of all of Sellers and, if prior to or at the Closing, the Acquired Companies, and shall be final, binding and conclusive upon each Seller and the Acquired Companies (but, with respect to the Acquired Companies, only to the extent related to any time period ending at or prior to the Closing) and Buyer may rely upon any such written decision, consent or instruction of Agent as being the decision, consent or instruction of each and every Seller and the Acquired Companies. Buyer and the other Buyer Indemnitees are hereby relieved from any liability to any Person for any acts done by them in accordance with such decision, consent or instruction of Agent. Sellers hereby jointly and severally agree to indemnify and hold Agent, its affiliates and their officers, employees, successors, assigns, attorneys and agents harmless from all losses, costs, claims, demands, expenses, damages, penalties and attorney's fees suffered or incurred by such Agent as a result of anything which it may do or refrain from doing in connection with this Agreement or any Transaction Document or any litigation or cause or action arising from or in conjunction with this Agreement or any Transaction Document or involving the subject matter hereof; provided that the foregoing indemnification shall not extend to the gross negligence or willful misconduct of Agent.

- 11.2 Fees and Expenses. Except as otherwise provided in this Agreement, Sellers and Buyer shall bear their own expenses and the expenses of their respective Affiliates (including, in the case of Sellers, each Acquired Company and Agent) in connection with the preparation and negotiation of this Agreement and the consummation of the Contemplated Transactions. Notwithstanding the foregoing, Sellers and Buyer shall each pay one-half of the fee for the Notification and Report Forms filed with the Department of Justice and the Federal Trade Commission under the HSR Act. Sellers and Buyer shall bear the fees and expenses of any broker or finder retained by such party and their respective Affiliates (including, in the case of Sellers, each Acquired Company and Agent) in connection with the Contemplated Transactions.
 - 11.3 Governing Law. This Agreement shall be construed under and governed by the Laws of the State of New York without regard to conflicts of laws principles.
- 11.4 <u>Amendment</u>. This Agreement may not be amended, modified or supplemented except upon the execution and delivery of a written agreement that specifically references this Agreement and is executed by the party to be charged with the amendment. Any amendment, modification or supplement signed by Agent shall be binding on and effective against all Sellers and, prior to Closing, the Acquired Companies.
- 11.5 <u>Assignment</u>. Neither this Agreement nor any of the rights, interests or obligations hereunder shall be assigned by any party hereto without the prior written consent of the other parties, except that Buyer may assign any of its rights under this Agreement to any of its Affiliates at any time. Subject to the preceding sentence, this Agreement will be binding upon, inure to the benefit of and be enforceable by, the parties hereto and their respective successors and permitted assigns, and is not intended to confer upon any Person other than the parties hereto and their respective successors and permitted assigns any rights or remedies hereunder.
- 11.6 Waiver. The rights and remedies of the parties to this Agreement are cumulative and not alternative. Neither the failure nor any delay by any party in exercising any right, power, or privilege under this Agreement or the documents referred to in this Agreement will operate as a waiver of such right, power, or privilege, and no single or partial exercise of any such right, power, or privilege will preclude any other or further exercise of such right, power, or privilege or the exercise of any other right, power, or privilege. To the maximum extent permitted by applicable Law, (a) no claim or right arising out of this Agreement or the documents referred to in this Agreement can be discharged by one party, in whole or in part, by a waiver or renunciation of the claim or right unless in writing signed by the other party; (b) no waiver that may be given by a party will be applicable except in the specific instance for which it is given; and (c) no notice to or demand on one party will be deemed to be a waiver of any obligation of such party or of the right of the party giving such notice or demand to take further action without notice or demand as provided in this Agreement or the documents referred to in this Agreement. Any waiver given by Agent shall be binding on and effective against all Sellers and, prior to Closing, the Acquired Companies.

11.7 Notices.

- (a) Any notice, demand, or communication required or permitted to be given by any provision of this Agreement shall be deemed to have been sufficiently given or served for all purposes if (i) personally delivered, (ii) sent by a internationally recognized overnight courier service, (iii) sent by registered or certified mail, return receipt requested, postage prepaid, or (iv) successfully transmitted by facsimile with confirmation of receipt to the recipient at the address or facsimile number, as applicable, indicated on Schedule 11.7 or to such other address or facsimile number as any party hereto may, from time to time, designate in a written notice given in like manner, provided, however, that any notice of change of address or facsimile number shall be effective only upon receipt. Any notice given to or by Agent in accordance with this Section will be deemed given to or by all Sellers.
- (b) Except as otherwise provided herein, any notice under this Agreement will be deemed to have been given (i) on the date such notice is personally delivered or delivered by facsimile, (ii) the next succeeding Business Day after the date such notice is delivered to the overnight courier service if sent by overnight courier, or (iii) five (5) Business Days after the date such notice is sent by registered or certified mail; provided that in each case notices received after 4:00 p.m. (local time of the recipient) shall be deemed to have been duly given on the next Business Day.
- 11.8 Complete Agreement. This Agreement, the Schedules and the other documents and writings referred to herein or delivered pursuant hereto contain the entire understanding of the parties with respect to the subject matter hereof and thereof and supersede all prior agreements and understandings, both written and oral, between the parties with respect to the subject matter hereof.
- 11.9 Counterparts. This Agreement may be executed in one or more counterparts, all of which shall be considered one and the same agreement and each of which shall be deemed an original.
- 11.10 <u>Publicity</u>. Sellers and Buyer will consult with each other and will mutually agree upon any publication or press release of any nature with respect to this Agreement or the Contemplated Transactions and shall not issue any such publication or press release prior to such consultation and agreement except as may be required by Law or by obligations pursuant to any listing agreement with any securities exchange or any securities exchange regulation or the rules, regulations or orders of any Governmental Authority, in which case the party proposing to issue such publication or press release shall make all reasonable efforts to consult with the other party before issuing any such publication or press release. Unless consented to by Agent and Buyer in advance or required by Law or by obligations pursuant to any listing agreement with any securities exchange or any securities exchange regulation or the rules, regulations or orders of any Governmental Authority, each party shall, and shall cause its Affiliates to, keep this Agreement strictly confidential and may not make any disclosure of this Agreement to any Person.

- 11.11 Headings. The headings, table of contents and index contained in this Agreement are for reference only and shall not affect in any way the meaning or interpretation of this Agreement.
- 11.12 Severability. Any provision of this Agreement which is invalid, illegal or unenforceable in any jurisdiction shall, as to that jurisdiction, be ineffective to the extent of such invalidity, illegality or unenforceability, without affecting in any way the remaining provisions hereof in such jurisdiction or rendering that or any other provision of this Agreement invalid, illegal or unenforceable in any other jurisdiction.
- 11.13 <u>Third Parties</u>. Nothing herein expressed or implied is intended or shall be construed to confer upon or give to any Person or corporation, other than the parties hereto and their permitted successors or assigns, any rights or remedies under or by reason of this Agreement, except that the Financing sources are hereby made express third-party beneficiaries of Section 6.17, 11.3, 11.14 and 11.19.

11.14 Jurisdiction; Service of Process.

- (a) Any action or proceeding seeking to enforce any provision of, or based on any right arising out of, this Agreement shall be brought against any of the parties in the United States District Court for the Southern District of New York, and each of the parties consents to the exclusive jurisdiction of such court (and of the appropriate appellate courts) in any such action or proceeding and waives any objection to venue laid therein. Process in any action or proceeding referred to in the preceding sentence may be served on any party anywhere in the world.
- (b) Notwithstanding the foregoing, each of the parties hereto hereby agrees (i) that any claim, suit, action or proceeding of any kind or description, whether in law or in equity, whether in contract or in tort or otherwise, involving any Financing source, or any of its Affiliates or representatives, arising out of or relating to the Contemplated Transactions, the Commitment Letter or the accompanying fee letter or the performance of services or consummation of transactions thereunder, shall be subject to the exclusive jurisdiction of any court of competent jurisdiction located within the Borough of Manhattan in the City of New York, New York, whether a state or Federal court, and (ii) not to bring, or permit any of its Affiliates to bring or support any other Person in bringing, any such claim, suit, action or proceeding in any other court.
- 11.15 <u>Further Assurances</u>. The parties agree (a) to furnish upon request to each other such further information, (b) to execute and deliver to each other such other documents, and (c) to do such other acts and things, all as the other party may reasonably request for the purpose of carrying out the intent of this Agreement and the documents referred to in this Agreement.
 - 11.16 Time of the Essence. With regard to all dates and time periods set forth or referred to in this Agreement, time is of the essence.
- 11.17 <u>Construction</u>. The parties have participated jointly in the negotiating and drafting of this Agreement. If an ambiguity of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the parties and no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any of the provisions of this Agreement.

- 11.18 <u>Specific Performance</u>. Each of the parties acknowledges and agrees that the other parties would be damaged irreparably in the event any of the provisions of this Agreement are not performed in accordance with their specific terms or otherwise are breached. Accordingly, each of the parties agrees that the other parties shall be entitled to an injunction or injunctions to prevent breaches of the provisions of this Agreement and to enforce specifically this Agreement and the terms and provisions hereof, in addition to any other remedy to which they may be entitled, at law or in equity.
- 11.19 <u>WAIVER OF JURY TRIAL</u>. EACH OF THE PARTIES HERETO IRREVOCABLY WAIVES ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY LEGAL PROCEEDING BETWEEN THE PARTIES HERETO ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY.
- 11.20 <u>Consent of Limited Partners</u>. Each of Lightfoot, International, KED and Tortoise, by executing this Agreement, hereby authorizes and consents to the sale of the interests of IRP GP in Resource Partners to Buyer. In accordance with, and pursuant to the terms of the Second Amended and Restated Agreement of Limited Partnership of International Resource Partners LP dated as of April 18, 2008 (the "Partnership Agreement"), Buyer, as transferee of the general partnership units of Resource Partners agrees to assume the rights and duties of IRP GP under the Partnership Agreement and to be bound by the provisions of the Partnership Agreement, as it may be amended from time to time.

IN WITNESS WHEREOF, each of the parties hereto has caused this Purchase Agreement to be executed by its duly authorized officer, in each case as of the date first above written.

JAMES RIVER COAL COMPANY

By: /s/ Peter T. Socha
Peter T. Socha

Chairman and Chief Executive Officer

Signature Page to Purchase Agreement

IN WITNESS WHEREOF, each of the parties hereto has caused this Purchase Agreement to be executed by its duly authorized officer, in each case as of the date first above written.

LIGHTFOOT CAPITAL PARTNERS, LP

By: Lightfoot Capital Partners GP LLC

Its: General Partner

By: /s/ Vincent T. Cubbage

Name: Vincent T. Cubbage Title: Chief Executive Officer

Signature Page to Purchase Agreement

IN WITNESS WHEREOF, each of the parties hereto has caused this Purchase Agreement to be executed by its duly authorized officer, in each case as of the date first above written.

INTERNATIONAL INDUSTRIES, INC.

By: /s/ James H. Harless

Name: James H. Harless Title: Chairman

Signature Page to Purchase Agreement

KAYNE ANDERSON ENERGY DEVELOPMENT COMPANY

By: /s/ James C. Baker

Name: James C. Baker Title: Vice President

TORTOISE CAPITAL RESOURCES CORPORATION

By: /s/ Ed Russell

Name: Ed Russell Title: President

INTERNATIONAL RESOURCE PARTNERS GP LLC

By: /s/ Vincent T. Cubbage

Name: Vincent T. Cubbage
Title: Chief Executive Officer

AGENT: INTERNATIONAL RESOURCE PARTNERS GP LLC

By: /s/ Vincent T. Cubbage

Name: Vincent T. Cubbage
Title: Chief Executive Officer

ESCROW AGREEMENT

by and among

JAMES RIVER COAL COMPANY,

WELLS FARGO BANK, NATIONAL ASSOCIATION, as escrow agent

and

INTERNATIONAL RESOURCE PARTNERS GP LLC
As agent for Sellers

ESCROW AGREEMENT

This Escrow Agreement (this "Escrow Agreement") is made and entered into as of the day of 2011, by and among:		
(1) JAMES RIVER COAL COMPANY, a corporation incorporated under the Laws of the State of Virginia ('Buyer');		
(2) WELLS FARGO BANK, NATIONAL ASSOCIATION, a national banking association, as escrow agent ('Escrow Agent''); and		
(3) INTERNATIONAL RESOURCE PARTNERS GP LLC, a limited liability company organized under the Laws of the State of Delaware, solely in its capacity as agent for Sellers ("Seller Agent");		
WITNESSETH:		
Whereas, Lightfoot Capital Partners, LP, a Delaware limited partnership (" <u>Lightfoot</u> "), International Industries, Inc., a West Virginia corporation (" <u>International Industries</u> "), International Resource Partners GP, LLC, a Delaware limited liability company (" <u>IRP GP</u> "), Kayne Anderson Energy Development Company, a Maryland limited partnership (" <u>KED</u> ") and Tortoise Capital Resources Corporation, a Maryland corporation (" <u>Tortoise</u> ", and Lightfoot, International Industries, IRP GP, KED and Tortoise, each a " <u>Seller</u> " and collectively " <u>Sellers</u> "), are parties to that certain Purchase Agreement, dated as of March, 2011 (the " <u>Purchase Agreement</u> ") pursuant to which Buyer will purchase all of the outstanding limited partnership units and general partnership units of International Resource Partners LP, a Delaware limited partnership (" <u>Resource Partners</u> ");		
Whereas, Sellers own all of the outstanding limited partnership units and general partnership units of Resource Partners;		
Whereas, the Sellers have approved the Purchase Agreement and the transactions contemplated thereby, including this Escrow Agreement;		
Whereas, under the Purchase Agreement, each of Buyer and its Affiliates and their respective officers, directors, partners, members, stockholders, employees, agents, representatives, successors and permitted assigns (collectively, the "Buyer Indemnitees") is entitled to certain rights to indemnification as provided in Article X and payments as provided in Section 2.8(b) and Section 2.8(c) of the Purchase Agreement;		
Whereas, to provide funds for the Escrow Fund (as defined below), Buyer will deposit with the Escrow Agent an aggregate amount equal to \$30,000,000; and		
Whereas, the Escrow Agent has agreed to hold, invest, release and distribute the Escrow Fund pursuant to the terms of this Escrow Agreement.		
-1-		

Now, Therefore, for and in consideration of the premises and the mutual covenants and agreements contained in this Escrow Agreement, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, and intending to be legally bound hereby, the parties agree as follows:

1. CERTAIN DEFINITIONS.

For purposes of this Escrow Agreement, the capitalized terms used but not defined herein shall have the meanings set forth in the Purchase Agreement.

2. APPOINTMENT OF THE ESCROW AGENT.

The Escrow Agent is hereby constituted and appointed as the escrow agent under this Escrow Agreement and the Escrow Agent accepts this appointment in accordance with the terms and conditions set forth in this Escrow Agreement and agrees to assume and perform the duties of the escrow agent pursuant to this Escrow Agreement.

3. DEPOSIT AND USE OF THE ESCROW AMOUNT.

- 3.1 Buyer shall deposit with the Escrow Agent an aggregate amount of \$30,000,000 (the "Escrow Amount") in immediately available funds on the date hereof with the Escrow Agent, of which (i) \$20,000,000 shall comprise the "General Escrow Amount" and (ii) \$10,000,000 shall comprise the "Special Escrow Amount". The Escrow Agent shall deposit such funds, and all earnings and interest on, and proceeds of such funds, into a segregated account (all of such amounts related to the General Escrow Amount, the "General Escrow Fund" and all of such amount related to the Special Escrow Amount, the "Special Escrow Fund" and, collectively, the "Escrow Fund"). The Escrow Fund shall be used to (i) satisfy any payment obligation to Buyer in the event that 10,000,000 tons of Additional Reserves (as defined in the Purchase Agreement) are not proven by the Escrow Release Date pursuant to Section 2.8(b) of the Purchase Agreement, (iii) satisfy any reimbursement of drilling and other reasonable out-of-pocket expenses up to \$1,000,000 in connection with any efforts to prove Additional Reserves pursuant to Section 2.8(b) of the Purchase Agreement; (iii) satisfy any indemnification liabilities of Sellers for fines, legal fees and damages related to any Prior Violations (as defined in the Purchase Agreement) pursuant to Section 2.8(c) of the Purchase Agreement and (iv) satisfy any indemnification liabilities of the Sellers pursuant to Article X of the Purchase Agreement.
 - 3.2 Immediately after receipt from Buyer of the Escrow Amount, the Escrow Agent shall confirm to Buyer and Seller Agent such receipt in writing.
- 3.3 The Escrow Agent agrees to hold, invest, release and distribute the Escrow Fund, and to act as Escrow Agent in accordance with the terms, conditions and provisions of this Escrow Agreement.

4. DISPOSITION OF THE ESCROW FUND.

4.1 The Escrow Fund shall be released and distributed by the Escrow Agent in accordance with the provisions set forth below and, in any event, upon its receipt of, and in accordance with, joint written instructions from Buyer and Seller Agent. Buyer and Seller Agent agree to jointly instruct the Escrow Agent to release and disburse the amounts in the Escrow Fund in a manner consistent with the terms and conditions of the Purchase Agreement and this Escrow Agreement.

4.2 Each Buyer Indemnitee, to the extent entitled to receive payment pursuant to Section 2.8(b) or to be indemnified pursuant to Article X of the Purchase Agreement or, if Buyer so elects pursuant to Section 4.3(e) hereof, pursuant to Section 2.8(c)(iii) of the Purchase Agreement, shall be entitled to claim such amounts as provided in the Purchase Agreement and receive payments from the General Escrow Fund, in each case, in accordance with the following provisions:

If any Buyer Indemnitee believes that it has incurred any Losses for which it is entitled to indemnification under Article X of the Purchase Agreement or payment under Sections 2.8(b) or 2.8(c)(iii), Buyer shall deliver to the Escrow Agent written notice (a "Claim Notice") (i) naming the Buyer Indemnitee making the claim (the "Buyer Indemnified Person"), (ii) describing such Losses and, to the extent known, the amount thereof (the 'Claim Amount"), (iii) stating that the Buyer Indemnified Person is entitled to payment under Section 2.8(b) or Section 2.8(c)(iii) or indemnification under Article X of the Purchase Agreement for such Losses and providing a reasonable explanation of the basis thereof, and (iv) demanding payment in the amount of such Losses, to the extent such Losses are known and then incurred. The Claim Notice must be given prior to the Escrow Release Date in order to be eligible to receive payments from the General Escrow Fund. At the time of delivery of any Claim Notice to the Escrow Agent, a duplicate copy of such Claim Notice shall be delivered by Buyer to Seller Agent. Promptly, and in any event within two (2) Business Days after its receipt of the Claim Notice (and notwithstanding any prior delivery of the Claim Notice by Buyer to Seller Agent), the Escrow Agent shall deliver a copy of the Claim Notice to Seller Agent. Subject to Paragraph 4.2(b) hereof, the Escrow Agent shall, promptly after the thirtieth (30h) day after the Escrow Agent's delivery of a copy of the Claim Notice to Seller Agent (such 30-day period, the "Response Period"), (x) with respect to indemnification or payment for a specified amount, pay or distribute to the Buyer Indemnified Person named in the Claim Notice cash out of the General Escrow Fund in the amount specified in the Claim Notice, designate and segregate out of the General Escrow Fund the contingent amount subject to the Claim Notice. Payment shall be distributed as specified in the Claim Notice, designate and segregate out of the General Escrow Fund the conting

Notwithstanding the provisions of <u>Paragraph 4.2(a)</u> hereof, the Escrow Agent shall not make any payment from the General Escrow Fund with respect to a Claim Notice if, during the Response Period, Seller Agent shall have delivered to the Escrow Agent a written objection to the claim for indemnification or payment made or the Claim Amount set forth in the Claim Notice (an "<u>Objection</u>"). If the Escrow Agent does not receive an Objection within the Response Period, then Seller Agent shall be deemed to have acknowledged the obligation to indemnify or make such payment, as applicable, and the correctness of such Claim Amount for the full amount thereof subject to a current claim for payment as specified in the Claim Notice. At the time of delivery of any Objection to the Escrow Agent, a duplicate copy of such Objection shall be delivered by Seller Agent to Buyer.

Upon receipt of an Objection pursuant to this Escrow Agreement, the Escrow Agent shall (i) deliver notice of the Objection to Buyer, (ii) distribute to the Buyer Indemnified Person cash out of the General Escrow Fund in an amount equal to that portion, if any, specified in the Claim Notice which is not disputed by Seller Agent in such Objection, and (iii) designate and segregate out of the General Escrow Fund the amount subject to the Claim Notice which is disputed by Seller Agent in the Objection or is not then payable pursuant to the terms of the Claim Notice. Thereafter, the Escrow Agent shall not distribute to the Buyer Indemnified Person or Seller Agent the segregated portion of the General Escrow Fund subject to the Claim Notice until the Escrow Agent shall have received joint written instructions from Buyer and Seller Agent, in form and substance reasonably satisfactory to the Escrow Agent, concerning the determination of the amount of the indemnification payment, if any, to which the Buyer Indemnified Person is entitled

After an Objection has been delivered to the Escrow Agent and Buyer, Buyer and Seller Agent shall attempt in good faith to agree upon the rights of the Buyer Indemnified Person with respect to each of the amounts that comprise the asserted Claim Amount (or the amount subject to the Claim Notice which is disputed by Seller Agent in the Objection). If Buyer and Seller Agent should so agree, a memorandum setting forth such agreement shall be prepared and signed by both Buyer and Seller Agent and, in the case of a demand for recovery from the General Escrow Fund, shall deliver to the Escrow Agent joint written instructions from Buyer and Seller Agent setting forth such agreement. If no such agreement can be reached after good faith negotiation within 30 days after delivery of an Objection, either Seller Agent or Buyer may pursue any and all remedies as shall be available to such party pursuant to the Purchase Agreement; provided, however, that if such dispute relates to a claim for payment related to any Prior Violations pursuant to Section 2.8(c)(iii) of the Purchase Agreement, the amounts in dispute shall be submitted to binding arbitration as set forth in Section 2.8(c)(iv) of the Purchase Agreement. The Escrow Agent will (i) pay the Buyer Indemnified Person out of the General Escrow Fund the amount that the Buyer Indemnified Person is entitled to receive as promptly as practicable after receiving joint written instructions concerning such final determination from Buyer and Seller Agent and (ii) if applicable, cause to be distributed to Seller Agent, for the benefit of and distribution to Sellers, such amount as to which Sellers would be entitled to distribution under Paragraph 4.6 hereof after giving effect to the foregoing joint written instructions.

4.3 Buyer, to the extent entitled to payment related to any Prior Violations pursuant to Section 2.8(c) of the Purchase Agreement, shall be entitled to claim such amounts equal to such Buyer Indemnitee's Losses and receive payments from the Special Escrow Fund, in each case, in accordance with the following provisions:

(a) If Buyer has incurred any Losses or other damages for which it is entitled to payment under Section 2.8(c) of the Purchase Agreement, Buyer shall deliver to the Escrow Agent a Claim Notice (i) describing such Losses and, to the extent known, the Claim Amount, (ii) stating that the Buyer Indemnified Person is entitled to payment under Section 2.8(c) of the Purchase Agreement for such Losses and providing a reasonable explanation of the basis thereof, and (iii) demanding payment in the amount of such Losses, to the extent such Losses are known and then incurred. The Claim Notice must be given prior to the Final Release Date in order to be eligible to receive payments from the Special Escrow Fund. At the time of delivery of any Claim Notice to the Escrow Agent, a duplicate copy of such Claim Notice shall be delivered by Buyer to Seller Agent. Promptly, and in any event within two (2) Business Days after its receipt of the Claim Notice (and notwithstanding any prior delivery of the Claim Notice by Buyer to Seller Agent), the Escrow Agent shall deliver a copy of the Claim Notice to Seller Agent. Subject to Paragraph 4.3(b) hereof, the Escrow Agent shall, promptly after the Response, (x) with respect to payment for a specified amount, pay or distribute to Buyer, cash out of the Special Escrow Fund in the amount specified in the Claim Notice or (y) with respect to contingent payment claims not then payable pursuant to the terms of the Claim Notice, designate and segregate out the contingent amount subject to the Claim Notice. Payment shall be distributed as specified in the Claim Notice.

- (b) Notwithstanding the provisions of <u>Paragraph 4.3(a)</u> hereof, the Escrow Agent shall not make any payment from the Special Escrow Fund with respect to a Claim Notice if, during the Response Period, Seller Agent shall have delivered to the Escrow Agent an Objection. If the Escrow Agent does not receive an Objection within the Response Period, then Seller Agent shall be deemed to have acknowledged the obligation to indemnify and the correctness of such Claim Amount for the full amount thereof subject to a current claim for payment as specified in the Claim Notice. At the time of delivery of any Objection to the Escrow Agent, a duplicate copy of such Objection shall be delivered by Seller Agent to Buyer.
- (c) Upon receipt of an Objection pursuant to this Escrow Agreement, the Escrow Agent shall (i) deliver notice of the Objection to Buyer, (ii) distribute to Buyer cash out of the Special Escrow Fund in an amount equal to that portion, if any, specified in the Claim Notice which is not disputed by Seller Agent in such Objection, and (iii) designate and segregate out of the Special Escrow Fund the amount subject to the Claim Notice which is disputed by Seller Agent in the Objection or is not then payable pursuant to the terms of the Claim Notice. Thereafter, the Escrow Agent shall not distribute to Buyer or Seller Agent the segregated portion of the Special Escrow Fund subject to the Claim Notice until the Escrow Agent shall have received joint written instructions from Buyer and Seller Agent, in form and substance reasonably satisfactory to the Escrow Agent, concerning the determination of the amount of the payment, if any, to which Buyer is entitled.
- (d) After an Objection has been delivered to the Escrow Agent and Buyer, Buyer and Seller Agent shall attempt in good faith to agree upon the rights of Buyer with respect to each of the amounts that comprise the asserted Claim Amount (or the amount subject to the Claim Notice which is disputed by Seller Agent in the Objection). If Buyer and Seller Agent should so agree, a memorandum setting forth such agreement shall be prepared and signed by both Buyer and Seller Agent and, in the case of a demand for recovery from the Special Escrow Fund, shall deliver to the Escrow Agent joint written instructions from Buyer and Seller Agent setting forth such agreement. If no such agreement can be reached after good faith negotiation within 30 days after delivery of an Objection, the amounts in dispute shall be submitted to binding arbitration as set forth in Section 2.8(c)(iv) of the Purchase Agreement. The Escrow Agent will (i) pay Buyer out of the Special Escrow Fund the amount that Buyer is entitled to receive as promptly as practicable after receiving joint written instructions concerning such final determination from Buyer and Seller Agent and (ii) if applicable, cause to be distributed to Seller Agent, for the benefit of and distribution to Sellers, such amount as to which Sellers would be entitled to distribution under Paragraph 4.7 hereof after giving effect to the foregoing joint written instructions.

- (e) Notwithstanding the foregoing, if, prior to the Escrow Release Date Mine 8 (as defined in the Purchase Agreement) is designated on a Pattern of Violation (POV) list by the Mine Safety and Health Administration that is attributable to the Prior Violations, Buyer shall be entitled to payment related to any Prior Violations pursuant to Section 2.8(c)(iii) of the Purchase Agreement, Buyer shall have the right, at its sole discretion, to submit a Claim Notice for Losses or other damages to the Escrow Agent and receive payments from the General Escrow Fund before proceeding against the Special Escrow Fund.
- 4.4 At any time prior to the final termination of this Escrow Agreement, the Escrow Agent shall, if jointly instructed in a writing signed by Buyer and Seller Agent, release or cause to be released from the Escrow Fund to Buyer or Seller Agent, for the benefit of and distribution to Sellers, as directed, the amount of cash specified in any such writing.
 - 4.5 The Escrow Agent shall not dispose of all or any portion of the amounts in the Escrow Fund other than as specifically provided in this Escrow Agreement.
- 4.6 On the fourteen (14) month anniversary of the date of this Escrow Agreement (the "Escrow Release Date"), the Escrow Agent shall cause to be distributed from the General Escrow Fund to:
- (a) Buyer, as promptly as practicable after receiving joint written instructions from Buyer and Seller Agent, an amount equal to (1) \$5,000,000 less amounts previously paid to Buyer pursuant to Section 2.8(b)(iv) of the Purchase Agreement <u>multiplied by</u> (2) the ratio of (x) 10,000,000 <u>minus</u> the number of tons of Additional Reserves (as defined in the Purchase Agreement) proven prior to the Escrow Release Date to (y) 10,000,000 (plus any accrued interest or earnings on such amount); <u>provided, however</u>, that if the number of tons of Additional Reserves is equal to or greater than 10,000,000 tons, Buyer shall not be entitled to any distribution from the General Escrow Fund pursuant to this <u>Paragraph 4.6(a)</u>; and
- (b) To Seller Agent, for the benefit of and distribution to Sellers, an amount equal to (1) any amount then remaining in the General Escrow Fund (including any remaining accrued interest or earnings thereon), less (2) if the distribution contemplated by Paragraph 4.6(a) hereof has not occurred, the amount that Buyer claims to be entitled to pursuant to Paragraph 4.6(a) as certified to the Escrow Agent in writing by Buyer no later than the Business Day preceding the Escrow Release Date_less (3) any amount subject to a Claim Notice delivered prior to the Escrow Release Date pursuant to Paragraph 4.2 to the extent such Claim Notice has not been resolved prior to such date. Any amount retained in escrow from the General Escrow Fund after the Escrow Release Date pursuant to the preceding sentence shall be held by the Escrow Agent and shall be used to make any such distribution or distributions contemplated by Paragraph 4.2 hereof and to indemnify or make a payment to the Buyer Indemnitees for such unresolved Claim Notices subject to the terms and conditions of the Purchase Agreement (but only as to Buyer and Seller Agent) and this Escrow Agreement (being the only document by which the Escrow Agent is bound) and upon resolution and if applicable payment out of the General Escrow Fund in satisfaction of the foregoing, any remaining amounts in escrow (including any remaining interest or earnings thereon), shall be promptly distributed by, on behalf of or at the direction of, the Escrow Agent to Seller Agent, for the benefit of and distribution to Sellers.

4.7 Upon the final resolution of all legal proceedings and fines relating to the Prior Violations, or on joint written direction of Buyer and Seller Agent (such date, the 'Final Release Date"), the Escrow Agent shall pay to Seller Agent, for the benefit of and distribution to Sellers, an amount equal to (1) any amount then remaining in the Special Escrow Fund (including any remaining accrued interest or earnings thereon), less (2) any amount subject to a Claim Notice delivered prior to the Final Release Date pursuant to Paragraph 4.3 to the extent such Claim Notices has not been resolved prior to such date. Any amount retained in escrow from the Special Escrow Fund after the Final Release Date pursuant to the preceding sentence shall be held by the Escrow Agent and shall be used to make any such distribution or distributions contemplated by Paragraph 4.3 hereof and to make a payment to Buyer for such unresolved Claim Notices subject to the terms and conditions of the Purchase Agreement (but only as to Buyer and Seller Agent) and this Escrow Agreement (being the only document by which the Escrow Agent is bound) and upon resolution and if applicable payment out of the Special Escrow Fund in satisfaction of the foregoing, any remaining amounts in escrow (including any remaining interest or earnings thereon), shall be promptly distributed by, on behalf of or at the direction of, the Escrow Agent to Seller Agent, for the benefit of and distribution to Sellers.

4.8 Subject to the other provisions of this Escrow Agreement, upon distribution of the entire amount of the Escrow Fund (including amounts withheld pursuant to unresolved Claim Notices, if any), this Escrow Agreement shall be terminated.

5. INVESTMENT; EARNINGS.

The Escrow Agent is hereby authorized and directed to hold and invest the Escrow Fund and to distribute such Escrow Fund in accordance with the terms of this Escrow Agreement. At the joint written direction of Buyer and Seller Agent, the Escrow Agent shall invest and reinvest the Escrow Fund in (a) obligations issued or guaranteed by the United States of America or any agency or instrumentality thereof; (b) certificates of deposits (including money market certificates and similar instruments) of or accounts with national banks or holding companies of national banks having capital and surplus in excess of \$100,000,000 at the time of investment; (c) any money market account or fund, whether insured or uninsured, including, without limitation, a money market account at Wells Fargo Bank, National Association; (d) repurchase agreements with U.S. Treasuries as collateral; and (e) other investments that are mutually approved in writing by Buyer and Seller Agent. In the absence of joint written instructions from Buyer and Seller Agent, the Escrow Fund shall be invested and reinvested in accordance with the Investment Selection Instructions set forth as Exhibit A hereto. Interest accruing on, and any profit realized from, such investment shall be credited, and any loss resulting from such investments shall be charged, to the Escrow Fund. Subject to Paragraph 8.4 below, all interest or earnings earned on amounts in the Escrow Fund, if any, shall be distributed to Seller Agent, for the benefit of and distribution to Sellers, at the times specified in Paragraphs 4.2, 4.3, 4.6 and 4.7 hereof. The Escrow Agent is hereby authorized and directed to sell and reduce to cash suchinvestments, it necessary, to make disbursements under this Escrow Agent and Agent shall not be responsible for any profit or loss realized on such investments supervision, recommendations, or advice.

6. LIABILITY AND COMPENSATION OF ESCROW AGENT.

6.1 The Escrow Agent's obligations and duties in connection herewith are confined to those specifically enumerated in this Escrow Agreement, which obligations and duties shall be deemed purely ministerial, and not fiduciary, in nature. The Escrow Agent shall not be required to take notice of the Purchase Agreement and shall have no duty or responsibility to take any action pursuant thereto. The Escrow Agent shall not be in any manner liable or responsible for the sufficiency, correctness, genuineness, or validity of any documents deposited with it or with reference to the form of execution thereof, or the identity, authority, or rights of any Person executing or depositing the same, and the Escrow Agent shall not be liable for any loss that may occur by reason of forgery, false representation, or the exercise of its discretion in any particular manner or for any other reason, except for its own fraud, gross negligence or willful misconduct. The Escrow Agent shall not be required to act upon or take notice of any direction, demand, notice, communication or instructions provided to the Escrow Agent by any Seller, but shall act upon and take notice solely of notices, communications or instructions provided to the Escrow Agent by Seller Agent and Buyer as provided for herein. The Escrow Agent shall have no liability with respect to the transfer or distribution of any funds effected by the Escrow Agent pursuant to wiring or transfer instructions provided to the Escrow Agent by any party to this Escrow Agreement in accordance with the provisions of this Escrow Agreement. The Escrow Agent shall not be obligated to take any legal action or to commence any proceedings in connection with the Escrow Fund or this Escrow Agreement, or to appear in, prosecute or defend in any such legal action or proceedings.

6.2 The Seller Agent, on behalf of each Seller, and Buyer jointly and severally covenant and agree to indemnify, defend and hold harmless the Escrow Agent and its officers, directors, employees and agents from and against any and all losses, claims, damages, liabilities, and reasonable out-of-pocket expenses, including without limitation, reasonable costs of investigation and counsel fees and disbursements which may be imposed on the Escrow Agent or incurred directly or indirectly in connection with this Escrow Agreement, its acceptance of this appointment as the Escrow Agent hereunder, or of the performance of its duties hereunder, including without limitation, any litigation arising from this Escrow Agreement or involving the subject matter hereof; provided, that if the Escrow Agent shall be found guilty of fraud, willful misconduct or gross negligence in connection with its services under this Escrow Agreement then, in that event, the Escrow Agent shall bear all such losses, claims, damages, liabilities, and expenses; provided further, that solely as between Buyer and Sellers, this Paragraph 6.2 hereof shall not prejudice any right or obligations of Buyer and Sellers among themselves pursuant to the Purchase Agreement. The provisions of this Paragraph 6.2 hereof shall survive the termination of this Escrow Agreement and any resignation or removal of the Escrow Agent.

6.3 If a dispute occurs among the parties hereto or otherwise with respect to the Escrow Fund, the Escrow Agent shall be entitled to tender into the registry or custody of any court of competent jurisdiction, upon an order of such court, all of the Escrow Fund, together with such legal pleadings as it or its counsel deems appropriate, and, thereupon be discharged from all further duties and liabilities under this Escrow Agreement. The filing of any such legal proceedings shall not deprive the Escrow Agent of its compensation (as provided for herein) earned prior to such filing and the provisions of <u>Paragraph 6.1</u> hereof shall survive as to such action.

- 6.4 The Escrow Agent's fees hereunder shall be as set forth in Schedule A hereto (the "Escrow Agent's Compensation") and shall be paid one-half by the Seller Agent, on behalf of each Seller, and one-half by Buyer at the times and in the manner set forth in Schedule A. In addition, Seller Agent, on behalf of each Seller, and Buyer shall each pay one-half of all other amounts due and owing Escrow Agent pursuant to Schedule A. This fee is intended as full compensation for the Escrow Agent's services as contemplated by this Escrow Agreement; provided, however, that if the conditions of this Escrow Agreement are not fulfilled or the Escrow Agent renders any reasonably required material service not contemplated in this Escrow Agreement, or there is any assignment of interest in the subject matter of this Escrow Agreement, or any material modification hereof, or if any material controversy arises hereunder, or the Escrow Agent is made a party to or justifiably intervenes in any litigation pertaining to this Escrow Agreement or the subject matter hereof, the Escrow Agent shall be reasonably compensated for such extraordinary services and reimbursed for all reasonable out-of-pocket costs and expenses, including reasonable attorney's fees, occasioned by such controversy, litigation, or event, and the same shall be payable by Seller Agent, on behalf of each Seller, and Buyer, pursuant to this Paragraph 6.4 hereof.
- **6.5** The Escrow Agent may resign at any time, upon thirty (30) days prior written notice to Buyer and Seller Agent, and prior to such resignation shall deposit the Escrow Fund with a successor escrow agent, which shall be a national bank, to be jointly designated in writing by Buyer and Seller Agent. Any such successor escrow agent must agree to be, and shall be bound by, and shall have all the rights, duties, and responsibilities of the Escrow Agent under this Escrow Agreement. If, upon the effective date of such resignation, no successor escrow agent shall have been designated, the Escrow Agent shall be entitled to treat such failure to designate a successor as a dispute subject to the provisions of <u>Paragraph 6.3</u> hereof. Such resignation shall not deprive the Escrow Agent of its compensation (as provided for herein) earned prior thereto.
- 6.6 Buyer and Seller Agent may terminate this Escrow Agreement (including in connection with the appointment of a new escrow agent) by joint written notice which shall specify the disposition of the Escrow Fund (which disposition must be in a manner consistent with the Purchase Agreement) and the Escrow Agent shall, upon receipt thereof, promptly disburse the Escrow Fund in accordance with such specified disposition.
- 6.7 The Escrow Agent may from time to time consult with legal counsel of its own choosing in the event of any disagreement or controversy or question or doubt as to the construction of any of the provisions hereof or its duties hereunder, and it shall incur no liability and shall be fully protected in acting in good faith in accordance with the written opinion or instructions of such counsel, except in instances of the Escrow Agent's own fraud, gross negligence or willful misconduct. Any reasonable fees and expenses of such legal counsel shall be considered part of the fees and expenses of the Escrow Agent described herein.

- **6.8** No provision of this Escrow Agreement shall require the Escrow Agent to risk or advance its own funds or otherwise incur any financial liability or potential financial liability in the performance of its duties or the exercise of its rights hereunder.
- 6.9 Subject to Paragraph 6.6, any corporation or association into which the Escrow Agent may be converted or merged, or with which it may be consolidated, or to which it may sell or transfer all or substantially all of its corporate trust business and assets as a whole or substantially as a whole, or any corporation or association resulting from any such conversion, sale, merger, consolidation or transfer to which the Escrow Agent is a party, shall be and become the successor escrow agent under this Escrow Agreement and shall have and succeed to the rights, powers, duties, immunities and privileges as its predecessor, without the execution or filing of any instrument or paper or the performance of any further act.
- 6.10 In the event that any of the Escrow Fund shall be attached, garnished or levied upon by any court order, or the delivery thereof shall be stayed or enjoined by an order of a court, or any order, judgment or decree shall be made or entered by any court order affecting the Escrow Fund, the Escrow Agent is hereby expressly authorized, in its sole discretion, to respond as it deems appropriate or to comply with all writs, orders or decrees so entered or issued, or which it is advised by legal counsel of its own choosing is binding upon it, whether with or without jurisdiction. In the event that the Escrow Agent obeys or complies with any such writ, order or decree it shall not be liable to any of the parties hereto or to any other person, firm or corporation, should, by reason of such compliance notwithstanding, such writ, order or decree be subsequently reversed, modified, annulled, set aside or vacated.
- **6.11** The Escrow Agent shall not be responsible or liable for any failure or delay in the performance of its obligation under this Escrow Agreement arising out of or caused, directly or indirectly, by circumstances beyond its reasonable control, including, without limitation, acts of God; earthquakes; fire; flood; wars; acts of terrorism; civil or military disturbances; sabotage; epidemic; riots; interruptions, loss or malfunctions of utilities, computer (hardware or software) or communications services; accidents; labor disputes; acts of civil or military authority or governmental action; it being understood that the Escrow Agent shall use commercially reasonable efforts which are consistent with accepted practices in the banking industry to resume performance as soon as reasonably practicable under the circumstances.

7. SELLER AGENT.

- 7.1 Pursuant to the Purchase Agreement, Seller Agent shall act on behalf of the Sellers and shall have such powers as are delegated under or pursuant to the Purchase Agreement. The Seller Agent shall cause to be distributed all distributions to it from the Escrow Fund to or for the benefit of the Sellers in accordance with and subject to the terms of the Purchase Agreement.
- 7.2 None of Buyer, any Buyer Indemnitee or the Escrow Agent shall be responsible or liable for any acts or omissions of Seller Agent acting in Seller Agent's capacity as such.

8. TAXES.

- 8.1 For all federal, state, local and foreign tax reporting purposes, all interest and earnings on the Escrow Fund in any calendar year shall for tax purposes be allocated to and reported by Buyer. Unless and until any portion of the Escrow Fund is distributed to Seller Agent, for the benefit of and distribution to Sellers, the Escrow Fund shall be considered as owned by Buyer for all federal, state or local tax purposes.
- 8.2 The Escrow Agent annually shall file information returns on Form 1099 with respect to such interest and earning payments consistent with the provisions of Paragraph 8.1 hereof. Buyer shall provide to the Escrow Agent all forms and information necessary to complete such information returns, including but not limited to IRS Form W-8 BEN or W-9, as applicable. The Escrow Agent shall have no duty to prepare or file any federal or state tax report or return with respect to the Escrow Fund or any income earned thereon.
- **8.3** In the event the Escrow Agent becomes liable for the payment of taxes, including withholding taxes, relating to income or earnings derived from the Escrow Fund or any payment made hereunder, the Escrow Agent may deduct such taxes from such income, earnings, or payment allocable or otherwise payable to Buyer for such taxes.
- 8.4 No later than fifteen (15) days after the end of each quarter, and upon the distribution to Seller Agent of any portion of the Escrow Fund pursuant to Paragraph 4.2, 4.3, 4.6 or 4.7 hereof or otherwise pursuant to this Escrow Agreement, the Escrow Agent shall distribute to Buyer forty percent (40%) of any interest or earnings accrued in respect of the Escrow Fund during such quarter or, in the case of a distribution of any portion of the Escrow Fund, from the first day of the quarter in which such distribution is made and until and including the date of such distribution (an "Income Tax Release"); provided that, for the avoidance of doubt, any Income Tax Release shall have priority and precede any other release of interest and earnings from the Escrow Fund.

9. SUCCESSORS AND ASSIGNS.

This Escrow Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns but shall not be assignable by any party hereto; provided that (i) Buyer or Seller Agent may assign its rights and delegate its obligations hereunder to any Affiliate or to any successor corporation in the event of a merger, consolidation or transfer or sale of all or substantially all of Buyer's or Seller Agent's stock or assets; and (ii) the Escrow Agent may assign its rights hereunder to a successor escrow agent appointed in accordance with the provisions hereof. In the event of any such proposed assignment, an amendment to this Escrow Agreement, in form and substance reasonably acceptable to the Escrow Agent, shall be executed and delivered if the Escrow Agent reasonably deems such an amendment to be necessary or desirable.

10. NOTICES.

Any notice, demand, or communication required or permitted to be given by any provision of this Escrow Agreement shall be deemed to have been sufficiently given or served for all purposes if (i) personally delivered, (ii) sent by an internationally recognized overnight courier service, (iii) sent by registered or certified mail, return receipt requested, postage prepaid. Except as otherwise provided herein, any notice under this Escrow Agreement will be deemed to have been given (i) on the date such notice is personally delivered or delivered by facsimile, (ii) the next succeeding Business Day after the date such notice is delivered to the overnight courier service if sent by overnight courier, or (iii) five (5) Business Days after the date such notice is sent by registered or certified mail; provided that in each case notices received after 4:00 p.m. (local time of the recipient) shall be deemed to have been duly given on the next Business Day. Notwithstanding anything to the contrary herein provided, the Escrow Agent shall not be deemed to have received any notice or communication hereunder prior to the Escrow Agent's actual receipt thereof. The addresses, as each may be updated from time to time by providing proper notice hereunder, of the parties for purposes of this Escrow Agreement are:

(i) If to Buyer: James River Coal Company

Attn: Peter Socha, Chairman and Chief

Executive Officer

901 E. Byrd Street, Suite 1600 Richmond, Virginia 23219-4080

with a copy, which shall not Kilpatrick Townsend & Stockton LLP

Attn: David A Stockton 1100 Peachtree Street

Suite 2800

Atlanta, Georgia 30309

(ii) If to Escrow Agent: Wells Fargo Bank, National Association

Attn: Stefan Victory; Corporate, Municipal and

Escrow Solutions

7000 Central Parkway, Suite 550

Atlanta, Georgia 30328

(iii) If to Seller Agent:

constitute notice, to:

International Resource Partners GP LLC c/o Lightfoot Capital Partners GP LLC Attn: Vince Cubbage, Chief Executive Officer

725 Fifth Avenue, 19th Floor New York, NY 10022

with a copy, which shall not Jackson Kelly PLLC

constitute notice, to: Attn: Louis S. Southworth, II

Elizabeth Osenton Lord

500 Laidley Tower

P.O. Box 553

Charleston, West Virginia 25302 (25322)

Any party may change the address to which notices, requests, demands or other communications to such party will be delivered or mailed by giving notice of the address change to the other parties to this Escrow Agreement in the manner provided in this Escrow Agreement.

11. MISCELLANEOUS

- 11.1 The Paragraph headings contained in this Escrow Agreement are inserted for convenience only and shall not affect in any way the meaning or interpretation of this Escrow Agreement.
- 11.2 Whenever possible, each provision of this Escrow Agreement shall be interpreted in such manner as to be effective and valid under applicable Law, but if any provision of this Escrow Agreement is held to be prohibited by or invalid under applicable Law, such provision shall be ineffective only to the extent of such prohibition or invalidity, without invalidating the remainder of such provision or the remaining provisions of this Escrow Agreement.
- 11.3 Any provision of this Escrow Agreement may be amended or waived only in writing signed by Buyer, the Escrow Agent and Seller Agent. No waiver of any provision hereunder or any breach or default thereof shall extend to or affect in any way any other provision or prior or subsequent breach or default. No course of dealing between the parties to this Escrow Agreement shall be deemed to affect or to modify, amend or discharge any provision or term of this Escrow Agreement. No delay by any party to this Escrow Agreement in the exercise of any of its rights or remedies shall operate as a waiver thereof, and no single or partial exercise of any of its rights or remedies shall operate as a waiver thereof, and no single or partial exercise by any party to this Escrow Agreement of any such right or remedy shall preclude any other or further exercise thereof. A waiver of any right or remedy on any one occasion shall not be construed as a bar to or waiver of any such right or remedy on any other occasion.
- 11.4 Except for the Purchase Agreement as it applies only to the parties who executed the Purchase Agreement, this Escrow Agreement constitutes the complete agreement among the parties hereto with respect to the subject matter hereof and supersedes any prior understandings, agreements or representations by or among the parties, written or oral, which may have related to the subject matter hereof in any way.
- 11.5 The parties have participated jointly in the negotiation and drafting of this Escrow Agreement. In the event an ambiguity or question of intent or interpretation arises, this Escrow Agreement shall be construed as if drafted jointly by the parties and no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any of the provisions of this Escrow Agreement. The word "including" shall mean including without limitation.
- 11.6 Nothing expressed or implied or referred to in this Escrow Agreement will be construed to give any Person other than the parties to this Escrow Agreement and the Sellers any legal or equitable right, remedy, or claim under or with respect to this Escrow Agreement or any provision of this Escrow Agreement.

- 11.7 This Escrow Agreement may be executed in multiple counterparts, any one of which need not contain the signatures of more than one party, but all such counterparts taken together shall constitute one and the same instrument.
- 11.8 Buyer and Seller Agent shall have the right under this Escrow Agreement, upon request, to examine and audit, during business hours or at such other times as might be reasonable under the circumstances and with reasonable notice to the Escrow Agent, any and all of the books, records, or other information of the Escrow Agent concerning this Escrow Agreement or the Escrow Fund, at the requesting party's expense.
- 11.9 The construction and performance of this Escrow Agreement will be governed by the Laws of the State of New York without giving effect to the choice or conflict of Law provisions thereof.
- 11.10 WAIVER OF JURY TRIAL. THE PARTIES HEREBY IRREVOCABLY WAIVE ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY LEGAL PROCEEDING ARISING OUT OF OR RELATED TO THIS ESCROW AGREEMENT, WHETHER NOW EXISTING OR HEREAFTER ARISING, AND WHETHER SOUNDING IN CONTRACT, TORT OR OTHERWISE. THE PARTIES AGREE THAT ANY OF THEM MAY FILE A COPY OF THIS PARAGRAPH 11.10 WITH ANY COURT AS WRITTEN EVIDENCE OF THE KNOWING, VOLUNTARY AND BARGAINED-FOR AGREEMENT AMONG THE PARTIES IRREVOCABLY TO WAIVE TRIAL BY JURY AND THAT ANY ACTION OR PROCEEDING WHATSOEVER BETWEEN THEM RELATING TO THIS ESCROW AGREEMENT SHALL INSTEAD BE TRIED IN A COURT OF COMPETENT JURISDICTION BY A JUDGE SITTING WITHOUT A JURY.
- 11.11 Contemporaneously with the execution and delivery of this Escrow Agreement and, if necessary, from time to time thereafter, each of the parties to this Escrow Agreement (other than the Escrow Agent) shall execute and deliver to the Escrow Agent a Certificate of Incumbency substantially in the form of Exhibit B-1 and Exhibit B-1 and Exhibit B-1 and Escrow Agent and Escrow Agent</

11.12 Escheat. Buyer and Seller Agent are aware that under applicable state law, property which is presumed abandoned may under certain circumstances escheat to the applicable state. The Escrow Agent shall have no liability to Buyer, Seller Agent, their respective heirs, legal representatives, successors and assigns, or any other party, should any or all of the Escrow Fund escheat by operation of law, except to the extent resulting from the Escrow Agent's gross negligence or willful misconduct.

[Signatures Appear on Following Page]

IN WITNESS WHEREOF, the parties hereto have executed this Escrow Agreement as of the date first written above. **BUYER:** JAMES RIVER COAL COMPANY By: Name: Title: Tax Identification Number: ESCROW AGENT: WELLS FARGO BANK, NATIONAL ASSOCIATION, as escrow agent By: Name: Title: INTERNATIONAL RESOURCE PARTNERS GP LLC solely in its capacity as Seller Agent: By: Name: Title:

SCHEDULE A

FEE SCHEDULE

Attached.

EXHIBIT A

INVESTMENT SELECTION INSTRUCTIONS

Agency and Custody Account Direction For Cash Balances Wells Fargo Money Market Deposit Accounts

Direction to use the following Wells Fargo Money Market Deposit Accounts for Cash Balances for the escrow account or accounts (the "Account") established under the Escrow Agreement to which this Exhibit A is attached.

You are hereby directed to deposit, as indicated below, or as we shall jointly direct further in writing from time to time, all cash in the Account in the following money market deposit account of Wells Fargo Bank, National Association:

Wells Fargo Money Market Deposit Account (MMDA)

Each of us, acting in our respective representative capacities only, understands that amounts on deposit in the MMDA are insured, subject to the applicable rules and regulations of the Federal Deposit Insurance Corporation (FDIC), in the basic FDIC insurance amount of \$250,000 per depositor, per insured bank. This includes principal and accrued interest up to a total of \$250,000.

Each of us, acting in our respective representative capacities only, acknowledge that we have full power to direct investments of the Account.

We each understand that we may change this direction at any time only upon the giving of joint written instructions signed by each of us and that it shall continue in effect until revoked or modified by us by joint written notice to you.

Authorized Representative	Authorized Representation
James River Coal Company	International Resource Partners GP LLC
Date	Date

EXHIBIT B-1

CERTIFICATE OF INCUMBENCY

The specimen signatures shown below are the specimen signatures of the individuals who have been designated as authorized representatives of James River Coal Company and are authorized to initiate and approve transactions of all types for the escrow account or accounts established under the Escrow Agreement to which this Exhibit B-1 is attached, on behalf of James River Coal Company.

Specimen Signature
Signature
Signature
Signature
Signature
, and the second

EXHIBIT B-2

CERTIFICATE OF INCUMBENCY

The specimen signatures shown below are the specimen signatures of the individuals who have been designated as authorized representatives of International Resource Partners GP LLC and are authorized to initiate and approve transactions of all types for the escrow account or accounts established under the Escrow Agreement to which this Exhibit B-2 is attached, on behalf of the Sellers.

Name / Title	Specimen Signature
Name	Signature
Title	
Name	Signature
Title	
Name	Signature
	•
Title	
Name	Signature
	Ç.
Title	

CERTIFICATIONS

- I, David J. Schulte, certify that:
- 1. I have reviewed this report on Form 10-Q of Tortoise Capital Resources Corporation;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: April 7, 2011

By: /s/ David J. Schulte

David J. Schulte

Chief Executive Officer

CERTIFICATIONS

- I, Terry Matlack, certify that:
- 1. I have reviewed this report on Form 10-Q of Tortoise Capital Resources Corporation;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: April 7, 2011

By: /s/ Terry Matlack

Terry Matlack

Chief Financial Officer

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of Tortoise Capital Resources Corporation (the "Company") on Form 10-Q for the period ended February 28, 2011 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), we, David J. Schulte, Chief Executive Officer of the Company and Terry Matlack, Chief Financial Officer of the Company, certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that: (1) the Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934, as amended; and (2) the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: April 7, 2011 By: /s/ David J. Schulte

David J. Schulte Chief Executive Officer

Date: April 7, 2011 By: /s/ Terry Matlack

Terry Matlack

Chief Financial Officer

A signed original of this written statement required by Section 906 has been provided to the Company and will be retained by the Company and furnished to the Securities and Exchange Commission or its staff upon request.